Enhancement-mode Polar Sourced Gallium Nitride MOSFET

by

Kyle M. Bothe

A thesis submitted in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

in

Solid State Electronics

Department of Electrical and Computer Engineering University of Alberta

© Kyle M. Bothe, 2015

### Abstract

All commercially fabricated Gallium Nitride (GaN) based power transistors to date have been heterojunction field effect transistors (HFET). The major down fall of this design architecture is the inability to produce an inherently true normally-off device. The more traditional metal-oxide-semiconductor field effect transistor (MOSFET) design has the potential for power efficiency and enhancement-mode device operation. GaN has been touted as the next promising semiconductor for use in high frequency and high power applications. Various potential applications range from low frequency switching solid state transformers to inverters beyond 10 GHz frequency. These devices require high current densities, large breakdown voltages and the ability to operate in high temperature environments. Modern HFET technology has higher off-state leakage current caused from the minimum carrier density under the channel being larger than a conventional depleted GaN MOSFET. This behavior is crucial for high power applications as the off-state power consumption has become one of the essential design parameters. To date, the limiting factors of producing a GaN MOSFET are the fabrication issues associated to essential components within the MOSFET architecture; a high quality gate dielectric and a large concentration of electrons along the source and drain.

Over the past decade, gate dielectrics grown on GaN have been extensively studied to obtain low interface defects, low leakage current and high channel mobility. Numerous fabrication methods have been attempted to produce an interface worthy for a device; however, the quality of the dielectrics on GaN have been insufficient to compete against an HFET design. The surface passivation of the semiconductor interface states has been studied and considered one of the most significant fabrication processes in a GaN MOSFET. Through a novel low temperature plasma-enhanced atomic layer deposition (PEALD) pretreatment, we have demonstrated interface traps densities below  $10^{11} cm^{-2} eV^{-1}$ , with corresponding dielectric capacitance densities greater than 2  $\mu F/cm^2$  for  $ZrO_2$  and  $HfO_2$  films. For the first time a dielectric on GaN has produced a high capacitance density (> 5x), whilst maintaining a high quality interface which significantly reduced the interface trap density (< 10x). This low temperature deposition technique has the ability to be applied to any current GaN technology for MOS applications.

GaN MOSFETs with a  $ZrO_2$  gate dielectric were fabricated on GaN-on-sapphire templates for the purpose of potential power switching applications. The GaN MOSFET design demonstrated enhancement-mode behavior without the complexity of high temperature fabrication and complex etching. The devices incorporated a unique selective deposition technique of ultra-thin PEALD AlN and  $ZrO_2$  films. This novel fabrication technique was one of the first reported low temperature processed unintentionally doped enhancement-mode GaN MOSFET with a peak current density of 50  $\mu A/mm$ , a peak transconductance of 40  $\mu S/mm$ , an intrinsic transconductance of 0.73 mS/mm and a channel electron mobility excess of 300  $cm^2/Vs$ . The advantage of the MOSFET architecture was depicted through the engineering capabilities of the threshold voltage of the device.

As power electronic systems desire improved internal power components for next generation circuit designs, the GaN MOSFET has shown great potential over the GaN HFET based on optimized simulations and the demonstration of high quality materials. This work has characterized novel low temperature PEALD gate dielectrics with improved properties on GaN for the potential of improved GaN MOSFET characteristics. Novel ultra-thin PEALD AlN films produced high electron densities along the source and drain regions through low temperature deposition. With the incorporation of these films and conventional commercial fabrication techniques the GaN MOSFET will have a distinct impact on power electronics.

### Acknowledgements

I would like to thank everyone who made this project possible. Without the support of many individuals I would not be writing this dissertation. First, I would like to thank Dr. Douglas Barlage as my academic advisor. His in-depth understanding of solid-state electronics provided great insight and assistance to this project.

This project would not have been possible without the support from Dr. Kenneth C. Cadien and his group members Dr. Amir Afshar  $(ZrO_2 \text{ and } HfO_2)$ , Pouyan Motamedi (AlN) and Triratna Muneshwar  $(ZrN_x)$ . The Cadien Nanofabrication Group provided some of the worlds best ultra-thin films for solid-state electronics. I would like to thank them for their countless hours of deposition, advice and material expertise.

I am proud to have had Alex Ma, Kevin Voon, Gem Shoute, Zohaib Hameed, Mourad Benlami, Brett Carnio, Mei Shen and Peter von Hauff as my colleagues through out my studies. Everyone was willing to collaborate, encourage and even criticize when needed. I wish them the best throughout their studies.

I am deeply thankful to my friends and family for all of the love and support. Finally to my wife Jennifer, without your encouragement and support this would not have been possible. Thank you.

### Preface

I, Kyle M. Bothe, am the principal contributor to all seven chapters in this thesis. In the thesis, Chapter 2 is based on the journal publication no. 4 and the conference publication no. 1 and no. 3. The work published in journal publication no. 1, no. 3 and no. 5 is described in Chapter 4. The basis of Chapter 5 included results published in journal publication no. 6 and conference publication no. 4 and no. 5. Finally, Chapter 6 is based on work included in the U.S. Provisional Patent no. 2 and conference publication no. 6. The work resulting in journal publication no. 7 and U.S. Provisional Patent no. 1 were not a part of the thesis.

Peter A. von Hauff had assisted me with the fabrication and characterization of the high- $\kappa$  dielectrics MOSCAPS discussed in Chapter 4 section 4.3 and 4.4.

Amir Afshar had provided the deposition of the optimized high- $\kappa$  rare earth metal oxides mentioned in Chapter 4, Chapter 5 and Chapter 6. Amir also assisted with the post fabricated annealing discussed in Chapter 5 section 5.4.1 and Chapter 6 section 6.3.1.

Kevin J. Voon had assisted me with the characterization and measurements of the AlN in Chapter 5 section 5.4 and ZrN Chapter 5 section 5.5.

Pouyan Motamedi had provided the deposition of the optimized PEALD AlN films mentioned in Chapter 5 and Chapter 6.

Triratna Muneshwar had provided the deposition of the optimized PEALD ZrN films mentioned in Chapter 5.

Alex M. Ma provided assistance with the fabrication of the GaN MOSFETs and the method for low-temperature lift-off of PEALD gate oxides.

Below is the list of published and submitted referred journals and conference papers that are resulted from my Ph.D. work.

#### Publications

#### **Journal Publications**

- Kyle M. Bothe, Peter A. von Hauff, Amir Afshar, Ali Foroughi-Abari, Kenneth C. Cadien and Douglas W. Barlage, "Capacitance Modeling and Characterization of Planar MOSCAP Devices for Wideband-gap Semiconductors with High-κ Dielectrics", IEEE Transactions on Electron Devices 59(10) 2662-2666 (2012)
- Alex M. Ma, Manisha Gupta, Fatema Rezwana Chowdhury, Mei Shen, Kyle M. Bothe, Karthik Shankar, Ying Tsui, and Douglas W. Barlage, "Zinc Oxide Thin Film Transistors with Schottky Source Barriers", Solid-State Electronics 76 104-108 (2012)
- Peter A. von Hauff, Amir Afshar, Ali Foroughi-Abari, Kyle M. Bothe, Kenneth C. Cadien and Douglas W. Barlage, "ZrO<sub>2</sub> on GaN Metal Oxide Semiconductor with Low Hysteresis via Atomic Layer Deposition" Applied Physics Letters 102, 251601 (2013)
- Kyle M. Bothe and Douglas W. Barlage, "Underlying Design Advantages for GaN MOSFETs Compared with GaN HFETs for Power Applications", Journal of Computational Electronics 13(1) 217-223 (2013)
- 5. Kyle M. Bothe, Peter A. von Hauff, Amir Afshar, Ali Foroughi-Abari, Kenneth C. Cadien and Douglas W. Barlage, "Electrical Comparison of HfO<sub>2</sub> and ZrO<sub>2</sub> Gate Dielectrics on GaN" IEEE Transaction on Electron Devices 60(12) 4119-4124 (2013)
- 6. Kevin J. Voon, Kyle M. Bothe, Pouyan Motamedi, Kenneth C. Cadien and Douglas W. Barlage, "Polarization Charge Properties of Low-Temperature Atomic Layer Deposition of AlN on GaN" Journal of Physics D: Applied Physics 47 345104 (2014)
- 7. Mourad Benlamri, Kyle M. Bothe, Alex M. Ma, Gem Shoute, Amir Afshar, Himani Sharma, Arash Mohammadpour, Manish Gupta, Kenneth C. Cadien, Ying Tsui, Karthik Shankar, and Douglas W. Barlage, "High-Mobility Solution-Processed Zinc Oxide Thin Films on Silicon Nitride" Physics Status Solidi: RRL (8) 871-875 (2014)

#### **U.S** Patent Applications

 Alex M. Ma, Manish Gupta, Kyle M. Bothe, Amir Afshar, Kenneth C. Cadien, Douglas W. Barlage, "Buried Source Schottky Barrier Thin Film Transistor And Method of Manufacture" U.S Provisional Patent Application No.61/913,601  Kyle M. Bothe, Alex M. Ma, Kevin J. Voon, Amir Afshar, Triratna Muneshwar, Pouyan Motamedi, Kenneth C. Cadien, Douglas W. Barlage, "Polar Source Gallium Nitride MOSFET Structure and Method of Fabrication" U.S Provisional Patent Application No. 61/967,239

### **Conference Proceedings**

- Kyle M. Bothe, Mark Johnson, Douglas W. Barlage, "TCAD Modeling and Simulation of a Field Plated GaN MOSFET for High Voltage Applications", CS MANTECH Conference, Palm Springs (2011)
- Peter A. von Hauff, Kyle M. Bothe, Amir Afshar, Ali Foroughi-Abari, Kenneth C. Cadien and Douglas W. Barlage, "High Mobility (210cm2/Vs), High Capacitance (7.2uF/cm2) ZrO<sub>2</sub> on GaN Metal Oxide Semiconductor" CS MANTECH Conference, Boston (2012)
- 3. Kyle M. Bothe, Peter A. von Hauff, Amir Afshar, Ali Foroughi-Abari, Kenneth C. Cadien and Douglas W. Barlage, "GaN MOSFET: Projections for High Power High Frequency Applications" International Conference on Simulation in Semiconductor Processes and Devices, Denver (2012)
- Kevin J. Voon, Kyle M. Bothe, Pouyan Motamedi, Kenneth C. Cadien and Douglas W. Barlage, "Engineered Tunneling Contacts with Low-Temperature Atomic Layer Deposition of AlN on GaN", CS MANTECH Conference, Denver (2014)
- Kyle M. Bothe, Alex M. Ma, Kevin J. Voon, Amir Afshar, Pouyan Motamedi, Kenneth C. Cadien, Douglas W. Barlage, "Selective Deposition of Low Temperature AlN Ohmic Contacts for GaN Devices" CS MANTECH Conference, Denver (2014)
- Kyle M. Bothe, Alex M. Ma, Amir Afshar, Pouyan Motamedi, Kenneth C. Cadien, Douglas W. Barlage, "Impact of Post Fabrication Annealing PEALD ZrO<sub>2</sub> for GaN MOSFETs" CS MANTECH Conference, Scottsdale (2015)

### In Preparation

 Kyle M. Bothe, Alex M. Ma, Kevin J. Voon, Amir Afshar, Triratna Muneshwar, Pouyan Motamedi, Kenneth C. Cadien, Douglas W. Barlage, "Reliability of Ultrathin PEALD Nitride Based Films in GaN FETs"  Kevin J. Voon, Kyle M. Bothe, Triratna Muneshwar, Kenneth C. Cadien, Douglas W. Barlage, "Polarization Charge Properties of Low-Temperature Atomic Layer Deposition of Semiconductor"

# Contents

1	Intr	roduction	1
	1.1	Motivation	1
	1.2	Figure of Merits	3
	1.3	Gallium Nitride Transistors	6
	1.4	Synopsis of Thesis	10
<b>2</b>	Dev	vice Design and Simulation	12
	2.1	Introduction	12
	2.2	Device Optimization	13
	2.3	Device Comparison	19
	2.4	Summary	24
3	Fab	rication Techniques	26
	3.1	Introduction	26
	3.2	Gate Dielectrics for GaN	28
	3.3	Source/Drain Carriers	29
	3.4	GaN Metalization	31
	3.5	GaN Substrates	32
	3.6	Device Isolation	33
	3.7	Summary	34
4	Gat	e Dielectric Characterization	35
	4.1	Introduction	35
	4.2	Dielectric Fabrication	36
	4.3	Current-Voltage Measurements	37
		4.3.1 Frenkel-Poole Conduction	37
		4.3.2 Quantum Mechanical Tunneling	39
	4.4	Capacitance-Voltage Measurements	42
	4.5	Summary	49

<b>5</b>	Sou	rce and Drain Characterization	50
	5.1	Introduction	50
	5.2	Nitride Film Fabrication	52
	5.3	Characterization Methods	53
	5.4	Aluminum Nitride	54
		5.4.1 Current-Voltage Characterization	55
		5.4.2 Capacitance-Voltage Characterization	58
		5.4.3 AlN Thickness Dependence	62
		5.4.4 AlN Summary	63
	5.5	Zirconium Nitride	65
		5.5.1 $ZrN_x$ on Silicon	65
		5.5.2 $ZrN_x$ on GaN	69
		5.5.3 $ZrN_x$ Summary	75
	5.6	AlN and $ZrN_x$ Comparison	75
	5.7	Nitride Reliability	77
	5.8	Summary	81
6	Gal	N MOSFET Fabrication and Characterization	82
	6.1	Introduction	82
	6.2	Device Fabrication	82
	6.3	Device Characterization	86
		6.3.1 GaN MOSFET Operation	87
		6.3.2 Breakdown Voltage Characterization	103
	6.4	Summary	111
7	Sun	nmary and Conclusions	113
	7.1	Directions for Future Work	116
Bi	ibliog	graphy	117

# List of Figures

1.1	The various markets and applications for GaN based devices. The inherent	
	material properties of GaN are starting to replace incumbent technologies	
	unable to advance electronic components required for new applications [3].	2
1.2	A visual comparison of peak operation capabilities for Si, GaAs and GaN	
	based on temperature and power applications. GaN has the ability to replace	
	all silicon power devices based on the ability to operate $200^{\circ}C$ higher and	
	with stand electric fields larger than 300 $V/\mu m$	5
1.3	The three most common GaN device architectures under investigation for	
	power applications are the (a) HFET, (b) MOS-HFET and (c) MOSFET	6
1.4	Peak current density with respect to gate length for the GaN HFET, MOS-	
	HFET and MOSFET.	8
1.5	HFET and MOSFET peak current density with respect to gate length based	
	on GaN properties. Improving the channel mobility and capacitance density	
	of current GaN MOSFET technology has the potential to surpass HFET	
	characteristics below 2 $\mu m$ gate lengths	9
2.1	The GaN MOSFET design and dimensions to be optimized for improved	
	device characteristics. The device architecture was optimized before field	
	plates were included which were used to improve the breakdown voltage	13
2.2	The maximum substrate thickness of the GaN and equivalent minimum gate	
	length were determined based on the depletion width determined from the	
	substrate doping concentration. Any combination of thickness and substrate	
	doping that remains below the curve will produce a fully depleted channel.	14
2.3	The simulated device breakdown voltage and peak current density based on	
	gate length. Through optimization of the gate length the power rating was	
	largest at a gate dimension of $1.2\mu m$	15
2.4	The gate-source length dependence on breakdown voltage and current density	
	for the GaN MOSFET. The gate-source separation was optimized to a peak	
	power rating of $0.7\mu m.$	16
2.5	Optimization of the source field plate based on length and height	17
2.6	The improved breakdown voltage with respect to drift region of the GaN	
	MOSFET with field plates compared to current HFET technology	18

2.7	(a) Simulated GaN MOSFET family of curves. (b) Voltage and electric field	
	<ul><li>dependence transfer characteristics for the simulate MOSFET and HFET.</li><li>(c) Breakdown voltage of the simulated HFET (600 V) and MOSFET (3500</li></ul>	
	V) with similar sized dimensions	20
2.8	The effective electron channel mobility was extracted from the simulated	20
2.0	family of curves for the GaN MOSFET and GaN HFET	21
2.9	The electron density during various modes of operation under the gate of the MOSFET and HFET. The electron distribution in the channel was examined near the source edge, middle and drain edge. (Top) The device is off, i.e. $V_{GS}$	
2.10	< $V_t$ , (Middle) $I_{DS} = 40 \ mA/mm$ , (Bottom) $I_{DS} = 175 \ mA/mm$ The electric field under the gate and drain region when ( <i>Left</i> ) $V_{DS}=100V$ and ( <i>Right</i> ) $V_{DS}=1600V$ . ( <i>Bottom</i> ) The electron density for both devices under the gate and drift region at $V_{DS}=1600V$ . The MOSFET suppressed the electric field and maintain a large superstantian of corrient under the sets	22
	the electric field and maintain a low concentration of carriers under the gate whereas the HFET was unable to control the channel depletion	23
	whereas the HTET was unable to control the channel depletion	20
3.1	Important regions and materials of the GaN MOSFET which require novel	
	fabrication techniques for improved device performance	27
3.2	Density of interface traps and capacitance density of dielectric on GaN. These	
	techniques were unable to obtain the high charge control and low interface	
	traps required for a MOSFET.	28
3.3	Polarized 2DEG carriers and corresponding sheet mobility for AlN and Al-	20
9.4	GaN films.	30
3.4	Contact resistance with respect to annealing temperatures to produce ohmic contacts on GaN. The current techniques require high annealing temperatures	
	and complex metal stacks.	31
3.5	Schematic of GaN MOSFET with mesa etching to reduce the leakage current	01
	through the unused bulk GaN.	33
4.1	Dielectric constant and effective electron mass for potential gate dielectrics	
	for GaN based devices. The corresponding band diagram and conduction	9.0
4.9	band offset can be inferred for each material	36
4.2	(Left) Schematic of the MOSCAP used to evaluate the dielectric/GaN in- terface. $(Right)$ TEM image of the high-k dielectric film grown on GaN via	
	PEALD	37
4.3	Measured current densities with respect to voltage for all film samples. The	51
	results were converted into F-P plots to obtain the conduction band offset	
	between the dielectric and GaN.	38

4.4	(Left) Cross section image of the MOSCAP. $(Right)$ The illustration of	
	the conduction band used to determine the electron tunneling through the	
	dielectric from the transfer matrix method.	40
4.5	The modeled quantum mechanical tunneling current was used to confirm the	
	measured conduction mechanisms and to extract the effective electron mass	
	of the $ZrO_2$ films. The model (with error bars) was matched to the measured	
	results by creating the band structure for a given voltage and adjusting the	
	effective mass of the dielectric. The effective electron mass of the $ZrO_2$ films	
	was determined as $m_{ZrO_2} = 0.2.$	41
4.6	The modeled quantum mechanical tunneling current was used to confirm	
	the measured conduction mechanisms and to extract the effective electron	
	mass of the the $HfO_2$ films. The model (with error bars) was matched to	
	the measured results by creating the band structure for a given voltage and	
	adjusting the effective mass of the dielectric. The effective electron mass of	
	the $HfO_2$ films was determined as $m_{HfO_2} = 0.18.$	42
4.7	(Left) An illustration of the modeled planar MOSCAP. (Right) The radial	
	cross section of the modeled planar MOSCAP.	43
4.8	Equivalent lumped circuit model under the inner contact for the planar	
	MOSCAP.	44
4.9	A small slice of the inner contact $(\partial \phi)$ visually converted into the equivalent	
	triangular tapered transmission line.	44
4.10	(a) Capacitance-frequency measurements used to characterize the films through	
	a planar MOSCAP model. (Inset) Voltage dependence of 40 cycles $HfO_2$	
	C-f plots (b) The extracted capacitance density of the MOSCAP obtained	
	from the C-f measurements. (Inset) Hysteresis of the 40 cycles C-V at 10 kHz.	46
4.11	Density of interface traps along the dielectric/GaN surface obtained from	
	$G_p - \omega$ and hysteresis measurements for (a) $HfO_2$ and (b) $ZrO_2$	47
4.12	The dielectric properties of the low temperature PEALD $HfO_2$ (red stars)	
	and $ZrO_2$ (blue stars) films compared to literature (triangles). The ultra-	
	thin films were able improve the capacitance density and interface traps by	
	a factor of 30	48
5.1	The proposed nitride based films for the source/drain regions compared three	
0.1	electrical properties imperative to produce low ohmic contacts, high carrier	
	densities and a reliable device; the effective electron mass, metal atomic radii	
	and polarization charge were compared between the nitride films. AlN has	
	the potential to form the largest 2DEG along the interface, while ensuring	
	the current would flow through the GaN	51
		91

5.2	The film thickness for AlN and $ZrN_x$ with respect to PEALD cycle were obtained from in-situ ellipsometry; courtesy of Pouyan Motamedi (AlN) and Triratna Muneshwar $(ZrN_x)$ .	52
5.3	Schematic of the test structures to evaluate the AlN and $ZrN_x$ films. The TLM structures were used to evaluate the contact and sheet resistance pro- duced by the 2DEG along the GaN interface. The MOS-TLM included a non-polarized region to extract the contact and sheet resistance of the nitride film, in addition to confirming the GaN sheet resistance. The advantage of the MOSHCAP was to extract the 2DEG through capacitance-voltage mea- surements.	53
5.4	The band structure of the AlN/GaN with the induced 2DEG carriers formed along the interface.	55
5.5	(a) The I-V measurements from the 75 $\mu m$ gap TLM followed the trend as the doping of the GaN was reduced. In addition, the Al metal contact produced higher current densities than the Cr counter part for both cases of GaN templates. (b) The Cr MOS-TLM results on N <sup>-</sup> GaN confirmed the contact resistance from two devices with different contact areas. The extracted contact resistivity between the different dimension devices was less then $5\%$	EG
5.6	than 5%	56 57
5.7	(a) As the TLM structures were annealed the current density of the 5 $\mu m$ TLM gap increased. (b) The improved conduction was strongly dependent on the reduced contact resistance with respect to annealing temperature. Annealing the device at 700°C was found to produce a low ohmic contact $(5x10^{-3} \ \Omega \cdot cm^2) \ \ldots \ $	58
5.8	The simulated accumulation thickness for the AlN/GaN and $ZrO_2$ /GaN to extract the dielectric constant of the films. The effective mass of the films determined the location of the peak centroid and the amount of carriers within the dielectric film. AlN allowed 4x the amount of carriers into the film compared to the $ZrO_2$ film.	59
5.9	The (a) capacitance and (b) conductance frequency dispersion based on the MOSCAP with and without a gate dielectric $(ZrO_2)$ under one of the metal contacts. By including the gate dielectric, the leakage conductance was reduced which provided less frequency dispersion than the AlN metal films.	60

5.10	The reverse C-V gate bias of the MOSHCAPs used to extract the 2DEG for	
	both GaN templates. The area between the C-V measurement and the the-	
	oretical depletion capacitance for GaN was the polarization interface charge.	
	The $N^+$ GaN (blue) produced a 2DEG roughly 10x larger than the $N^-$ GaN	
	(green)	61
5.11	The extracted contact and sheet resistance from the 5.7 nm and 7.9 nm	
	AlN/GaN films	62
5.12	(a) The comparison between carrier density to channel mobility of AlN films	
	on GaN. The low temperature films produced equivalent 2DEG charge and	
	carrier mobility as conventional high temperature deposition techniques. (b)	
	The contact resistance of the doped GaN tunneling contact produced similar	
	values to high temperature annealed metal contacts on GaN	64
5.13	(a) The measured I-V of the $ZrN_x/Si$ junction, courtesy of Kevin Voon.	-
0.20	(b) The frequency dependent C-V measurements of the $ZrN_x/Si$ , courtesy	
	of Kevin Voon. The rectifying current density confirmed that a Schottky	
	contact or pn-junction was formed. The extract built-in potential ( $\phi_{bi} =$	
	1.27 eV) of the material system confirmed a p-n junction was formed at the	
	$ZrN_x$ /Si interface.	66
5.14	Inverse squared capacitance of the measured test structure which was used	
0	to extract the electrical properties of the $ZrN_x$ film. The built-in potential	
	was obtained from the x-axis intercept and the silicon doping concentration	
	of was extracted from the slope	67
5.15	The expected band diagrams for $ZrN_x$ on different background doped GaN	
	templates based on the extracted electrical parameters obtained from the	
	silicon measurements.	69
5.16	(a) The current-voltage results of the $N^-$ (Solid) and $N^+$ (Dashed) GaN	
	templates. (b) The total resistance with respect to TLM distance used to	
	extract the contact resistance and sheet resistance of the $ZrN_x$ /GaN interface.	70
5.17	Reverse bias C-V measurements of the $ZrO_2/ZrN_x$ /GaN MOSHCAPs ex-	
	tracted the 2DEG along the GaN interface for (a) $N^+$ and (b) $N^-$ GaN	
	templates. The plots included the theoretical 2DEG expected from $ZrN_x$	
	modulation doping $(N_{ZrN})$ . The additional charge in the reverse bias con-	
	firmed that $ZrN_x$ had polarization properties	72
5.18	Low frequency C-V to obtain the dielectric characteristics of $ZrN_x$ on (a)	
	$N^+$ GaN and (b) $N^-$ GaN. (c) The band diagram at zero bias and strong	
	accumulation regime on N <sup>+</sup> GaN were both along the $ZrN_x/GaN$ interface.	
	(d) The accumulation of the electrons for the $N^-$ GaN template occurred	
	along the $Zr0_2/ZrN_x$ interface.	73
		-

5.19	(a) 2DEG carriers density and (b) Contract resistance based on the film thickness of AlN and $ZrN_x$	75
5.20	Contact resistance of the nitride films compared to the 2DEG formed along	
	the GaN interface.	76
5.21	Measured gate voltage based on injected gate current for different MOS GaN	
	devices. The voltage increased as the materials in the device failed due to	
	defects produced within the films	77
5.22	Capacitance-voltage measurements (100 kHz) for (a) $AlN/ZrO_2$ (b) $ZrN_x/ZrO_2$	
	and (c) $ZrO_2$ were shifted as breakdown occurred due to electron induced	-
	defects within the films.	78
5.23	Flat-band voltage shift and the associated traps within the film caused from	
	electrons injected through the gate. Lifetime of the devices based on worst	
	case scenario of 10 $\mu A/mm$ gate leakage. The lifetime ratio remains constant	
	regardless of the leakage current; i.e. The AlN film degraded 8x faster than	
	the $ZrN_x$ film.	79
5.24	(a) Reverse breakdown characteristics of the AlN and $ZrN_x$ films to extract	
	the materials critical electric field at $25^{\circ}C$ . (b) The materials reverse break-	
	down based on temperature. Based on the temperature dependence, the	
	$ZrN_x$ was more stable compared to the AlN	80
6.1	Alignment marks were placed on the GaN template to align future layers.	83
6.2	The negative photoresist lift-off technique defined the source and drain re-	
	gions where the ultra-thin ALN films was deposited. The gate region was	
	protected with photoresist to ensure volatile Al atoms did not disturb the	
	GaN interface.	84
6.3	The regions for the $ZrO_2$ gate dielectric were defined with same negative	
	photoresist process that defined the source and drain regions. The $ZrO_2$	
	overlapped the ALN along the source and drain to ensure the gate metal	
	would not come into contact with AlN.	85
6.4	The source, drain and gate contacts were all deposited at the same time to	
	reduce the fabrication steps of the device	86
6.5	Confocal image of the $100 \mu m \ge 100 \mu m$ FatFETs used to evaluate the quality	
	of the GaN MOSFET fabricated with the ultra-thin PEALD films. The gate	
	was extended 10 $\mu m$ beyond the width of the source and drain to reduce the	
	substrate leakage current.	87
6.6	A representation of the leakage current under the channel based dependence	
	on the depletion width (dashed region). The lower doped GaN produced a	
	larger depleted channel which was expected to create a lower leakage current.	88

(a) The inverse squared capacitance with respect to gate voltage for the  $N^+$ 6.7GaN (red) and  $N^-$  GaN (green). The x-axis intercept was used to extract the accumulation voltage of the channel MOS. (b) The drain current (red) at  $V_{DS} = 0.1$  V for  $N^+$  GaN with respect to gate voltage. . . . . . . . . 88 (a) The family of curves for the  $N^-$  GaN MOSFET. The drain saturation 6.8 voltage of the MOSFET was consistent with conventional MOSFET models where  $V_{D_{sat}} = V_{GS} - V_t$ . (b) The drain current (green solid) and transconductance (blue dashed) at  $V_{DS} = 1$  V for  $N^-$  GaN with respect to gate voltage. The device demonstrated enhancement-mode operation, with a turn-on volt-89 6.9 The transfer and transconductance characteristics of the GaN MOSFET had a strong dependence with the work function of the gate metal. When the gate metal was exchanged from Al ( $\Psi_{Al} = 4.2 \ eV$ ) to Cr ( $\Psi_{Cr} = 4.7 \ eV$ ) the MOS-FET characteristics shifted from normally-on  $(V_t = -0.25 V)$  to normally-off  $(V_t = 1.3 V)$ . The enhancement-mode MOSFET improved current density (2x) and transconductance (5x) over the Al gate device.  $\ldots$ 91 6.10 The sub-threshold slopes of the GaN MOSFETs were extracted from the low transverse electric field transfer characteristics for the Al (Green) and Cr (Blue) gate metals. The enhancement-mode GaN MOSFET produced a 92sharper sub-threshold slope compared to the depletion-mode transistor. . . 6.11 (a) The gate-source capacitance of the oxide was obtain at low frequency (50) kHz) to determine the peak capacitance density and extract the flat band voltage of the MOS stack, shown in the inset. During reverse gate bias, the Cr metal was capable of depleting more charge from under the gate compared to the Al gate stack. (b) The density of interface traps were obtained from the conductance frequency spectrum. The  $Al/ZrO_2$  gate stack degraded the interface, which produced a 5x more traps compared to the Cr metal. . . . 936.12 The internal voltage drops are dependent on the external voltage biases and parasitic resistances along the source and drain interconnects. The MOSFET only experienced the internal bias for the drain  $(V_D)$  and gate  $(V_G)$  based on 956.13 (a) The external and internal transconductance of the 100  $\mu m \ge 100 \mu m$ GaN MOSFET at  $V_D = 3$  V. The peak intrinsic transconductance was 0.73 mS/mm. (b) The capacitance-voltage based on the external and internal 96

6.14	The effective electron mobility of the FatFET obtained at low drain bias.	
	The mobility based on the external gate biases was determined to have a	
	peak mobility of 230 $cm^2/Vs$ Conversely, the peak electron mobility of the	
	GaN MOSFET without parasitic resistance has the potential to reach 480	
	$cm^2/Vs$	98
6 15	(a) The gate leakage current of the FatFET as the gate voltage was swept	00
0.10	from reverse to forward bias. Higher annealing temperatures increased the	
	conduction leakage through the dielectric. (b) The extracted density of inter-	
	face traps based on annealing temperature. The gate oxide was electrically	
	activated between $400^{\circ}C$ and $500^{\circ}C$ before the gate dielectric was critically	
	degraded	99
6 16	0	99
0.10	(a) The FatFET family of curves after the device was annealed at $300^{\circ}C$ (red)	
	and $500^{\circ}C$ (blue). The drain current was improved by a factor a 4x over the	
	low temperature annealed device. (b) The transfer characteristics after the $1 + 2000 G$ (c $1$ ) $- 1 = 5000 G$ (1 $1 - 2$ ). The transfer characteristics after the	
	device was annealed at $300^{\circ}C$ (red) and $500^{\circ}C$ (blue). The transconduc-	
	tance quadrupled over the original fabricated device due to the reduce series	100
	resistances.	100
6.17	(a) The leakage current through the gate dielectric with respect to gate volt-	
	age. The leakage current was reduced by a factor of 100 as the gate length	
	was reduced from 100 $\mu m$ to 10 $\mu m$ . (b) The GaN MOSFETs extracted	
	intrinsic transconductance were compared based on gate length. The 10 $\mu m$	
	channel increased the intrinsic transconductance to $15 mS/mm$ over the 100	
	$\mu m$ GaN MOSFET channel	102
6.18	A confocal image of the GaN MOSFET used to measure the breakdown	
	voltage. (a) The dashed lines are where the transparent films were patterned	
	and deposited. (b) The source and gate contacts were grounded while the	
	drain voltage was increased. The confocal image allowed for the transparent	
	$ZrO_2$ and ZrN films to be imaged. The AlN optical properties were identical	
	to GaN making imaging nearly impossible.	103
6.19	The substrate leakage current with respect to GaN background doing as a	
	function of drain voltage. The semi-insulating GaN was capable of preventing	
	the substrate leakage current beyond 50 V	104
6.20	The GaN MOS-HFET and MOSFET transistor architectures were compared	
	for the potential maximum drain voltage before the leakage current becomes	
	80 $\mu A/mm$ . The significant variation between transistors was the ultra-thin	
	AlN film under the gate metal	105

6.21 (a) The measured drain leakage current during the off-state of the GaN MOSFET. As the gate-drain length was increase, the device was capable of improving the maximum drain voltage. (b) The extrapolated breakdown voltage trend for the GaN MOSFET (red square) with respect to the drift region. The breakdown voltage of the leading GaN HFET device (green star) and the simulated GaN MOSFET (blue) were included. . . . . . . . . . 1066.22 The measured drain off-state leakage current for a 10  $\mu m$  gate GaN MOSFET with (solid green) and without (dashed blue) a floating field plate. The floating field plate increased the average critical electric field along the drain from 0.74 MV/cm to 1.4 MV/cm. 1086.23 The gate length extrapolated maximum drain voltage of GaN MOSFET architectures with different drift regions;  $L_{GD} = 1 \ \mu m$  (squares)  $L_{GD} = 5 \ \mu m$ 109(circles). 6.24 The breakdown of the device can be considered when the shift of the threshold voltage reached the low drain threshold voltage  $(V_{th}^{low})$ . This is where the 1106.25 The extrapolated breakdown voltage based on the substrate leakage current was examined as the AlN on the source-gate edge was removed from under the gate metal. A source-gate overlap of 100 nm ensured the breakdown voltage was close to 100 V/um. Removing the source carriers from under the

### gate prevented the transistor from withstanding larger drain voltages. . . . 111

# List of Tables

1.1	Material Properties of Semiconductors for Electronic Devices	2
1.2	Figure of Merits based on Material Properties for Power Electronic Devices	5
1.3	GaN Device Characteristics for Different Design Architectures	7
2.1	Optimized GaN MOSFET and HFET Dimensions	19
3.1	Gate Dielectrics for GaN Applications	29
3.2	Contact Resistance for GaN based Devices	32
4.1	Extracted Dielectric Properties on GaN	48
5.1	Extracted Contact and Sheet Resistance Properties of AlN on GaN templates	57
5.2	AlN Dielectric Constant Extracted from Peak Capacitance Density	59
5.3	Extracted Sheet and Contact Resistance	63
5.4	ZrN Dielectric Constant Extracted from Peak Capacitance Density	67
5.5	Extracted Electronic Properties of $ZrN_x$	68
5.6	$ZrN_x$ 2DEG Properties	74
5.7	$ZrN_x$ Contact Resistance Properties	74
6.1	The Extracted Density of Interface Traps	94
7.1	Low Temperature PEALD Gate Dielectrics on GaN	113
7.2	Ultra-thin Low Temperature PEALD Al N Films on GaN Templates $\ \ . \ . \ .$	114
7.3	GaN MOSFET Device Characteristics	114
7.4	GaN MOSFET Breakdown Voltage Characteristics	115

# Chapter 1

## Introduction

### 1.1 Motivation

The perpetual strive to advance electronic systems has become a repetitive occurrence between improving device characteristics and optimizing the design of the system. A power systems overall electronic performance such as power handling, efficiency, size, and reliability are strongly dependent on the enhancements made to individual semiconductor devices. Many power electronic systems require the transistors to operate under extreme temperatures and large power modes of operation [1]. Until now, silicon (Si) has been the dominating semiconductor substrate in power electronics because of the prior knowledge and processing techniques acquired from complementary-metal-oxide-semiconductor (CMOS) technology. However, current silicon based technology has been unable to maintain next generation specifications due to the physical properties that limit the device performance [2]. This is not due to the design of the silicon transistors; rather, the electrical properties of a narrow band-gap has started to reach its limits in the power electronics industry. Conversely, gallium nitride (GaN) has gained significant interest over many researched semiconductors because of the preferential material properties ideal for a wide range of electronic applications [3], shown in Figure 1.1. The next generation power market for GaN devices has been estimated to be \$300 million by 2015 with a growth rate over 30% per year [4]. A conservative growth rate, predicts that GaN based devices should reach approximately 5% of the \$40 billion power market by 2020.

Gallium nitride has unique semiconductor properties, such that a combination of bandgap, critical electric field, and electron saturation velocity has the potential to advance the



Figure 1.1: The various markets and applications for GaN based devices. The inherent material properties of GaN are starting to replace incumbent technologies unable to advance electronic components required for new applications [3].

transistors switching frequency, operating temperature and blocking voltage (off-state drain voltage) [5]. The material properties of well-known semiconductors, [6–10], commonly used to fabricate microelectronic devices were compiled and tabulated below in Table 1.1. The band-gap,  $E_G$ , intrinsic carrier concentration,  $n_i$ , dielectric constant,  $\epsilon_r$ , electron mobility,  $\mu_n$ , critical electric field,  $E_C$ , electron saturation velocity,  $\nu_s$ , and thermal conductivity,  $\Phi_k$ were considered the most important electronic properties of semiconductors. The wider band gap semiconductors, GaN and SiC, inherently produce lower intrinsic carrier concentrations which were beneficial for lower junction leakage at higher operating temperatures. Furthermore, wide band gap semiconductors have demonstrated larger critical electric fields compared to small band gap semiconductors such as Si and GaAs. The semiconductors with

Table 1.1: Material Properties of Semiconductors for Electronic Devices

Material	Si	GaAs	4H-SiC	GaN
$E_G (eV)$	1.12	1.42	2.6	3.37
$n_i \ (cm^{-3})$	$1.5 \mathrm{x} 10^{10}$	$2.1 \mathrm{x} 10^{6}$	$8.2 \text{x} 10^{-9}$	$1.9 \mathrm{x} 10^{-10}$
$\epsilon_r$	11.8	10.8	9.6	9.7
$\mu_n \ (cm^2/Vs)$	1350	8500	700	900
$E_C (MV/cm)$	0.3	0.4	3	5
$\nu_s \ (cm/s)$	$1.3 \mathrm{x} 10^{7}$	$1.2 x 10^{7}$	$0.8 x 10^{7}$	$3.0 \mathrm{x} 10^{7}$
$\Phi_k (W/cmK)$	1.3	0.6	4.5	1.3

higher critical electric fields will achieve the desired power rating within a smaller drift region; i.e. smaller device dimensions. Thus, the specific on-resistance of the devices will be lower for GaN and SiC compared to Si or GaAs [11]. In addition, the reduced switching losses and improved temperature handling will further permit higher switching frequencies and reduce the size and weight of many passive components within the circuit.

### **1.2** Figure of Merits

The main device characteristics used to evaluate power devices are breakdown voltage,  $V_{BV}$ , on-resistance,  $r_{on}$ , peak current density,  $I_{max}$ , substrate leakage,  $I_{off}$ , and material reliability. These electrical characteristics are all dependent on the material properties of the semiconductor. Figure of merits (FOM) were generated from similar systems relative to material properties to compare the potential transistor improvements between various semiconductors with respect to silicon. The relationship between basic material properties of the semiconductor and common device characteristics for transistors were used to create Johnson's figure of merit (JFOM), Baligas figure of merit (BFOM), Baliga's high frequency figure of merit (BHFFOM), Keyes figure of merit (KFOM) and the combined figure of merit (CFOM).

Johnson's figure of merit was formulated to compare the switching capabilities of semiconductors based on the potential cut-off frequency of an equivalent dimensioned device [12]. The trade off between cut-off frequency,  $f_T$ , and maximum applied voltage,  $V_m$ , were shown to be equivalent to the critical electric field and saturation velocity of the semiconductor shown in (1.1). The *JFOM* suggested that wider band gap materials would improve the switching characteristics over narrow band gap semiconductors (i.e. Si) which inherently have low critical electric fields.

$$JFOM = V_m f_T = \frac{E_C \nu_s}{2\pi} \tag{1.1}$$

Similarly, Baliga's (BFOM) comparison for power devices (1.2) was originated from the relationship of the transistors on-resistance and channel depletion width [13]. By applying the comparison of the breakdown voltage,  $V_B$  with respect to the specific on-resistance,  $r_{on}$ , of the drift region, the BFOM, compared the dielectric constant, carrier mobility and critical electric field of semiconductors for power devices. Like the JFOM, the BFOMstrongly suggested wide band gap materials were required to improve and advance the device characteristics over silicon based power devices. The evaluation of semiconductors for high frequency and high power applications (1.3) known as Baliga's high frequency figure of merit (BHFFOM) was created from the material properties which effected the on-resistance,  $r_{on}$ , and input capacitance,  $C_{in}$ , of the device [11]. The on-resistance was dependent on the critical electric field, mobility and dielectric constant and the input capacitance was based on the semiconductors dielectric constant and depletion width. Again, this comparison suggested that wide band gap materials were capable of withstanding large electric fields, while maintaining larger channel mobilities will be successful.

$$BFOM = \frac{V_B^2}{r_{on}} = \epsilon_r \mu_n E_C^3 \tag{1.2}$$

$$BHFFOM = \frac{1}{r_{on}C_{in}} = \mu_n E_C^2 \tag{1.3}$$

Devices that operate in high temperature applications are most effected by the degradation of carrier mobility and background doping concentration [14–16]. These effects were known to reduce the on-off ratio of the device, through reduced current density and increased leakage current. To account for thermal degradation of various semiconductors, the Keyes figure of merit (KFOM) for thermal management in integrated circuits, (1.4), was dictated by the thermal conductivity, saturation velocity and dielectric constant of the material. This figure of merit was originally used to compare the impact of high density circuitry in Si-based CMOS rather than high temperature environments in high power electronic applications. However, the results with respect to the materials of interest for thermal management were still applicable.

Lastly, the combined figure of merit (CFOM) evaluated the substrate quality for the potential in high temperature, high power and high frequency applications [17]. The *CFOM* (1.5) was created through the combination of *BHFFOM* and *KFOM*. A semiconductor with the capability of high frequency and high power applications must also have the potential to manage self heating from the device switching. For example, if a device operated at  $300^{\circ}C$  the intrinsic substrate background doping for a narrow band-gap (Si) and a wide band-gap (GaN) are  $6\times10^{14}$  cm<sup>-3</sup> and  $1\times10^3$  cm<sup>-3</sup> respectively. The small band gap material depletion width would significantly shrink due to the increased background doping leading to a higher leakage current.

$$KFOM = \frac{\Phi_k \nu_s}{\epsilon_r} \tag{1.4}$$

$$CFOM = \Phi_k \epsilon_r \mu_n \nu_s E_C^2 \tag{1.5}$$

The previously discussed figure of merits were tabulated and normalize to silicon in Table 1.2. Based on the physical properties of the semiconductors, GaN has the potential for a broad range of power applications compared to the other materials discussed. In addition,

Material	Si	GaAs	4H-SiC	GaN
JFOM	1	2	38	1479
BFOM	1	16	531	1206
BHFFOM	1	11	64	87
CFOM	1	11	33	164
KFOM	1	0.3	2.5	4.3
Overall	1	$10^{3}$	$10^{8}$	$10^{11}$

Table 1.2: Figure of Merits based on Material Properties for Power Electronic Devices

silicon carbide was also ranked well for the potential use in power devices. However, many fundamental issues of the material limited the advancement of this semiconductor over GaN for power devices. The most detrimental properties were the crystal orientation, mobility and band gap defects [18, 19]. In addition, the total consideration of all the figures of merits strongly suggested that GaN has the best possibility for next generation high power, high frequency and high temperature applications portrayed in Figure 1.2. Each material shown was only capable of operating within the triangular region defined by temperature, breakdown voltage and current density. As depicted below, GaN has the largest possible range of device operation for transistor applications compared to Si and GaAs.



Figure 1.2: A visual comparison of peak operation capabilities for Si, GaAs and GaN based on temperature and power applications. GaN has the ability to replace all silicon power devices based on the ability to operate  $200^{\circ}C$  higher and withstand electric fields larger than  $300 V/\mu m$ .

### 1.3 Gallium Nitride Transistors

Over the last decade, GaN based devices have been regularly discussed in literature regarding the device characteristics and reliability for high frequency and high power applications [20–41]. The three main design architecture for GaN field effect transistors, shown in Figure 1.3, are the heterojunction field effect transistor (HFET), metal oxide semiconductor heterojunction field effect transistor (MOS-HFET) and the metal oxide semiconductor field effect transistor (MOSFET). The devices look similar in shape and metalization design; however, the gate film stacks along the channel are significantly different for each transistor. The HFET and MOS-HFET both have a 2-dimension electron gas (2DEG) formed along the AlGaN/GaN or AlN/GaN polarized interface [42]. The major difference is the HFET uses a Schottky metal at the gate, whereas the MOS-HFET has a gate dielectric. The Schottky gate has many issues, such as depletion-mode operation and higher gate leakage [43]; however, the HFET is currently capable of producing high current densities for a given gate length. The MOS-HFET has a similar architecture as the HFET, but the gate dielectric has been included to reduce the gate leakage current during operation [43]. Fluorine plasma pretreatments along the GaN channel surface was a common technique to shift the devices turn-on voltage positive. This pretreatment to the GaN adjusted the Fermi level under the channel, requiring a positive gate bias to form the 2DEG channel rather than a negative bias to turn the device off [44]. This has improved the sub-threshold



Figure 1.3: The three most common GaN device architectures under investigation for power applications are the (a) HFET, (b) MOS-HFET and (c) MOSFET.

leakage current; however, the performance of the device was degraded due to higher sheet resistance and lower channel mobility due to fluorine induced interface defects [45].

Lastly, the metal oxide semiconductor field effect transistor (MOSFET) architecture only has a dielectric along the gate [14, 15]. By removing the AlN or AlGaN film and the inherent 2DEG along the interface, the turn-on voltage of the device becomes strictly positive. This produced an enhancement-mode transistor where the device was normally-off at zero gate bias. Various gate dielectrics for GaN MOSFETs have been investigated to rectify issues that have reduced the capacitance density and channel field effect mobility. In-turn an improved dielectric and interface along the channel has the potential to advance the peak current density and cutoff frequency of the device.

The majority of the scientific research has been aimed towards improving the off-state of the HFET design for high power applications. The following Table 1.3 highlights significant

	T	T	TZ.	Т	~	I.		
	$L_G$	$L_{GD}$	$V_t$	$I_{DS}$	$g_m$	$V_{BV}$	$r_{on}$	Ref.
	$(\mu m)$	$(\mu m)$	(+/-)	(mA/mm)	(mS/mm)	(V)	$(m\Omega cm^2)$	[2.0]
HFET	4	3	-	10	28	30*	6.6	[20]
	1.5	10	-	800	160	594*	3.3	[21]
	2	4	+	130	70	300	30	[22]
	2	125	-	150	50	8300*	186	[23]
	2	7.5	+	200	70	800	2.6	[24]
	1.5	12	+	275	150	1100	6.2	[25]
	1.5	12	+	380	246	1100*	/	[26]
	3	20	-	463	102	1160	7.5	[27]
	0.2	1.3	-	1300	470	/	/	[28]
MOSHFET	2	15	+	300	180	600	9.3	[29]
	2	2	+	326	130	274	0.7	[30]
	2	6	-	770	142	/	/	[31]
	2	6	-	640	147	/	/	[31]
	2	6	-	140	140	/	/	[31]
	1.8	2	-	650	160	130*	0.2	[32]
	0.12	0.2	+	1150	510	8	/	[33]
	1	5	-	0.15	/	/	200	[34]
	0.07	0.04	+	740	250	12	/	[35]
	5	10	+	225	136	385	/	[36]
MOSFET	20	/	+	25	2	/	/	[37]
	4	16	+	/	/	730	34	[38]
	1	16	+	10	4	/	/	[39]
	4	16	+	4	4	/	/	[39]
	0.16	10	+	530	100	565	2	[40]
	0.25	5	+	263	72	39	/	[41]
* indicates $V_{GS} < 0$ V when breakdown voltage was measured								

Table 1.3: GaN Device Characteristics for Different Design Architectures



Figure 1.4: Peak current density with respect to gate length for the GaN HFET, MOS-HFET and MOSFET.

devices found in literature with their corresponding device characteristics that are important for high power applications. The main advantage current MOS-HFET technology has over the GaN MOSFET are large current densities for a given gate length, shown in Figure 1.4. However, roughly half of the devices are depletion mode devices, whereas all of the GaN MOSFETs are normally-off enhancement-mode transistors. The MOSFET architectures inherent off-state behavior is advantageous over techniques required to shift the threshold voltage of a depletion mode device [46].

The peak current density trend-line as the gate lengths are reduced vary for the HFET and MOSFET architecture. This is mostly due to the HFET behaving as a junction field effect transistor (JFET) rather than a MOSFET. The major difference of the device is that at zero gate bias the JFET has a 2DEG formed, whereas the MOSFET has a depleted channel with virtually no carriers. The JFET (1.6) and the MOSFET (1.7) have significantly different expressions of the saturation current density where the mobility ( $\mu_n$ ), electron charge (q), device width (W), gate length (L), 2DEG ( $n_s$ ), source resistance ( $R_s$ ), voltage L ( $V_L$ ), capacitance density ( $C_{ox}$ ), and gate-threshold voltage ( $V_{GT} = V_G - V_{th}$ ) were used.

$$I_{sat_{HFET}} = \frac{\frac{qn_s W\mu_n}{L} V_{GT}}{1 + \frac{qn_s W\mu_n}{L} R_s + \sqrt{1 + 2\frac{qn_s W\mu_n}{L} R_s + \left(\frac{V_{GT}}{V_L}\right)^2}}$$
(1.6)



Figure 1.5: HFET and MOSFET peak current density with respect to gate length based on GaN properties. Improving the channel mobility and capacitance density of current GaN MOSFET technology has the potential to surpass HFET characteristics below 2  $\mu m$  gate lengths.

$$I_{sat_{MOSFET}} = \frac{\mu_n C_{ox} W}{2L} V_{GT}^2 \tag{1.7}$$

The HFET has significantly larger current densities than the MOSFET at large gate lengths (> 10  $\mu m$ ) due to the JFET behavior of the device and large number of carriers under the channel. However, current fabricated GaN MOSFET devices have comparable saturation current density as the HFET when the gate lengths are reduced below 100 nm, shown in Figure 1.5. Furthermore, through an improved gate dielectric and interface, the channel mobility and capacitance density of the GaN MOSFET has the potential to improve devices saturation current by minimally 10x. This could be achieved by improving the dielectric capacitance from  $0.5 \ \mu F/cm^2$  to  $2.5 \ \mu F/cm^2$  and the saturation channel mobility from  $100 \ cm^2/Vs$  to  $200 \ cm^2/Vs$ . This improvement along the channel has the potential to compete with the HFET saturation current density around a gate length of  $2 \ \mu m$ . Through improved material properties of the device, the GaN MOSFET has the potential to compete with the GaN HFET for current density for power applications.

In the following chapters, the design advantages of a GaN MOSFET over a GaN HFET were examined along the with characterization of gate dielectrics and the fabrication of an enhancement-mode GaN MOSFET for the potential use in power electronics.

### 1.4 Synopsis of Thesis

Current GaN based devices have failed to reach the full potential of the material due to the inherent behavior of the HFET design and unsolved material engineering for the MOSFET. Through development at the University of Alberta, ultra-thin films grown with atomic layer deposition on GaN make it possible to fabricate high quality gate oxides. This thesis is focused on the design, fabrication and characterization of a novel enhancement-mode GaN MOSFET for power applications. Through the unique design this device attempts to improve the current density and breakdown characteristics for GaN transistors.

The following chapter examines the potential GaN MOSFETs have as a power transistor through simulations and models to predict the device characteristics given near ideal materials. Through simulation, the MOSFET design was optimized and compared to a modeled fabricated GaN HFET. An in-depth study was conducted to determine the underlying advances the GaN MOSFET has over HFETs for power devices.

Limitations that separate the simulations of GaN MOSFETs from fabricated devices are the non-ideal material complications and techniques used to fabricate the device. Current fabrication methods for GaN HFET technology will not guarantee a fully functional MOSFET due to high temperature degradation of the gate dielectric. Chapter 3 examines the material and technology requirements for an enhancement-mode GaN MOSFET to be fabricated with minimal processing issues. With novel materials and deposition techniques, the GaN MOSFET can be realized for power electronics.

Chapter 4 contains the investigation of low temperature high- $\kappa$  rare earth metal oxides on GaN for the MOS gate dielectric. The electrical characteristics of  $ZrO_2$  and  $HfO_2$  MOS capacitors on GaN were examined to evaluate the quality of the interface and charge control of the GaN channel. The low temperature plasma enhanced deposition technique was shown to significantly improve the capacitance density, gate leakage, field effect mobility and charge control within the channel over previous attempts of dielectrics shown in literature.

For transistors to operate correctly the source and drain must provide enough carriers to ensure that device saturation occurs. Chapter 5 examines the materials used to create high carrier concentrations in select regions of the device. Novel low temperature AlN and  $ZrN_x$  films were fabricated and characterized on GaN templates. Through electrical characterization the contact resistance, 2 dimensional electron gas, sheet resistance, and interface mobility of the films were examined. To ensure device longevity, the reliability of the films proposed were examined for potential device lifetimes.

Chapter 6 covers the fabrication methods used to build the GaN MOSFET through the use of low temperature plasma-enhanced atomic layer deposition of  $ZrO_2$  and AlN. The electrical characterizations of the MOSFET examine the effect of dimensions, thicknesses and materials. The transistors design architecture and device dimensions were examined to

improve the maximum drain voltage during the off-state.

The final chapter summarizes the overall results of the novel GaN MOSFET fabricated and characterized in this report. The novel GaN MOSFET was demonstrated for the potential to provide a novel normally-off transistor and advance power transistor characteristics. This chapter will also provide future insight into work related to this project and potential MOS projects worth pursuing with the novel rare earth metal oxides.

# Chapter 2 Device Design and Simulation

### 2.1 Introduction

The promising material properties that GaN has to offer as a semiconductor device for high power switching applications are only possible when the architectural design of the device has been utilized to its fullest extent. Before the fabrication of a device began, a systematic optimization of dimensions, film thicknesses and design features (Figure 2.1) were required to demonstrate an improved device performance over current technology. Technology computer aided design (TCAD) simulations have allowed semiconductor devices to be optimization with relatively no time or cost when compared to full fabrication processes. Many properties of the MOSFET device, such as electron, current and electric field strength distributions, require simulations to extract these behaviors since these results are not capable of being physically measured. Following the design and optimization of the GaN MOSFET, a detailed comparison to a modeled GaN HFET was investigated to examine the underlying design advantages the GaN MOSFET has over the HFET.

The peak current density and off-state breakdown voltage are two main device characteristics which are important for high voltage and high power applications in GaN FETs [11]. The microelectronic device simulation software, Synopsis Sentaurus TCAD, was used to simultaneously solve the Poisson equation along with the continuity of carriers and the Schrodinger equation in defined regions and domains of the device [47]. The coupled equations were solved with finite element analysis methods which included the material properties such as electron mobility, effective electron mass, dielectric constant, and band gap [48]. To predict the drain-source breakdown voltage of the device, avalanche generation models



Figure 2.1: The GaN MOSFET design and dimensions to be optimized for improved device characteristics. The device architecture was optimized before field plates were included which were used to improve the breakdown voltage.

were included based on well-known impact ionization parameters for each material [49,50]. These semiconductor material dependent parameters determine whether a carrier has the required energy to excite another carrier into the conduction band through evaluation of material space charge and mean free path within a collision [51,52]. The generation of an electron has the potential for a repetitive multiplication process, known as the avalanche effect, to be catastrophic for electronic devices [53,54]. Through systematic simulations, the architecture of the device was manipulated to minimize the effects of early device break-down.

### 2.2 Device Optimization

To optimize the GaN MOSFET multiple lengths and dimensions were considered to produce an enhancement-mode high power transistor. Similar to silicon technology, the substrate thickness was fixed to be less than 1/3 the gate length of the device [55]. Based on the desired gate length and body thickness, the maximum doping concentration of the substrate was determined, as shown in Figure 2.2. The voltage dependent depletion width under the MOS channel (2.1) was obtained from the MOS surface depletion effect described in many solid state device physics textbooks [14, 15, 56]. The gate voltage was defined as the sum of flatband voltage,  $V_{FB}$ , the semiconductor band bending surface voltage drop,  $\phi_s$ , and voltage drop across the dielectrics,  $V_{ox}$  [56]. The flatband voltage (2.2) was defined by the work function offset between the gate metal and the semiconductor which was dependent on



Figure 2.2: The maximum substrate thickness of the GaN and equivalent minimum gate length were determined based on the depletion width determined from the substrate doping concentration. Any combination of thickness and substrate doping that remains below the curve will produce a fully depleted channel.

the gate work function,  $\psi_m$ , semiconductor electron affinity,  $\chi_s$ , doping concentration,  $N_D$ , and effective conduction band density of states,  $N_C$  [14,15]. The depletion width extracted from (2.1) was used to relate the doping concentration to substrate deletion width at zero gate bias as shown in (2.3).

$$V_G = V_{FB} + \phi_s + V_{ox} = V_{FB} + \frac{qN_D W_{DEP}^2}{2\epsilon_s \epsilon_o} + \frac{qN_D W_{DEP}}{C_{ox}}$$
(2.1)

$$V_{FB} = \psi_m - \chi_s - k_B T \ln\left(\frac{N_D}{N_C}\right) \tag{2.2}$$

$$W_{DEP} = \sqrt{\left(\frac{\epsilon_s \epsilon_o}{C_{ox}}\right)^2 + \frac{2\epsilon_s \epsilon_o}{qN_D} \left[\psi_m - \chi_s - k_B T \ln\left(\frac{N_D}{N_C}\right)\right]} - \frac{\epsilon_s \epsilon_o}{C_{ox}}$$
(2.3)

To determine the approximate substrate thickness and doping concentration, initial material properties of the gate were assumed to be comparable to Platinum ( $\psi_m$ =5.3 eV) [57]. However, if the gate metal had a smaller work function, such as Nickel ( $\psi_m$ =4.6 eV) [57], a lower doping concentration would provide an equivalent depletion width [14]. Furthermore, the dielectric capacitance density was conservatively estimated as an ultra-thin (6 nm)  $HfO_2$  film. Based on an anticipated relative dielectric constant,  $\epsilon_r=24$ , and dielectric thickness, the corresponding oxide capacitance density (2.4) was initially approximated as  $3.5\mu F/cm^2$ . Unlike the gate metal work function, the depletion width was not significantly affected by the dielectric capacitance density for substrates with doping concentrations below  $5 \times 10^{15} cm^{-3}$ . The channel of the device was not fully depleted when substrates were thicker than the respective depletion width (above the line in Figure 2.2), which led to larger off-state leakage current. A maximum substrate doping concentration obtained from (2.3) ensured that the depletion width was always larger than the GaN thickness to ensure a fully depleted device.

$$C_{ox} = \frac{\epsilon_r \epsilon_o}{t_{ox}} \tag{2.4}$$

To optimize the gate length, shown in Figure 2.3, the corresponding body thickness and doping concentrations were adjusted to ensure consistent body depletion discussed earlier. The gate-source and gate-drain lengths were simulated with lengths of  $1\mu$ m and  $12\mu$ m, respectively for all gate lengths. As predicted through parallel-plate breakdown



Figure 2.3: The simulated device breakdown voltage and peak current density based on gate length. Through optimization of the gate length the power rating was largest at a gate dimension of  $1.2\mu$ m.

theory (2.5), a larger fully depleted gate channel was able to assist the overall breakdown voltage of the device [11]. This agreed with simulation; however, the breakdown voltage saturated after a gate length of  $1\mu$ m. Beyond this gate length the breakdown voltage was more dependent on the gate-drain separation. Conversely, the MOSFET current density described in (2.6) was used to confirm that the current degraded as the gate length was increased [55,56]. The optimized gate length was determined when the breakdown voltage began to saturate and the current density was not critically degraded. The optimized gate length (1.2 $\mu$ m) was determined when the peak power rating of the device ( $P = V_{BV} * I_{DS}$ ) was maximized based on a large breakdown voltage and peak current density.

$$V_{BV} = \frac{qN_D L_G^2}{2\epsilon_s \epsilon_o} \tag{2.5}$$

$$I_{DS} = \frac{W}{2L_G} C_{ox} \mu_n \left( V_{GS} - V_t \right)^2$$
(2.6)

Following the optimization of the gate length, the gate-source separation length simulations were swept from 100nm to  $5\mu$ m. Again, the breakdown voltage and current density,



Figure 2.4: The gate-source length dependence on breakdown voltage and current density for the GaN MOSFET. The gate-source separation was optimized to a peak power rating of  $0.7\mu$ m.
shown in Figure 2.4, were compared to determine the peak overall power rating of the device. As the gate-source length was increased, the breakdown voltage of the device improved until saturation began around  $1\mu$ m. Conversely, as the breakdown voltage improved the current density decreased which created a peak maximum power rating based on the source-gate length. The peak power rating was maximized when the gate-source length was  $0.7\mu$ m. Any gate-source separation larger than the optimized length not only reduced the overall power rating but increased the total length of the device. The larger device length decreased the overall number of die for a given template which was undesired for potential fabrication purposes.

Following the optimization of the GaN MOSFET for high power applications, the addition of novel architectures were examined to further improve the device power rating (Figure 2.5). Through the addition of field plates attached to the source and gate with silicon nitride encapsulation, it was shown that the breakdown voltage of the device increased without significant degradation of the transistors device characteristics [58–60]. First the field plate length was optimized by holding the height constant at  $3\mu$ m. The breakdown voltage was improved as the field plate length was stretched longer; however, the parasitic capacitance become significantly large. The increased gate-source capacitance  $(C_{GS})$  prevents a higher switching capability (2.7) known as the cutoff frequency [61]. The optimal field plate length was determined by selecting the largest breakdown voltage where the parasitic capacitance was minimal. With the breakdown voltage no longer improving beyond  $3\mu$ m, only the parasitic capacitance increased for any longer field plates. Thus, the optimized source field plate length placed along the drain region was determine to be 1/3 of the gate-drain separation. Similar methods used to determine the length of the field plate were conducted to optimize the height of the source field plate to improve the breakdown voltage of the device during off-state operation. The field plate height was lowered from  $10\mu m$  to 500nm. At  $10\mu m$  the



Figure 2.5: Optimization of the source field plate based on length and height.

field plated behaved in a similar manor as the device without a field plate. Below the field plate height of  $1.3\mu$ m would result in a higher parasitic capacitance without a significant increase in breakdown voltage. Similar methods used to optimize the source field plate were used to optimize the gate field plate. The gate field plate was found to be most effective at a height of 300nm and a length set to 1/2 the overlap of the source field plate (ie  $1.5\mu$ m for a  $L_{GD}=9\mu$ m). Utilizing both the source and gate to suppress the electric fields along the drain at the ends of the field plates, the breakdown voltage of the device was significantly improved [62].

$$f_T = \frac{g_m}{2\pi C_{GS}} \tag{2.7}$$

The overall breakdown voltage, shown in Figure 2.6, compares different gate-drain (drift region) lengths of the optimized GaN MOSFET with and without field plates to one of the leading field plated GaN HFET. The GaN HFET exceeded the GaN MOSFET without field plates for all drift region lengths. However, as the field plates were included to the MOSFET design, the MOSFET had the capability to withstand the higher electric fields. Through TCAD simulation of the GaN transistors, an in-depth understanding of the device behavior indicated how the MOSFET produced larger breakdown voltages then the HFET.



Figure 2.6: The improved breakdown voltage with respect to drift region of the GaN MOS-FET with field plates compared to current HFET technology.

### 2.3 Device Comparison

To understand the underlying design advantages the MOSFET has over the HFET, similar dimensioned GaN devices, shown in Table 2.1, were simulated. Both devices utilized the optimized field plates for both the source and gate. First the conventional device characteristics such as the family of curves  $(I_{DS}-V_{DS})$ , transfer curve  $(I_{DS}-V_{GS})$ , and the breakdown voltage were examined. Following investigation of the device characteristics, the internal material properties such as mobility, carrier concentration, and electric field distribution produced from the device architecture were examined for important modes of device operation. Simulations were required to extract the electron distribution and electric field due to the inability of a direct electrical measurement of these characteristics.

Dimension	GaN MOSFET	GaN HFET
Gate Length $(L_G)$	$1.5 \ \mu m$	$1.5 \ \mu m$
Drain Length $(L_{GD})$	$8 \ \mu m$	$8 \ \mu m$
Source Length $(L_{GS})$	$0.7 \ \mu m$	$0.7 \ \mu m$
Body Thickness $(t_{GaN})$	$0.4 \ \mu m$	$0.4 \ \mu m$
AlN Thickness $(t_{AlN})$	N/A	2  nm
Oxide Thickness $(t_{ox})$	10 nm	3  nm
Source FP Height $(H_{S_{FP}})$	$1.3 \ \mu m$	$1.3 \ \mu m$
Source FP Length $(L_{S_{FP}})$	$2.6 \ \mu m$	$2.6 \ \mu m$
Gate FP Height $(H_{G_{FP}})$	$0.2 \ \mu m$	$0.2 \ \mu m$
Gate FP Length $(L_{G_{FP}})$	$1.3 \ \mu m$	$1.3 \ \mu m$

Table 2.1: Optimized GaN MOSFET and HFET Dimensions

The results of the family of curve and transfer curve for both the GaN MOSFET and GaN HFET, shown in Figure 2.7, were simulated for a gate length of  $1.5\mu$ m. The gate length of the MOSFET was increased from the optimized value to match the simulated GaN HFET which replicated one of the leading fabricated devices [25]. The architectural difference between the MOSFET and HFET designs and the material layers and thicknesses along the gate produced different turn-on voltages. The polarized channel in the GaN HFET required a gate voltage of 0.5V or larger to produce conduction whereas the MOSFET required 1.6 V. However, comparing the drain current with respect to the applied electric field in the channel, it was shown that both devices behave similarly. Regardless of the architecture, the electric field dependent characteristics were nearly identical.

During off-state operation (i.e.  $V_{GS} = 0$  V), the drain bias was increased to determine the breakdown voltage of the devices. The current density with respect to voltage was simulated (Figure 2.7c) to determine the leakage current. Once the current density surpassed 80



Figure 2.7: (a) Simulated GaN MOSFET family of curves. (b) Voltage and electric field dependence transfer characteristics for the simulate MOSFET and HFET. (c) Breakdown voltage of the simulated HFET (600 V) and MOSFET (3500 V) with similar sized dimensions.

 $\mu$ A/mm, the device was considered to have too large of a leakage current to be considered off. This voltage was considered to be the breakdown voltage of the device. With both devices simulated with equivalent gate and drain lengths, the GaN MOSFET (3500 V) was found to potentially produce a breakdown voltage five times larger than the Gan HFET (600 V). The MOSFET was able to produce similar current densities and withstand larger drain voltages than the HFET. To determine the underlying behavior between devices, the carrier mobility, electron distribution and electric field distribution during on and off states were examined.



Figure 2.8: The effective electron channel mobility was extracted from the simulated family of curves for the GaN MOSFET and GaN HFET.

From the family of curves simulations the effective field effect mobility of the electrons, shown in Figure 2.8, was extracted by using the conventional experimental technique applied to the simulated data [16]. These results were in agreement with measured MOS mobility results of oxides on GaN [63–65]. The GaN HFET was found to produce a larger peak mobility than the MOSFET during low electric fields. However, the higher electron mobility during sub-threshold operation does not simply imply an improved device. As both devices were operating significantly beyond turn-on voltage, the field effect mobility was equivalent. Thus, the total current was roughly equal during this regime of device operation.

During on-state operation, the MOSFET and HFET produced equivalent field effect mobility and current densities. By including quantized models, which include the quantum mechanical effects, the electron distribution beneath the channel were extracted from the simulations. Classically, the peak carrier distribution was assumed to travel along the interface, whereas the quantum mechanics showed that the dielectric layers cause the peak carriers to travel a few nanometers within the GaN substrate. The carrier transport electron distribution profiles under the channel at the source edge, middle and drain edge were compared between the MOSFET and HFET for different modes of operation, shown in Figure 2.9.

When the gate voltage was below threshold voltage for both devices, the GaN HFET had  $10^4$  more carriers under the channel than the MOSFET. For a regime where both devices should be off, the MOSFET has a lower number of carriers within the channel compared to the HFET. The HFET was unable to maintain a low concentration due to the 2DEG channel formed from the polarized AlN/GaN interface. During sub-threshold gate biasing, the MOSFET completely depleted the channel; whereas the HFET was not as effective



Figure 2.9: The electron density during various modes of operation under the gate of the MOSFET and HFET. The electron distribution in the channel was examined near the source edge, middle and drain edge. (Top) The device is off, i.e.  $V_{GS} < V_t$ , (Middle)  $I_{DS} = 40 \ mA/mm$ , (Bottom)  $I_{DS} = 175 \ mA/mm$ .

which led to more carriers within the gate region.

Electron profiles of both devices during on-state operation (40 mA/mm and 175 mA/mm) are shown in Figure 2.9. These peak electron distributions are highest under the channel on the source side and shrink relatively linearly to the gate-drain edge. These simulations were consistent with the channel pinch off and channel length shortening effects found in enhancement mode devices. Again, this confirmed that the GaN MOSFET and GaN HFET both behaved as enhancement mode devices [14, 15]. For a given equivalent drain current density, both devices exhibit similar carrier densities under the channel even though the HFET produced slightly high peak density. However, the MOSFET channel was found to have a broader carrier distribution which produced an equivalent current density as the HFET.



Figure 2.10: The electric field under the gate and drain region when  $(Left) V_{DS}=100V$ and  $(Right) V_{DS}=1600V$ . (Bottom) The electron density for both devices under the gate and drift region at  $V_{DS}=1600V$ . The MOSFET suppressed the electric field and maintain a low concentration of carriers under the gate whereas the HFET was unable to control the channel depletion.

Though both devices produced similar family of curves under low drain bias, the two devices' architectures did not exhibit similar breakdown voltages. The early breakdown effect of the HFET was examined through the simulations that included the electric field and carrier density as the devices before and after breakdown voltage (Figure 2.10). The source and gate field plates were able to manage the electric field along the drain and suppress the breakdown effect. However, once breakdown was initiated, the electric field was too large for the field plate structures to suppress the avalanche effect. Though both devices has the same optimized field plate structures, the GaN MOSFET was able to maintain a breakdown five times larger. The inherent design architecture and physical nature of the GaN HFET caused an earlier breakdown than the GaN MOSFET due to channel control under the gate. The electron density along the device from gate to drain, shown in Figure 2.10, depicts the low carrier concentration in the MOSFET at  $1/2 V_{VB}^{MOSFET}$ ; whereas, the HFET was beyond breakdown and was unable to prevent current from conducting during post breakdown at 3  $V_{VB}^{HFET}$ . By using the MOSFET architecture, the channel region truly depletes the entire body correctly, allowing for a complete off-state. Conversely, the HFET did not remove all of the carriers within the channel when the device was off at zero bias. The induced polarization along the interface between the AlN/GaN created a 2DEG making the minimum carrier concentration much higher than the fully depleted GaN MOSFET. Thus, given the similar device dimensions as a GaN HFET, the MOSFET should potentially surpass the breakdown voltage while producing similar on-state characteristics.

### 2.4 Summary

Through calibrated models and simulations, the GaN MOSFET was optimized for high power applications. With multiple variables required, a systematic process was conducted to determine the lengths and thicknesses of the architecture. Once the optimal device was created, the MOSFET was compared to a simulated HFET. Through simulation, the internal operations of the devices were examined. It was shown that both devices obtained equivalent carrier concentrations within the channel. Though the peak concentration and distribution varied slightly, the overall current density was comparable. During sub-threshold gate voltages, the GaN MOSFET was capable of completely depleting the channel and removing almost all of the electrons. Conversely, the polarization effects in the HFET hindered the ability for the HFET to remove as many carriers as the MOSFET. Thus, the minimum carrier concentration of both devices became an important consideration when attempting to prevent early breakdown effects within power devices. The simulations confirmed the GaN MOSFET was more proficient at maintaining a lower minimum carrier concentration over the GaN HFET; thus a higher standoff voltage was capable (5x breakdown voltage). Through the use of the MOSFET, GaN has the potential to surpass current GaN based devices for high power applications. The requirement for improved materials within the device has prevented the GaN MOSFET from approaching the maximum potential of the design. The following chapters examine improved fabrication techniques and subsequent material characteristics which were incorporated into the GaN MOSFET.

# Chapter 3

# **Fabrication Techniques**

### 3.1 Introduction

Semiconductor device simulations predicted that the GaN MOSFET was advantageous over the commonly used AlGaN/GaN or AlN/GaN HFET designs for power applications. The inherent properties of a MOSFET provides the potential for higher device switching frequency, improved charge control within the channel, reduced leakage current during off-state operation and higher breakdown voltage. The following diagram, Figure 3.1, depicts some of the most important regions of the device where detrimental fabrication issues are most likely to occur. The following listed device regions are considered the six most important components required to produce an effective enhancement-mode transistor.

1. Gate Dielectric

- The GaN/dielectric interface must produce an unpinned, low interface trap density and a large peak capacitance to ensure optimal charge control within the channel.

2. Source/Drain Doping

- The source and drain doped regions must produce a high number of carriers whilst maintaining a large sheet mobility to reduce parasitic resistances.

3. Source/Drain Contacts

- The metal contacts used for the source and drain must produce a low contact resistance to the GaN.

#### 4. Gate Metal

- To produce an enhancement-mode device, the gate metal must create a positive turn-on voltage to prevent conduction at zero gate voltage.

#### 5. GaN Substrate

- The substrate material must have low density of dislocations and defects to ensure a high quality film for high breakdown applications.

6. Device Isolation

- The device must be unable to conduct current from the source to drain without traveling under the gate.

To fabricate a fully-functioning enhancement-mode GaN MOSFET, novel films and fabrication techniques are required to provide higher quality interfaces. The gate dielectric and metal stack are the most important region of the MOSFET because the dielectric/GaN interface forms the channel and dictates the turn-on voltage and current density of the device. The number of carriers provided along the source and drain are crucial to ensure enough carriers are able to inject into the channel. Furthermore, for a true off-state to occur, the defects within the bulk are desired to be obsolete to prevent uncontrolled conduction around the gate. The ohmic contacts and device isolation are important; however these regions do not dictate whether the transistor is an enhancement-mode or depletion-mode device. To reduce the fabrication cost and processing time, these regions of the device were not optimized.



Figure 3.1: Important regions and materials of the GaN MOSFET which require novel fabrication techniques for improved device performance.

### 3.2 Gate Dielectrics for GaN

High gate leakage is one of the current limitations that prevent GaN based transistors from high power applications. MOSFETs require low leakage current to ensure low noise and improved reliability along the channel. Furthermore, high frequency dispersion is related to the charging of surface states near the drain edge of the gate. These traps do not respond quickly to the high frequency signal thereby affecting the high frequency performance.

To reduce this effect, the interface between the gate dielectric and semiconductor must have a low density of interface states (interface traps) and high capacitance density, shown in Figure 3.2. The improved interface will in turn increase the carrier channel mobility of the device, thus further improving the current density and cut-off frequency of the device. Likewise, a large capacitance density gate dielectric will further improve the charge control under the gate region.

Over the past decade, the attempts to produce a low  $D_{it}$  and high  $C_{ox}$  on GaN for MOS applications, (Table 3.1), have barely reached the  $1\mu F/cm^2$  and  $1 \times 10^{11} eV^{-1} cm^{-2}$  [66–80]. The collection of these films were deposited by chemical vapour deposition (CVD),



Figure 3.2: Density of interface traps and capacitance density of dielectric on GaN. These techniques were unable to obtain the high charge control and low interface traps required for a MOSFET.

Film	Growth	Film	$C_{ox}$	$D_{it}$	Year
Type	Technique	Thickness	$(\mu F/cm^2)$	$(eV^{-1}cm^{-2})$	
$SiO_2$	PECVD	56  nm	0.06	$8.9 \mathrm{x} 10^{11}$	1996
$TiO_2$	MOCVD	$19  \mathrm{nm}$	0.62	$5.0 \mathrm{x} 10^{11}$	1997
$Ta_2O_5$	RF-Sputter	$91 \mathrm{nm}$	0.05	$4.1 \mathrm{x} 10^{11}$	2000
$SiO_2/SiN$	JVD	20/20  nm	0.14	$5.0 \mathrm{x} 10^{11}$	2001
$Sc_2O_3$	MBE	$10 \ \mathrm{nm}$	0.40	$1.2 \mathrm{x} 10^{12}$	2001
$Sc_2O_3$	MOCVD	48  nm	0.31	$5.0 \mathrm{x} 10^{11}$	2002
MgO	MBE	80  nm	0.47	$2.0 \mathrm{x} 10^{11}$	2002
$SiO_2/Ga_2O_3$	Sputter/DO	$100/15 \ \mathrm{nm}$	0.03	$3.9 \mathrm{x} 10^{10}$	2003
$SiO_2$	Sputter	100  nm	0.03	$4.5 \mathrm{x} 10^{12}$	2003
$Al_2O_3$	ALD	$16 \ \mathrm{nm}$	0.20	$6.0 \mathrm{x} 10^{12}$	2005
$HfO_2$	ALD	$9 \mathrm{~nm}$	1.10	$2.0 \mathrm{x} 10^{11}$	2007
$HfO_2$	MOCVD	$13 \mathrm{~nm}$	0.36	$1.5 \mathrm{x} 10^{12}$	2007
$ZrO_2$	MOCVD	$13 \mathrm{nm}$	0.36	$1.0 \mathrm{x} 10^{12}$	2007
$Ta_2O_5$	ALD	$6 \mathrm{nm}$	0.87	$8.0 \mathrm{x} 10^{11}$	2011
$HfO_2$	ALD	6  nm	0.74	$1.4 \mathrm{x} 10^{12}$	2011

Table 3.1: Gate Dielectrics for GaN Applications

plasma-enhance CVD (PECVD), metal-organic chemical vapour deposition (MOCVD), jet vapour deposition (JVD), molecular beam epitaxy (MBE), dry oxidation (DO) and atomic layer deposition (ALD) with various surface pretreatments. One of the most promising pretreatments for GaN has been a low temperature oxygen-plasma in conjunction with PEALD [65,81].

### **3.3** Source/Drain Carriers

Supplying carriers to the channel of a MOSFET is crucial for high current densities when the device is operating at the maximum gate voltage. If the source and drain regions of the device do not have sufficient number of carriers the device will lack conduction and gate control, typically observed in the transfer characteristics of the device. Conventional MOSFET devices use ion implantation to define the channel, source and drain regions of the device. Through heavily doping the source and drain regions the assumption is that the channel has an ample amount of carriers.

One of the major issues with doping GaN is the large activation energy ( $\sim 150 \text{ meV}$ ) required for the p-type dopant magnesium (Mg) compared to ( $\sim 20 \text{ meV}$ ) for the n-type dopant silicon. The large activation energy produces a large temperature variation between 300 K and 500 K operation. The most common technique to produce a dopant within



Figure 3.3: Polarized 2DEG carriers and corresponding sheet mobility for AlN and AlGaN films.

semiconductors is ion implantation. The main complications associated with this technique and GaN are high post anneal to activate the dopant. This technique requires a high nitrogen over-pressured environment to ensure that the nitrogen within the semiconductor does not off gas.

Polarized films, AlGaN and AlN, have been used to create high carrier densities with large electron mobilities along the GaN interface. The conventional technique to produce AlN films is with high temperate techniques such as MBE and metal organic vapor phase epitaxy (MOVPE) to produce high carrier densities, shown in Figure 3.3 [82–89]. However, the carrier mobility of the 2DEG is dependent on the quality of the films interface. Novel low temperature PEALD techniques have been investigated to produce high carrier density films on GaN [34, 90]. This allowed for the selective deposition of AlN through PEALD photoresist lift-off. Through the integration of low temperature AlN, novel fabrication techniques will avoid the requirement of etching.

### 3.4 GaN Metalization

To ensure low loss and high conduction, the contact resistance of the source and drain must be as low as possible. Most commonly used metals on GaN form Schottky barrier contacts, due to the high electron affinity of GaN compared to many metals such a Al, Cr, Ni and Pt. Through empirical studies, the metal stack of Ti/Al/Ni/Au has formed low contact resistance with GaN through high temperature annealing, shown in Figure 3.4. Alternate methods such as tunneling contacts have been studied to reduce the high processing temperatures and cost of complex metal stacks [91]. The incorporation of tunneling contacts significantly reduced the required annealing temperatures for low resistive contacts on GaN [82–84,92–95]. Through a lower annealing temperature the overall thermal budget of the device can be reduced to prevent gate oxide degradation.



Figure 3.4: Contact resistance with respect to annealing temperatures to produce ohmic contacts on GaN. The current techniques require high annealing temperatures and complex metal stacks.

Contact	AlN	Temperature	Time	$R_c$
Metal	Film	$(^{o}C)$	(sec)	$(\Omega \cdot mm)$
	3.0 nm	700	120	0.46
	1.7  nm	800	30	0.50
	3.0  nm	800	30	0.63
	4.5  nm	800	30	0.70
Ti/Al/Ni/Au	5.5  nm	800	30	0.90
, , , ,	3.0  nm	700	30	3.38
	3.0  nm	800	30	0.31
	_	750	30	2.00
	1.0  nm	750	30	0.20
	2.5 nm	600	30	2.50
	2.5  nm	700	30	1.00
	2.5  nm	800	30	0.20
Ti/Al/Mo/Au	2.5  nm	850	30	0.25
	2.5  nm	900	30	0.30
	2.5  nm	1000	30	0.50
	30  nm	550	60	2.20
	30  nm	625	60	0.48
	30  nm	650	60	0.28
	30  nm	700	60	0.15
Ti/Al/Ni/Au	30  nm	750	60	0.20
	30  nm	800	60	0.20
	30  nm	850	60	0.29
	30  nm	900	60	0.39
	30  nm	800	60	0.30
Al	—	500	30	1200
Au	_	500	30	600
Ti/Au	_	500	30	120
Ti/Au	_	700	20	12.0
Ti/Al	_	700	20	1.20
Ti/Al	—	900	30	0.01
Ti/Al/Ti/Au	_	700	600	0.12

Table 3.2: Contact Resistance for GaN based Devices

## 3.5 GaN Substrates

Semiconductor devices require high purity and low defect density to ensure large yields and stable device design. Over the past 10 years, the quality of GaN-on-sapphire and GaN-onsilicon templates have improved drastically through hydride vapour phase epitaxy (HVPE) developed by Kyma technologies. The major advantage of this method is the higher growth rates and low material costs compared to MOCVD. Currently, macro defects are nearly minimized  $< 5 \ cm^{-2}$  while effective doping concentrations of semi-insulating templates are well below  $10^{14} \ cm^{-3}$ . The effective doping was caused from interstitial defects where electrons within the semiconductor are promoted into the conduction band. As mentioned in Chapter 2, the effective doping of the material strongly dictates the depletion width under the channel of the device. The substrate doping has been reduced enough to allow a Cr gate metal to deplete the entire semiconductor under the channel. This has allowed the potential for improved off-state and enhancement-mode devices.

### 3.6 Device Isolation

Fabricated semiconductor devices, such as the architecture discussed earlier, require electrical isolation between the contact pads. This prevents the source and drain from unintentionally conducting around the gate when the device should be off. The active area of a silicon based device is conventionally defined by mesa etching or implant isolation. However, fabrication complexities arise in III-V devices with the inability to oxidize regions of the wafer which has been a common technique used in silicon technology.

A simplistic suggestion to isolate the active device was to remove all of the unwanted semiconductor surrounding the device, shown in Figure 3.5. This method removed the possibility of current flowing around the device. However, etching such a durable material has become a minor nuisance. The most common and most efficient method to etch GaN is by a reactive ion beam etching technique with various gas mixtures such as  $BCl_3$  or  $CH_4/H_2/Ar/N_2$  [96].



Figure 3.5: Schematic of GaN MOSFET with mesa etching to reduce the leakage current through the unused bulk GaN.

An alternative to etching GaN for isolation is through high energy oxygen ion implantation. This technique has been shown to produce non-conductive GaN [97]. The oxygen isolation is similar to doping a substrate; however, the oxygen implanted GaN template becomes highly resistive. The methods mentioned are effective at device isolation and are common techniques used in GaN HFET technology.

### 3.7 Summary

To produce a viable enhancement-mode GaN MOSFET, novel fabrication techniques are required to improve the oxide interface, carrier density and contact resistance. Low temperature processing has the potential to minimize complex and expensive processing techniques to reduce the overall cost of the device. With the incorporation of novel low temperature selective deposition lift-off methods the ultra-thin films will not require complex etching. Furthermore, the reduced chemical processing defects and surface states will be minimized to maintain the improved MOS channel and material interfaces. The following chapters will characterize low temperature deposited films which improved various regions of the GaN MOSFET. These improvements suggested that an enhancement-mode GaN MOSFET has the potential to advance power electronic applications.

# Chapter 4

# Gate Dielectric Characterization

### 4.1 Introduction

Arguably the most fundamental regions of the MOSFET are the surfaces along the channel and the interface behavior between the gate dielectric and the semiconductor. The interaction between these two materials ultimately dictates the quality of the MOS channel and therefore the device mobility and performance. To determine the optimal and compatible gate dielectric for GaN MOS applications, various dielectrics materials were considered. The ability to fabricate a high quality gate dielectric with a compatible dielectric/GaN conduction band offset has the potential to reduce the electron tunneling probability and improve the gate control of the device. A suitable gate dielectric must have a large enough band-gap to align above and below the band-gap of GaN, shown in Figure 4.1. The conduction band offset must be sufficiently large enough (>1eV) to reduce the detrimental effect of a thermally excited electron jumping over the conduction band edge [98,99]. The effective mass of the electron in the gate oxide significantly contributes to the probability of an electron tunneling through the film. It has been shown through the Wentzel-Kramers-Brillouin (WKB) tunneling approximation that larger effective electron masses in the dielectric reduce tunneling leakage [100,101]. The peak capacitance density of the MOS structure strongly depends on the dielectric constant of the gate dielectric. Based on dielectric constant, effective mass and conduction band offset, as shown in Figure 4.1, the best choice of dielectrics to examine are  $HfO_2$  and  $ZrO_2$ . Both films were expected to produce large relative dielectric constants, [102], capacitance densities and low effective masses [100, 103, 104].



Figure 4.1: Dielectric constant and effective electron mass for potential gate dielectrics for GaN based devices. The corresponding band diagram and conduction band offset can be inferred for each material.

### 4.2 Dielectric Fabrication

To examine the interface quality between the high- $\kappa$  film and GaN, planar MOSCAP structures, shown in Figure 4.2, were fabricated on  $N^+$  GaN templates  $(5 \times 10^{18} cm^{-3})$ . The high- $\kappa$  dielectrics were grown by an optimized low temperature plasma enhanced atomic layer deposition (PEALD) technique [105]. The growth of the films were deposited in a Kurt J. Lesker chamber whilst maintaining a constant temperature  $(100^{\circ}C)$  and pressure (1.07 Torr). The alternating precursor gases dose/purge times were respectively oxygen (2.00/2.00s) and either tetrakis(dimethylamido)-zirconium (0.04/5.00s) for a  $ZrO_2$  film or tetrakis(dimethylamido)-hafnium (0.04/5.00s) for a  $HfO_2$  film. The templates were deposited with three different thicknesses (58, 40, 30 cycles) for both dielectric films. The contacts were fabricated with sputtered chromium because of the inherent sticking behavior and compatible work function to produce a positive flatband voltage. An atmospheric post fabrication anneal at  $415^{\circ}C$  for 15 minutes was used to activate the gate dielectric.



Figure 4.2: (Left) Schematic of the MOSCAP used to evaluate the dielectric/GaN interface. (Right) TEM image of the high-k dielectric film grown on GaN via PEALD.

### 4.3 Current-Voltage Measurements

Conventional device measurements were obtained on a Keithley 4200-Semiconductor Characterization System to evaluate the dielectric-semiconductor quality and behavior. The measurements were biased through the inner contact while the outer portion of the MOSCAP was grounded (Figure 4.2). First, current-voltage measurements (Figure 4.3) were obtained for all of the samples. By examining the voltage and film dependent leakage current properties, we were able to characterize the behavior and quality of the films deposited on GaN.

#### 4.3.1 Frenkel-Poole Conduction

The Frenkel-Poole (F-P) leakage conduction model was used to predict the field enhanced thermal emission of carriers from bulk and interface traps [106–108]. By applying an electric field, E, across the dielectric, the barrier height of the traps are reduced by  $\zeta \sqrt{E}$ . The F-P current density model was described in (4.1), where the elementary charge, q, the Boltzmann constant,  $k_b$ , the absolute temperature, T, the dielectric constant,  $\epsilon_r$ , and oxide parameter  $\zeta$ , shown in (4.2) were used. The films' material parameters were extracted from capacitancevoltage measurements. The barrier height of the traps or the effective conduction band offset,  $\Phi_{CB}$ , was obtained by matching the model to experimental results for each film thickness.

$$J = CE \exp\left(\frac{q\Phi_{CB} - \zeta\sqrt{E}}{\xi k_b T}\right) \tag{4.1}$$

$$\zeta = \sqrt{\frac{q^3}{\pi \epsilon_r \epsilon_o}} \tag{4.2}$$

$$\ln\left(\frac{J}{E}\right) = \frac{\zeta}{\xi k_b T} \sqrt{E} + \ln C - \frac{q \Phi_{CB}}{\xi k_b T}$$
(4.3)

The linearized plot (4.3) was achieved by plotting (4.1) in the form of  $\ln \left(\frac{J}{E}\right)$  vs  $\sqrt{E}$ . The converted current leakage was used to extract the barrier height between the GaN conduction band and the effective conduction band of the dielectric [109]. The material parameter constant,  $C = qN_C\mu_n$  was dependent on the GaN conduction band density of states,  $N_C$ , and the electron mobility in the channel,  $\mu_n$  [107]. By fitting the results in (Figure 4.3) to the F-P model (4.3), we were able to find the conduction band offset for  $ZrO_2$  (1.3+/-10%) and  $HfO_2$  (1.33+/-10%). These values were consistent with expected values found in literature for  $ZrO_2$  (1.4 eV) and  $HfO_2$  (1.5 eV) [110]. Based on the results, the effective barrier height was not significantly degraded from trap assisted tunneling within the film. Thus, the main conduction mechanism which caused the majority of leakage current in the ultra-thin films was quantum mechanical direct tunneling. The  $HfO_2$  films were slightly degraded more than the  $ZrO_2$  films which was the first indication that the  $ZrO_2$  films potentially have lower interface defects than the  $HfO_2$  films.



Figure 4.3: Measured current densities with respect to voltage for all film samples. The results were converted into F-P plots to obtain the conduction band offset between the dielectric and GaN.

#### 4.3.2 Quantum Mechanical Tunneling

The MOSCAP direct tunneling current was modeled with quantum mechanical equations that were solved by a novel algorithm known as the transfer matrix method (TMM) [111, 112]. This second-order differential expansion technique has been used to solve similar equations found in microwave waveguides and optoelectronic applications [113]. Through the use of TMM, complex equations can be quickly solved while maintaining high precision [111]. The wave properties of the electrons in the MOS can be described by the secondorder steady state differential equation (4.4), known as the time-independent Schrodinger equation.

$$\left[\frac{-\hbar^2}{8\pi^2}\frac{d}{dx}\frac{1}{m^*(x)}\frac{d}{dx} + V(x)\right]\Psi(x) = E\Psi(x)$$
(4.4)

$$\Psi(x) = A_j e^{p_j(x)} + B_j e^{-p_j(x)}$$
(4.5)

$$p_j(x) = \begin{cases} \Gamma_0 \cdot x & j = 0\\ \Gamma_j \cdot (x - x_{j-1}) & j > 0 \end{cases}$$
(4.6)

$$\Gamma_j(E) = i\sqrt{\frac{8\pi^2 m_j^*}{\hbar^2} \cdot (E - V_j)}$$
(4.7)

The potential profile of the MOS (Figure 4.4) was discretized into N regions, where the potential and effective electron mass at position  $x_j$  are  $V_j$  and  $m_j^*$ , respectively. The corresponding solutions (4.5) to region j were expressed as complex exponential functions along with (4.6) and (4.7). To define the constants  $A_j$  and  $B_j$ , the boundary conditions found in (4.8) and (4.9) relate the constants found in layer j + 1 to the constants in layer j. As defined in (4.8), the wave function must remain continuous along x at each interface. Conversely, the derivative boundary condition (4.9) ensured the system maintains stationary eigenstates for boundaries with different effective mass properties [111, 114].

$$\Psi_{j-1}(x_{j-1}) = \Psi_j(x_{j-1}) \tag{4.8}$$

$$\frac{1}{m_{j-1}^*} \frac{d}{dx} [\Psi_{j-1}(x_{j-1})] = \frac{1}{m_j^*} \frac{d}{dx} [\Psi_j(x_{j-1})]$$
(4.9)



Figure 4.4: (Left) Cross section image of the MOSCAP. (Right) The illustration of the conduction band used to determine the electron tunneling through the dielectric from the transfer matrix method.

Applying the boundary condition through each layer, we can express any region with constants A and B as shown (4.10) and (4.11). Applying (4.10) to each layer produced a relation of the coefficients (A and B) between the input (j = 0) and output layers (j = N + 1) (4.12). This in-turn produced the transmission probability (4.13) of the electrons with a given energy level, E. The wavenumber, k, was used to determine the transmission probability and defined as  $k_j = \sqrt{2m_j^*(E - V_j)}/\hbar$  in region j.

$$\begin{bmatrix} A_{j+1} \\ B_{j+1} \end{bmatrix} = M_j \begin{bmatrix} A_j \\ B_j \end{bmatrix}$$
(4.10)

$$M_{j} = \frac{1}{2} \begin{bmatrix} \left(1 + \frac{m_{j+1}^{*}k_{j}}{m_{j}^{*}k_{j+1}}\right) e^{-i(k_{j+1}-k_{j})x_{j}} & \left(1 - \frac{m_{j+1}^{*}k_{j}}{m_{j}^{*}k_{j+1}}\right) e^{-i(k_{j+1}+k_{j})x_{j}} \\ \left(1 - \frac{m_{j+1}^{*}k_{j}}{m_{j}^{*}k_{j+1}}\right) e^{i(k_{j+1}+k_{j})x_{j}} & \left(1 + \frac{m_{j+1}^{*}k_{j}}{m_{j}^{*}k_{j+1}}\right) e^{i(k_{j+1}-k_{j})x_{j}} \end{bmatrix}$$
(4.11)

$$\begin{bmatrix} A_j \\ B_j \end{bmatrix} = M \begin{bmatrix} A_0 \\ B_0 \end{bmatrix}$$

$$M = \prod_{l=0}^{j-1} M_l = \begin{bmatrix} M_{11} & M_{12} \\ M_{21} & M_{22} \end{bmatrix}$$
(4.12)

$$D(E) = \frac{m_{N+1}^*}{m_0^*} \frac{k_0}{k_{N+1}} \left| \frac{1}{M_{22}} \right|^2$$
(4.13)

Once the tunneling probability was determined for the energy level in the GaN channel, the number of carriers (Q) within the channel (obtained from C - V) was multiplied to determine the total current density (4.14) for each gate voltage [115]. The extracted dielectric properties were obtained from current and capacitance measurement such as conduction band offset, voltage dependent carrier density and dielectric constant. Subsequently, the effective electron mass of dielectric included in the tunneling probability was the only unknown property to match the model to the current-voltage measurements.

$$J_G = 0.6 * \frac{2q}{\left(3\pi\hbar q m_{GaN}\right)^{1/3}} \left(\frac{\epsilon_{ox} F_{ox}}{\epsilon_{GaN}}\right) QD(E)$$
(4.14)



Figure 4.5: The modeled quantum mechanical tunneling current was used to confirm the measured conduction mechanisms and to extract the effective electron mass of the  $ZrO_2$  films. The model (with error bars) was matched to the measured results by creating the band structure for a given voltage and adjusting the effective mass of the dielectric. The effective electron mass of the  $ZrO_2$  films was determined as  $m_{ZrO_2} = 0.2$ .



Figure 4.6: The modeled quantum mechanical tunneling current was used to confirm the measured conduction mechanisms and to extract the effective electron mass of the the  $HfO_2$  films. The model (with error bars) was matched to the measured results by creating the band structure for a given voltage and adjusting the effective mass of the dielectric. The effective electron mass of the  $HfO_2$  films was determined as  $m_{HfO_2} = 0.18$ .

Forward biased current density measurements, shown in Figure 4.5 and Figure 4.6 were obtained for all of the films and compared to the quantum tunneling results. As expected, the thinner films produced larger leakage currents due to the higher tunneling probability through the film. To match the measurements, the effective electron mass was only adjusted and determined as 0.2+/-10% and 0.18+/-10% for  $ZrO_2$  and  $HfO_2$ , respectively. This method confirmed the theoretical calculation of the low effective electron mass of these high- $\kappa$  films.

### 4.4 Capacitance-Voltage Measurements

As stated earlier, capacitance measurements were used to obtain some of the dielectric properties that were required for the current density models. To evaluate the ultra-thin films capacitance-frequency measurements, a distributed model was formulated to extract the capacitance density and effective field effect mobility. By describing the MOSCAP as an equivalent lumped circuit model through the use of transmission line theory [63], we were



Figure 4.7: (Left) An illustration of the modeled planar MOSCAP. (Right) The radial cross section of the modeled planar MOSCAP.

able to separate the properties of the dielectric and semiconductor. This model described the dielectric and semiconductor accumulation layer of the MOSCAP shown in Figure 4.7. By extracting these characteristics of the device, it was possible to understand the quality of the dielectric-semiconductor interface as well as provide an insight into the gate quality and effective channel mobility for a MOSFET structure.

The equivalent circuit model used to define the MOSCAP behavior, shown in Figure 4.8, were represented as passive electrical components. The characteristic impedance of the lumped circuit was modeled as an accumulation layer represented by the series resistance while the ultra-thin dielectric was represented by a shunt capacitance and conductance. This technique has been commonly used to describe the channel resistance in the semiconductors [116]. However, the impedance,  $Z_L$ , which connects the semiconductor under the inner contact to the outer contact, was considered as a resistor that was dependent on the doping concentration of the semiconductor and the distance between contacts.

By using the transmission line lumped circuit model technique, the characteristic impedance  $Z_0$  (4.15) and corresponding propagation delay  $\gamma$  (4.16) are described by using the model parameters  $R_s$ ,  $C_p$ ,  $G_p$  and the operating frequency ( $\omega = 2\pi f$ ). Through the use of complex algebra identities, the imaginary square root of the propagation delay was described by the real values  $\alpha$  and  $\beta$  in terms of the model parameters as show in (4.17) and (4.18) respectively.

$$Z_0 = \sqrt{\frac{R_s}{G_p + j\omega C_p}} \tag{4.15}$$

$$\gamma = \sqrt{R_s(G_p + j\omega C_p)} = \alpha + j\beta \tag{4.16}$$



Figure 4.8: Equivalent lumped circuit model under the inner contact for the planar MOSCAP.

$$\alpha = \frac{\sqrt{2}}{2} \sqrt{\sqrt{(R_s G_p)^2 + (\omega C_p R_s)^2} + R_s G_p}$$
(4.17)

$$\beta = \frac{\sqrt{2}}{2} \sqrt{\sqrt{(R_s G_p)^2 + (\omega C_p R_s)^2} - R_s G_p}$$
(4.18)

An infinitesimals slice of the inner contact was considered as a transmission line with a characteristic impedance of  $Z_0$ . The combination of the theory of small reflections (4.19) and tapered transmission line theory (4.20) were used to determine the reflection coefficient of the infinitesimal slice [117]. The tapered line of the inner contact (Figure 4.9) assumed that the characteristic impedance began at the center of the inner contact and varied as a function of distance. The edge of the inner contact was assumed to connect to the load impedance that represented the resistance of the semiconductor and resistance to the outer contact. From triangular tapered line theory stated that the area gradually changes the impedance as a function of radii from the characteristic impedance to the load impedance shown in (4.20).



Figure 4.9: A small slice of the inner contact  $(\partial \phi)$  visually converted into the equivalent triangular tapered transmission line.

$$\Gamma(\omega) = \frac{1}{2} \int_0^L e^{-2\gamma r} \frac{\partial}{\partial r} \ln\left(\frac{Z}{Z_0}\right) dr$$
(4.19)

$$Z(r) = \begin{cases} Z_0 e^{2\frac{r}{L} \ln\left(\frac{Z_L}{Z_0}\right)} & 0 \le r \le \frac{L}{2} \\ Z_0 e^{\left(4\frac{r}{L} - 2\frac{r^2}{L^2} - 1\right) \ln\left(\frac{Z_L}{Z_0}\right)} & \frac{L}{2} \le r \le L \end{cases}$$
(4.20)

Through the use of the two equations above, the reflection coefficient of the infinite decimal area with radius L, was explicitly calculated and shown in the equation below (4.21). Once the reflection coefficient of the transmission line was determined, the effective input impedance can be calculated from the characteristic impedance and the ration  $(1 + \Gamma)/(1 - \Gamma)$ . The entire input impedance of the inner contact was calculated in (4.22), by taking the integral of the input impedance of the infinitesimal area ( $Z_{Slice}$ ) with respect of the radial direction ( $\partial \phi$ ). By assuming that the small tapered region was independent about any rotation a simple multiplication of the factor  $2\pi$  was sufficient.

$$\Gamma(\omega) = \frac{\ln\left(\frac{Z_L}{Z_0}\right)}{2L(\alpha+j\beta)^2} \left[1 - e^{-L(\alpha+j\beta)} \left(\alpha L + j\beta L + 1\right)\right] + \frac{\ln\left(\frac{Z_L}{Z_0}\right)}{2L^2(\alpha+j\beta)^2} \left[e^{-2L(\alpha+j\beta)} + e^{-L(\alpha+j\beta)} \left(\alpha L + j\beta L - 1\right)\right]$$
(4.21)

$$Z_{in} = 2\pi Z^{slice} = 2\pi Z_0 \left(\frac{1+\Gamma}{1-\Gamma}\right)$$
(4.22)

The input impedance of the total inner contact contained real and imaginary components. The real component of the admittance corresponded to the physical measured conductance ( $G_{Measured}$ ), while the imaginary portion described the measured capacitance of the planar MOSCAP. By plotting (4.23) with respect to frequency, the capacitancefrequency measurement was replicated for a given inner contact voltage by adjusting the model parameters, shown in Figure 4.10a. Once the experimental data was correctly fitted to the model for a given device length L, the four parameters ( $R_s$ ,  $C_p$ ,  $G_p$ , and  $Z_L$ ) were used to extract and characterize the interface quality.

$$C(f) = \frac{imag(Y_{in})}{2\pi f} = \frac{imag(1/Z_{in})}{2\pi f}$$
(4.23)

The capacitance density, shown in Figure 4.10b, was extracted from the model for each voltage by matching the measured C - f spectrum to the model. The model and measurements confirmed the  $ZrO_2$  films produced larger capacitance densities compared to the  $HfO_2$  films for an equivalent growth cycle. The relative dielectric constant of the films were found through (4.24) which included the accumulation thickness within the GaN. Based on in-situ ellipsometry measurements during the growth, the  $ZrO_2$  and  $HfO_2$  films produced nearly equivalent growth rates and film thicknesses. Measurements indicated that the dielectric constant increased as the films became thinner for both rare earth metal oxides. This agreed with glancing XRD measurements of similar depositions which confirmed the film changed from amorphous to semi-crystalline as the film thickness increased.

$$C_{peak} = \epsilon_0 \left(\frac{t_{ox}}{\epsilon_r} + \frac{t_{acc}}{\epsilon_s}\right)^{-1}$$
(4.24)

Low frequency hysteresis C-V measurements (Figure 4.10) were used to compare the density of interface traps along the dielectric and GaN surface. By sweeping the capacitance with respect to voltage, the interface traps were filled which caused a charging effect to shift the flatband voltage in the reverse sweep. By using a low frequency signal (10 kHz), we



Figure 4.10: (a) Capacitance-frequency measurements used to characterize the films through a planar MOSCAP model. (Inset) Voltage dependence of 40 cycles  $HfO_2$  C-f plots (b) The extracted capacitance density of the MOSCAP obtained from the C-f measurements. (Inset) Hysteresis of the 40 cycles C-V at 10 kHz.



Figure 4.11: Density of interface traps along the dielectric/GaN surface obtained from  $G_p - \omega$  and hysteresis measurements for (a)  $HfO_2$  and (b)  $ZrO_2$ .

ensured that the interface traps would respond and create the largest possible shift. The hysteresis,  $\Delta V$ , was used to determine the number of traps along the interface (4.25); where dielectric capacitance,  $C_{ox} = \epsilon_{ox}/t_{ox}$ , and the band-gap,  $E_G$ , of the semiconductor were known.

$$D_{it} = \Delta V \frac{C_{ox}}{qE_G} \tag{4.25}$$

The hysteresis results were compared to the conductance method (Figure 4.11) by extracting the conductance with respect to frequency while the gate was biased above the flatband voltage. Both techniques were upon agreement that the  $HfO_2$  films (Figure 4.11a) produced roughly 2-10 times more interface traps than the  $ZrO_2$  films (Figure 4.11b). The narrow distribution of traps with respect to frequency in the  $ZrO_2$  suggested there was nearly a single energy level along the interface rather than the broad  $HfO_2$  distribution resulting from a more continuous distribution of traps along the energy levels throughout the GaN band-gap [16].

The results were compiled and compared (Table 4.1) for all films and thicknesses. The most important characteristics extracted from the model were the capacitance density, the dielectric constant, the peak electron channel mobility, the density of interface traps and the maximum operating voltage. The electron channel mobility was obtained from the MOSCAP model parameters  $C_p$  and  $R_s$  [63]. The maximum voltage for each film was determined when the current density increased due to high leakage; this was also observed

Film	#	Capacitance	Relative	Peak	$D_{it}$	$D_{it}$	Max
Type	Cycles	Density	Dielectric	Mobility	$(\Delta V)$	$(G_p - \omega)$	Voltage
	30	$3.8 \ \mu F/cm^2$	34.8 +/- 4	$360 \ cm^2/Vs$	$5 x 10^{10}$	$3x10^{10}$	2V
$ZrO_2$	40	$2.6~\mu F/cm^2$	22.8 + / - 2	$387 \ cm^2/Vs$	$3x10^{10}$	$4x10^{10}$	4V
	58	$1.1~\mu F/cm^2$	12.1 + - 1	$370 \ cm^2/Vs$	$6 x 10^{10}$	$2x10^{10}$	$5\mathrm{V}$
	30	$3.0 \ \mu F/cm^2$	22.9 + - 4	$263 \ cm^2/Vs$	$40 x 10^{10}$	$7 x 10^{10}$	2V
$HfO_2$	40	$2.1~\mu F/cm^2$	16.9 + / - 2	$243 \ cm^2/Vs$	$50 x 10^{10}$	$5 x 10^{10}$	4V
	58	$0.8~\mu F/cm^2$	8.1 + - 1	$233 \ cm^2/Vs$	$20 x 10^{10}$	$9x10^{10}$	$5\mathrm{V}$

Table 4.1: Extracted Dielectric Properties on GaN

when the measured capacitance drastically dropped due to Fowler-Nordhiem (F-N) leakage [100, 118]. This occurred when the conduction band of the dielectric was significantly bent due to high electric fields across the ultra-thin oxide. In turn, the effective thickness appeared obsolete which exponentially increased the tunneling probability and leakage current density as well. Compared to literature, these films were able to obtain high capacitance densities while maintaining a low interface trap density, shown in Figure 4.12 [66–80]. These characterized films provided a high capacitance density and low density of interface traps for a potential GaN MOSFET to surpass current HFET technology.



Figure 4.12: The dielectric properties of the low temperature PEALD  $HfO_2$  (red stars) and  $ZrO_2$  (blue stars) films compared to literature (triangles). The ultra-thin films were able improve the capacitance density and interface traps by a factor of 30.

Based on the experimental results of the planar MOSCAP, it was determined that  $ZrO_2$  films enhanced the performance and device characteristics over  $HfO_2$  and other gate dielectrics. The  $ZrO_2$  films produced the largest capacitance density and lowest interface defects which in turn produced large dielectric constants and large peak electron mobility, respectively. Though the thinnest  $ZrO_2$  films (30 cycles) produced the best results, the maximum voltage before uncontrollable gate leakage (F-N) occurred at half of 40 and 58 cycles maximum voltage. Thus, the 40 cycle  $ZrO_2$  film would maintain sufficient gate leakage while providing high quality control of the GaN channel.

### 4.5 Summary

The novel low temperature plasma-enhanced atomic layer deposition of rare earth metal oxides improved the GaN channel interface over conventional dielectric deposition techniques. The films were able to produce a high capacitance while preventing a low density interface traps along the channel. From electrical measurements, the MOSCAP mobility of the  $ZrO_2$  films were improved over the  $HfO_2$  films by > 100  $cm^2/Vs$ . The improvement was a direct results of the GaN and  $ZrO_2$  interface and the reduce density of traps. Based on the extracted results a  $ZrO_2/GaN$  gate dielectric would have the potential to improve the charge control more effectively than the PEALD  $HfO_2$  and films found in literature. The minimized density of interface traps suggested the channel mobility on the MOSFET has the potential to surpass 300  $cm^2/Vs$ .

With the most important interface of the device resolved and characterized, the following chapter will examine the novel technique used to overcome the complications arising from ion implantation in GaN for the source/drain regions as well as the substrate doping required for optimal device performance.

# Chapter 5 Source and Drain Characterization

### 5.1 Introduction

Following the characterization of the gate dielectric and channel interface, PEALD films were examined for use along the source and drain to supply carriers to the channel of the device. A low sheet resistance in addition to a low contact resistance are imperative to minimize overall parasitic resistances within the MOSFET. Negligible parasitic resistance will allow the drain current to be dictated by the channel, rather than being limited by the contact resistances outside of the device. As discussed earlier the techniques required to physically dope the semiconductor through ion implantation or growth diffusion are expensive and complex. Following an ion implantation, high annealing temperatures are required to fix the defects formed in the substrate and along the interface. To circumvent this issue, polarized films grown on GaN were examined to produce a high concentration of carriers along the interface to behave as the source and drain regions.

It has been well documented how specific polarized films on GaN are capable of creating a 2-dimensional electron gas carriers (2DEG) along the interface. These large carrier densities along the interface have the equivalence to highly doped regions in GaN from conventional doping techniques [83,86,119,120]. Based on Wentzel-Kramers-Brillouin (WKB) approximation of quantum mechanic tunneling, the incorporation of an ultra-thin film, a low conduction band offset and a small effective electron mass have the potential to realize tunneling contacts [100, 101]. In addition, the proposed 2DEG formed along source and drain regions has been shown to provide a high mobility interface to further reduce the device parasitic resistances. A large band gap film was preferred to promote conduction through the GaN rather than the thin film. Lastly, the reliability of the films for high power applications must be able to surpass current AlGaN HFET technology. With GaN striving to become the next generation high power transistor, the reliability of the films are essential for longevity of the device.

Based on effective mass, band-gap, electron affinity, polarization and atomic mass, as shown in Figure 5.1, the nitride films that were examined for the incorporation in the GaN MOSFET were AlN and ZrN. ZrN has been demonstrated as a metallic film when the chemical composition between nitrogen and zirconium was 1:1. However, as the nitrogen content was increased the  $ZrN_x$  film demonstrated semiconductor behavior [121–123]. The  $ZrN_x$  semiconductor phase, like AlN films, has the potential to create a 2DEG along the GaN interface; however, to date this has not been investigated to the fullest extent. InN and TiN were not consider since TiN does not have a band gap and InN/GaN would form a channel in the InN rather than the GaN. Thus, these two materials were not useful nitride films for the source and drain region. Conversely, AlN films have the potential to provide a large 2DEG and wide band-gap, ideal for the source and drain regions, while ultra-thin ZrN films provided a compatible electron affinity to GaN, a large atomic mass for the potential of improved reliability for incorporation into the contacts and high electric field regions.



Figure 5.1: The proposed nitride based films for the source/drain regions compared three electrical properties imperative to produce low ohmic contacts, high carrier densities and a reliable device; the effective electron mass, metal atomic radii and polarization charge were compared between the nitride films. AlN has the potential to form the largest 2DEG along the interface, while ensuring the current would flow through the GaN.

### 5.2 Nitride Film Fabrication

The fabrication techniques used to grow the AlN and  $ZrN_x$  films were similar to the techniques discussed in the previous chapter for the gate dielectrics. However, rather than using an oxygen plasma as one of the precursors, forming gas  $(H_2 + N_2)$  was used as the nitrogen plasma precursor. The films were deposited in a Kurt J. Lesker chamber at a constant pressure of 1 Torr. The alternating precursor gases used were forming gas and trimethylaluminum (TMA) for AlN, or tetrakis(dimethylamido)-zirconium (TDMAZ) for  $ZrN_x$  films. The growth temperature of the films, AlN  $(230^{\circ}C)$  and  $ZrN_x$  (150°C), were based on optimized growth conditions based of XRD described in [124]. The AlN films does/purge times for the forming gas and TMA were (10.00/7.00s) and (0.02/7.00s), respectively. Likewise, the  $ZrN_x$  films were optimized for forming gas and TDMAZ with does/purge times of (9.00/9.00s) and (0.10/9.00s), respectively. The  $ZrN_x$  deposition parameters were adjusted such that the semiconductor phase was grown rather than the metallic  $ZrN_x$  film (Zr:N = 1:1). The semiconductor film was produced by creating the  $ZrN_x$  ( $x \approx 1.3$ ) chemical structure, which created a band-gap of approximately  $E_G = 1 - 2eV$  [121–123,125,126].

To initially characterize the films, approximately 3 nm films were deposited on GaN, shown in Figure 5.2. Depending on the material, the number of PEALD cycles were different



Figure 5.2: The film thickness for AlN and  $ZrN_x$  with respect to PEALD cycle were obtained from in-situ ellipsometry; courtesy of Pouyan Motamedi (AlN) and Triratna Muneshwar  $(ZrN_x)$ .
for AlN (40 cycles) and  $ZrN_x$  (30 cycles). The growth per cycle rate for each material was determined from in-situ ellipsometry. Following the growth of the films, the metals under investigation were sputtered and patterned by a photoresist lift-off technique. Once these films were characterized, other film thicknesses of AlN and  $ZrN_x$  were examined to further improve the contact and sheet resistance.

## 5.3 Characterization Methods

Test structures, shown in Figure 5.3, were used to characterize the films along the source and drain regions through current-voltage (I-V) and capacitance-voltage (C-V) measurements. These measurements were used to extract the contact resistance, sheet resistance, 2DEG carriers and 2DEG mobility. The three test devices fabricated were transfer length method structures (TLM), metal oxide semiconductor transfer length method structures (MOS-TLM) and metal oxide semiconductor heterojunction capacitors (MOSHCAP). The MOS-TLMs and MOSHCAP included the characterized  $ZrO_2$  film (40 cycles) discussed in the previous chapter.



Figure 5.3: Schematic of the test structures to evaluate the AlN and  $ZrN_x$  films. The TLM structures were used to evaluate the contact and sheet resistance produced by the 2DEG along the GaN interface. The MOS-TLM included a non-polarized region to extract the contact and sheet resistance of the nitride film, in addition to confirming the GaN sheet resistance. The advantage of the MOSHCAP was to extract the 2DEG through capacitance-voltage measurements.

Current-voltage measurements were obtained for the TLM and MOS-TLM to evaluate the contact behavior. The total resistance between the gaps were used to determine the sheet resistance  $(R_{sh})$  of the films interface and contact resistance  $(R_c)$  of the metal deposited [16]. A first indication whether the nitride film and GaN interface induced a polarized charge was determined from the sheet resistance. An increased carrier density along the films interface behaved similar to a doped region along the GaN template which would lower the sheet resistance. The incorporation of a tunneling contact has the potential to reduce the complexity of the metal stack and high annealing temperatures [84, 127]. It has been shown that the contact resistance of tunneling contacts has a strong dependence on the thickness of the film, the carrier density at the interface and the workfunction of the contact metal [82, 128].

The 2DEG density formed along the interface of the  $ZrN_x$ /GaN and AlN/GaN interfaces was determined from the capacitance measurements of MOS-HFET diodes. This was achieved by determining the amount of charge under the channel between reverse bias pinch-off voltage ( $V_{Pinch}$ ) and flat-band voltage ( $V_{FB}$ ). A gate dielectric was used to create a capacitor which could control the 2DEG channel. By comparing the depletion capacitance measurements with expected values of a MOS, the 2DEG could be determined. The conductivity of the interface can be related to the sheet resistance of the films (5.1) and the 2DEG concentration (5.2) to extract the 2DEG mobility of the film [14–16]. The interface mobility (5.3) was calculated based on the extracted sheet resistance obtained from the TLMs and the 2DEG carriers extracted from the MOSH capacitors. This technique is commonly used to characterize the mobility and interface quality of the 2DEG of the thin films grown on a semiconductor [85].

$$\sigma = \frac{1}{\rho} = \frac{1}{R_{sh} \cdot t_{2DEG}} \tag{5.1}$$

$$\sigma = nq\mu_e = \frac{N_{2DEG}}{t_{2DEG}}q\mu_e \tag{5.2}$$

$$\mu_e = \frac{1}{qR_{sh}N_{2DEG}} \tag{5.3}$$

## 5.4 Aluminum Nitride

The incorporation of AlN on GaN, shown in Figure 5.4, has be known to form a 2DEG along the interface due to an induced polarization charge. To characterize the AlN film Si-doped GaN (N<sup>+</sup>) and unintentionally-doped GaN (N<sup>-</sup>) templates were used to examine the 2DEG formation along the interface and tunneling contact resistance. It has been shown that the 2DEG formed along the interface has a strong dependence on the surface potential of the GaN substrate due to substrate doping [129]. An ultra-thin 3 nm AlN film was deposited on different doped GaN templates in an attempt to produce a low ohmic contact with a high tunneling probability of electrons the barrier. The tunneling probability of the films was constant; however, the formation of the 2DEG along the interface was strongly dependent on the substrate doping [130]. Current and capacitance measurements associated with voltage and frequency dependence were examined to evaluate the films for the potential use in the GaN MOSFET architecture.



Figure 5.4: The band structure of the AlN/GaN with the induced 2DEG carriers formed along the interface.

## 5.4.1 Current-Voltage Characterization

The DC current-voltage measurements were obtained for tradition TLMs as-well as MOS-TLMs to extract the contact and sheet resistance of the AlN films grown on the various doped GaN templates. The higher doped GaN produced large current densities, shown in Figure 5.5a, regardless of the metal contact compared to the unintentionally doped GaN. However, The Al contacts were able to improve the contact resistance by roughly 20% over the Cr metalization contact. The low metal workfunction improved the tunneling probability which in-turn reduced the tunneling contact resistance. The normalized contact resistance was determined to be consistent for all device dimensions, as shown in Figure 5.5b. The smaller devices features produced more resistive contacts, however the contact resistivity of the contacts were consistent.

The results of the N<sup>+</sup> GaN MOS-TLMs and N<sup>-</sup> GaN TLMs, shown in Figure 5.6, were used to extract the sheet resistance and tunneling contact resistance of the metal/AlN/GaN. The total resistance of the MOS-TLM (5.4) was described by the contact resistance, and sheet resistance of the gate, source and drain regions. The slope of the gate-drain MOS-TLM was used to extract the sheet resistance of the AlN/GaN. Likewise, the contact resistance was extracted from the y-intercept once the source and gate region was canceled out. As



Figure 5.5: (a) The I-V measurements from the 75  $\mu m$  gap TLM followed the trend as the doping of the GaN was reduced. In addition, the Al metal contact produced higher current densities than the Cr counter part for both cases of GaN templates. (b) The Cr MOS-TLM results on N<sup>-</sup> GaN confirmed the contact resistance from two devices with different contact areas. The extracted contact resistivity between the different dimension devices was less than 5%.

expected the higher doped templates produced a lower sheet resistance compared to the N- GaN substrates. Furthermore, the large amount of carriers under the metal contact, significantly reduced the contact resistance.

$$R_{tot} = 2R_c + R_{AlN}^S \frac{L_{GS}}{W} + R_{ZrO_2}^G \frac{L_G}{W} + R_{ZrO_2}^D \frac{X_{ox}L_{GD}}{W} + R_{AlN}^D \frac{(1 - X_{ox})L_{GD}}{W}$$
(5.4)

The extracted sheet resistance and contact properties obtained from the TLM I-V results were compiled in Table 5.1. As expected, the metal contact did not effect the 2DEG sheet resistance at the AlN/GaN interface. However, the background doping concentration significantly dictated the sheet resistance and in-turn the tunneling contacts. Through the use of a the lower electron affinity metal (Al), the contact resistance was reduce for both GaN doping concentrations.



Figure 5.6: (a) The N<sup>+</sup> GaN Cr MOS-TLM and the (b) N<sup>-</sup> GaN TLM results used to extracted the contact and sheet resistance. The N<sup>+</sup> GaN templates produced lower contact and sheet resistance compared to the N<sup>-</sup> GaN.

Table 5.1: Extracted Contact and Sheet Resistance Properties of AlN on GaN templates	Table 5.1: Extracte	d Contact and S	Sheet Resistance	Properties of A	AlN on GaN	templates
--	---------------------	-----------------	------------------	-----------------	------------	-----------

GaN Doping	Device	$\frac{R_{Cr}}{(\Omega \cdot mm)}$	$R_{Al} \ (\Omega \cdot mm \ )$	$\frac{R_{GaN}}{(\Omega/sqr)}$	$\frac{R_{AlN}}{(\Omega/sqr)}$
$N^+$	TLM MOS-TLM	$0.61 \\ 0.58$	$0.45 \\ 0.40$	616	184 236
$N^{-}$	TLM MOS-TLM	$45 x 10^3$ $39 x 10^3$	$35 x 10^3$ $28 x 10^3$	5000	$\begin{array}{c} 2309 \\ 2084 \end{array}$

#### **Post Fabrication Annealing**

It has been well documented that post fabricated annealing has the ability to reduce the contact resistance by orders of magnitude [92–95]. Extensive studies of AlGaN/GaN have shown the contact resistance significantly improves at higher temperatures for a wide range of annealing times [83,93,131]. This allows the process to be accurate and consistent with each batch fabricated. The 3 nm AlN on  $N^-$  GaN template was used for the annealing study to improve the contact resistance. The GaN template was exposed to the noted temperature for 15 minutes before electrical characterization was performed. Conventional TLM structures were measured to examine the contact resistance of the device as discussed earlier.



Figure 5.7: (a) As the TLM structures were annealed the current density of the 5  $\mu m$  TLM gap increased. (b) The improved conduction was strongly dependent on the reduced contact resistance with respect to annealing temperature. Annealing the device at 700°C was found to produce a low ohmic contact (5x10<sup>-3</sup>  $\Omega \cdot cm^2$ )

The contact resistance at each annealing temperature was extracted from the I-V measurements of the 5  $\mu m$  TLM gap, shown in Figure 5.7a. The sheet resistance of the AlN/GaN gap was neglected since the gap resistance was more than 2 orders of magnitude smaller than the contact resistance. The contact resistance with respect to annealing temperature was displayed in Figure 5.7b. The results agreed with literature, that higher temperatures, regardless of the annealing time will improve the contact resistance for GaN based devices.

With the contact and sheet resistance characterized from the I-V results, the amount of the carriers formed along the thin film and GaN interface was examined through the frequency dispersion of capacitance and conductance results. In-conjunction with the sheet resistance, the capacitance-voltage measurements were used to extract the polarization as well as provide the characteristics required to obtain the 2DEG mobility along the AlN/GaN interface.

## 5.4.2 Capacitance-Voltage Characterization

The capacitance-voltage frequency response of a semiconductor device is the most accurate method to extract the behavior of the carriers within and along the interface of ultra-thin films. The 2DEG carriers formed along the interface were evaluated through the comparison between convention MOSCAPs and AlN MOSCAPs during reverse depletion regime. The

Gate	Thickness	Capacitance	$t_{acc}$	c
Stack	(nm)	$(\mu F/cm^2)$	(nm)	$\epsilon_r$
$ZrO_2$	$6.0 \pm 0.2$	2.56	1.0	$22.8\pm2.0$
$ZrO_2/AlN$	$6.0\pm0.2$	1.36	0.8	$22.8\pm2.0$
$Z T O_2 / A t N$	$3.1\pm0.2$	1.50	0.8	$9.3\pm0.6$
AlN	$3.1\pm0.2$	2.15	0.8	$9.3 \pm 0.5$

Table 5.2: AlN Dielectric Constant Extracted from Peak Capacitance Density

accumulated carriers below the theoretical flatland voltage were a result of the induced polarization effect caused from the AlN and GaN [132]. To improve the accuracy of the C-V results the characterized  $ZrO_2$  was included in the gate stack to reduce the tunneling conductance through the gate. The difference in peak capacitance was a result of the gate stack with and without the  $ZrO_2$ , shown in Table 5.2.

The peak capacitance density obtained in strong accumulation in conjunction with the accumulation thickness determined the dielectric constant of the AlN film. The accumulation thickness, shown in Figure 5.8, was determined from the 2D Poisson-Schroedinger equation. The AlN was assumed to have a narrow channel, similar to the simulated AlN (0.75-0.95 nm). Likewise, the  $ZrO_2$  was expected to have a slightly larger channel than



Figure 5.8: The simulated accumulation thickness for the AlN/GaN and  $ZrO_2$ /GaN to extract the dielectric constant of the films. The effective mass of the films determined the location of the peak centroid and the amount of carriers within the dielectric film. AlN allowed 4x the amount of carriers into the film compared to the  $ZrO_2$  film.

the AlN. Whereas, the effective mass ratio between  $ZrO_2/\text{GaN}$  was expected to produce a smaller channel than the conventional  $SiO_2/\text{GaN}$  (1.2-1.4 nm). Thus, the accumulation centroids for  $ZrO_2$  and AlN were assumed 1.0 nm and 0.8 nm, respectively.

#### **Polarized Charge Extraction**

The inclusion of the gate dielectric significantly improved the consistency of the capacitance frequency spectrum, Figure 5.9a, and reduced the conductance of the thin films, Figure 5.9b, allowing for a more accurate measurement of the 2DEG. A reverse biased capacitancevoltage measurement was required to extract the 2DEG charge formed along the AlN/GaN interface. As the herterojunction was depleted, the 2DEG charge  $(Q_{2DEG})$  was removed from the polar induced quantum well at the pinch-off voltage  $(V_{Pinch})$ . The area beneath the capacitance measurement  $(C_m)$  (5.5) between flat-band voltage  $(V_{FB})$  and  $V_{Pinch}$  was considered as the total charge produced from the AlN film. Removing the theoretical depletion charge within the GaN provided a more accurate extraction of the 2DEG charge and in-turn the 2DEG carriers (5.6).



$$Q_{2DEG} = \int_{V_{Pinch}}^{V_{FB}} C_m dV - Q_{Dep}$$

$$\tag{5.5}$$

Figure 5.9: The (a) capacitance and (b) conductance frequency dispersion based on the MOSCAP with and without a gate dielectric  $(ZrO_2)$  under one of the metal contacts. By including the gate dielectric, the leakage conductance was reduced which provided less frequency dispersion than the AlN metal films.



Figure 5.10: The reverse C-V gate bias of the MOSHCAPs used to extract the 2DEG for both GaN templates. The area between the C-V measurement and the theoretical depletion capacitance for GaN was the polarization interface charge. The  $N^+$  GaN (blue) produced a 2DEG roughly 10x larger than the  $N^-$  GaN (green).

$$N_{2DEG} = \frac{Q_{2DEG}}{q} \tag{5.6}$$

The charge carrier density, Figure 5.10, was determined from the area beneath the capacitance-voltage measurements between flat-band voltage and pinch-off voltage. The inclusion of the dielectric film reduced the leakage current which produced a more consistent measurement of the 2DEG,  $2.9 \times 10^{13} \pm 0.4 \times 10^{13} cm^{-2}$ , with negligible frequency dispersion. The 2DEG carriers obtained from the capacitance for the N<sup>+</sup> and N<sup>-</sup> GaN confirmed that the variation between the TLM sheet resistance was consistent. The unintentionally doped GaN produced a lower 2DEG extracted from the C-V measurements, shown in Figure 5.10. The theoretical depletion capacitance for N<sup>+</sup> and N<sup>-</sup> GaN were included to confirm the increase of carriers from a non-polarized interface. This has been theorized that the film crystal orientation and effective polarization was dependent on substrate doping [129]. The peak carrier density along the AlN/GaN interface was determined from the thickness dependence 2DEG polarization effect.

## 5.4.3 AlN Thickness Dependence

The post fabrication anneal was shown to improve the contact resistance of the structures, however the sheet resistance remained unchanged. The 2DEG carriers along the GaN interface have been shown to increase with AlN thickness [85, 129]. The PEALD process grew 5.7 nm and 7.9 nm films which required 70 and 100 cycles, respectively. Improving the sheet resistance along the AlN/GaN interface would imply that the carriers within the 2DEG increased. In-turn, the tunneling contact resistance would improve, if the increased 2DEG offset the reduce tunneling probability of the larger film thickness.

Current-voltage measurements of conventional TLMs were used to extract the sheet and contact resistance, shown in Figure 5.11. As the AlN film thickness doubled from 40 to 70 cycles, the sheet resistance was reduced by a factor of 4. This suggested the amount of 2DEG carrier along the interface increased, assuming the channel mobility remained roughly constant. The 100 cycles AlN demonstrated a significant increase in the sheet resistance compared to the 70 cycle film, Figure 5.11b. It has been theorized that strain relaxation and dislocations within the AlN reduce the polarization strain when the thickness was beyond a critical value [129, 132]. The expected critical relaxation thickness for AlN was theoretically 6 nm [133], which agreed with the maximum 2DEG determined for the 70 cycle film. Beyond the critical film thickness, strain relaxation occurred, preventing an increase in 2DEG for thicker AlN films.



Figure 5.11: The extracted contact and sheet resistance from the 5.7 nm and 7.9 nm AlN/GaN films.

The tunneling contact resistance and extracted sheet resistance and 2DEG for each film thickness was compiled in Table 5.3. The largest 2DEG produced by the 70 cycle AlN film improved the contact resistance by more than an order of magnitude compared to the 40 cycle and 100 cycle films. As the AlN films were increased to 70 and 100 cycles the sheet resistance obtained from the TLMs decrease and increase, respectively. This was consistent with the prediction based on the strain relaxation of the AlN beyond 6 nm. Based on the extracted sheet resistance from the I-V measurements, and 2DEG carrier density the interface channel mobility (5.3) was obtained. The results, shown in Table 5.3, were also compared to the 3 nm AlN on N<sup>+</sup> GaN and a semi-insulating (SI) GaN template. The SI GaN template was patterned with a 6 nm AlN. To increase the 2DEG and reduce the sheet resistance the channel was diffusion doped with Si. Titanium contacts were deposited and annealed at  $415^{\circ}C$  to form ohmic contacts with Si doped AlN/GaN regions. Based on the contact resistance, the doped GaN was capable of providing low contact resistance without the required high temperature annealing. However, the semi-insulating template required high temperature annealing to produce equivalent sheet resistance and contact resistant as the unintentionally doped GaN.

GaN	AlN	$R_C$	$R_{AlN}$	$N_{2DEG}$	$\mu_e$
Doping	(nm)	$(\Omega \cdot cm^2)$	$(\Omega/sqr)$	$(10^{13} cm^{-2})$	$(cm^2/Vs)$
N <sup>+</sup>	3.2	$3.1 \mathrm{x} 10^{-4}$	$210\pm10$	2.39	$1215 \pm 46$
	3.2	$1.7 \text{x} 10^{1}$	$2178\pm130$	0.28	$1024\pm58$
N <sup>-</sup>	5.7	$2.9 \mathrm{x} 10^{-2}$	$592 \pm 30$	1.14	$926 \pm 45$
	7.9	$1.6 \mathrm{x} 10^{-1}$	$1307\pm50$	0.47	$1017\pm38$
SI	5.7	$3.5 \mathrm{x} 10^{-2}$	$610 \pm 50$	-	-

Table 5.3: Extracted Sheet and Contact Resistance

## 5.4.4 AlN Summary

Low temperature PEALD ALN films were demonstrated for the potential incorporation along the source and drain of a GAN MOSFET. The ultra-thin films produced large 2DEG carrier densities along the GaN interface while maintaining large channel mobilities. The PEALD films were able to produce similar 2DEG carrier densities and channel carrier mobility (Figure 5.12a) compared to convention MOCVD and MBE deposition techniques with growth temperatures  $600^{\circ}C$  higher than the PEALD films characterized. This confirmed the large carriers densities and high interface mobility was possible with low temperature PEALD deposition techniques. The large induced carrier density along the GaN interface provided the possibility of tunneling contacts through the AlN film. This reduced the required annealing temperature to achieve an ohmic contact, shown in Figure 5.12b, compared to conventional ohmic contacts for GaN. Furthermore, the metal contact did not required complex metal stacks; instead the tunneling contact resistance was based on the work function of the metal. Moreover, through annealing the contacts, the undoped GaN produced equivalent contact resistances compared to literature.

The electrical properties of the low temperature AlN films confirmed large carrier densities for the potential use in a GaN MOSFET along the source and drain regions. Through AlN tunneling contacts, doped GaN provided a low ohmic contact comparable to current technology, where as the undoped GaN required high temperature annealing to achieve similar values. The following section examined novel  $ZrN_x$  materials to further improve the reliability of the high electric field regions and potential advance the overall device characteristics of the GaN transistor.



Figure 5.12: (a) The comparison between carrier density to channel mobility of AlN films on GaN. The low temperature films produced equivalent 2DEG charge and carrier mobility as conventional high temperature deposition techniques. (b) The contact resistance of the doped GaN tunneling contact produced similar values to high temperature annealed metal contacts on GaN.

## 5.5 Zirconium Nitride

The semiconductor phase  $ZrN_x$  was also considered for regions along the source/drain of the GaN MOSFET because of the potential polarization, longevity and stability of the material. As mentioned early, the  $ZrN_x$  films electrical behavior (metallic or semiconductor) was based on the stoichiometric ratio between Zr and N. Again, the conventional metallic ZrN was produced by maintaining a 1:1 ratio between Zr and N. However, increasing the nitrogen content within the films has shown to shift the material from a metal to a semiconductor with a band-gap around 2 eV [125]. X-ray photon spectroscopy (XPS) of the deposited  $ZrN_x$  films contained an increase N content (x = 1.2) for the films discussed in this chapter.

The electrical material properties of  $ZrN_x$ , such as effective electron mass  $(m_e)$ , relative dielectric constant  $(\epsilon_r)$ , conduction band density of states  $(N_C)$ , valence band density of states  $(N_V)$ , and electron mobility  $(\mu_e)$ , have not been extensively examined or extracted from electrical measurements. To obtain these properties, ultra-thin PEALD films were grown on p-type Si, N<sup>+</sup> GaN and N<sup>-</sup> GaN templates. Various test structures discussed earlier were fabricated to evaluate the material properties of  $ZrN_x$  and extracted the electrical properties. The silicon wafer was used to confirm the XPS data that the  $ZrN_x$  film was non-metallic through characterization the film grown on the most well known semiconductor. The p-type silicon was expected to produce a pN heterojunction which would allow for the extraction multiple material properties through the C-V depletion region.

## **5.5.1** $ZrN_x$ on Silicon

The  $ZrN_x/Si$  stack was pattern with chromium metal contacts to ensure adhesion and form an ohmic contact on the pN junction. The 3 nm  $ZrN_x$  was characterized with currentvoltage and capacitance-voltage measurements through the pN heterojunction to the back side silicon wafer. The silicon was placed to ground while the metal contact was biased on the top side of the structure.

Through the current-voltage measurement, shown in Figure 5.13a, the results confirmed that the films formed a non-ohmic contact. The  $Cr/ZrN_x/Si$  stack restricted current in one direction while producing a large current in the other; i.e. a diode. This suggested a pN-junction or Schottky barrier was formed along one of the interfaces. To determine which junction was formed along the materials interface, the capacitance-voltage frequency spectrum (Figure 5.13b) was obtained. The reverse C-V bias behaved similar to an accumulation MOS capacitance on silicon. The theoretical built-in potential for metallic ZrN on p-type silicon would be a maximum of 0.4 V, compared to the extracted  $\phi_{bi} = 1.27$  eV. This was the first electrical confirmation that the  $ZrN_x$  was a semiconductor rather than a metal. Furthermore, the capacitance of the Schottky metal contact would have dropped



Figure 5.13: (a) The measured I-V of the  $ZrN_x/Si$  junction, courtesy of Kevin Voon. (b) The frequency dependent C-V measurements of the  $ZrN_x/Si$ , courtesy of Kevin Voon. The rectifying current density confirmed that a Schottky contact or pn-junction was formed. The extract built-in potential ( $\phi_{bi} = 1.27 \text{ eV}$ ) of the material system confirmed a p-n junction was formed at the  $ZrN_x/Si$  interface.

to zero for voltage biases beyond the built-in potential. During the accumulation portion of the measurement, the dielectric constant of the  $ZrN_x$  was extracted through the peak capacitance density. The accumulation of holes along the surface and a native oxide was included in the model to describe a more accurate representation of the series capacitance (5.7).

$$\epsilon_{ZrN} = \frac{t_{ox}}{\epsilon_0} \left( \frac{1}{C_m} - \frac{t_{acc}}{\epsilon_{Si}\epsilon_0} - \frac{t_{SiO_2}}{\epsilon_{SiO_2}\epsilon_0} \right)$$
(5.7)

The peak capacitance density  $(0.75 \ \mu F/cm^2)$  was extracted from the low frequency (10 kHz) C-V measurement to extract the dielectric constant of  $ZrN_x$ , shown in Table 5.4. The accumulation thickness of holes along the silicon channel was assumed to have a finite thickness of roughly 1.1 nm [15,134]. This thickness was assumed equivalent to an inversion layer of holes in an n-type silicon. The native oxide  $(SiO_2)$  was assumed to have a negligible thickness less than a quarter of a nanometer. The dielectric constant ( $\epsilon_{ZrN} = 7.2$ ) remained constant before the cut-off frequency (10<sup>5</sup> Hz) of the structure.

Through Poisson and Gauss equations the depletion lengths within the materials were described by the dielectric properties and doping concentrations [14]. Boundary conditions

Table 5.4: ZrN	Dielectric	Constant	Extracted	from Peak	Capacitance	Density

$\frac{C_m}{(\mu F/cm^2)}$	$t_{ZrN}$ $(nm)$	$\begin{array}{c}t_{acc}\\(nm)\end{array}$	$\begin{array}{c}t_{SiO_2}\\(nm)\end{array}$	$\epsilon_r$
0.75	$7.3\pm0.25$	1.1	0.25	$7.2 \pm 0.3$

were applied at the interface to ensure the continuity of electric displacement and built-in potential were consistent from either side of the junction. Through these equations the depletion widths in the  $ZrN_x$  (5.8) and Si (5.9) were defined accordingly. The depletion width of the entire junction was the two lengths combined to form a capacitor along the junction (5.10). The built-in potential of a pN-junction was also defined from the difference in Fermi levels between the two materials, represented in (5.11). The theoretical values of effective conduction band and valence band density of states for Si were  $3.2 \times 10^{19} cm^{-3}$  and  $1.8 \times 10^{19} cm^{-3}$ , respectively. The third term was neglected as the theoretical density of states for silicon and  $ZrN_x$  were expected to have similar values. Using the slope and intercept of the measured inverse squared capacitance (Figure 5.14), in-conjunction with (5.10) and (5.11) the built-in potential, the  $ZrN_x$  doping, the silicon doping and the expected  $ZrN_x$ band-gap were obtained. In addition, the effective conduction and valence band density of states for were extracted, to further determine the effective masses for electrons ( $m_e$ ) and holes ( $m_h$ ).



Figure 5.14: Inverse squared capacitance of the measured test structure which was used to extract the electrical properties of the  $ZrN_x$  film. The built-in potential was obtained from the x-axis intercept and the silicon doping concentration of was extracted from the slope.

$$x_n = \sqrt{\frac{2N_{A_{Si}}\epsilon_{Si}\epsilon_{ZrN_x}\left(\phi_{bi} + V\right)}{qN_{D_{ZrN_x}}\left(N_{D_{ZrN_x}}\epsilon_{ZrN_x} - N_{A_{Si}}\epsilon_{Si}\right)}}$$
(5.8)

$$x_p = \sqrt{\frac{2N_{D_{ZrN_x}}\epsilon_{Si}\epsilon_{ZrN_x}\left(\phi_{bi}+V\right)}{qN_{A_{Si}}\left(N_{D_{ZrN_x}}\epsilon_{ZrN_x}-N_{A_{Si}}\epsilon_{Si}\right)}}$$
(5.9)

$$C_{dep} = \sqrt{\frac{q N_{D_{ZrN_x}} N_{A_{Si}} \epsilon_{ZrN_x} \epsilon_{Si}}{2 \left( N_{D_{ZrN_x}} \epsilon_{ZrN_x} + N_{A_{Si}} \epsilon_{Si} \right) \left( \phi_{bi} + V \right)}}$$
(5.10)

$$\phi_{bi} = \frac{\Delta E_C - \Delta E_V}{2} + \frac{kT}{q} \ln \frac{N_{D_{ZrN_x}} N_{A_{Si}}}{n_{i_{ZrN_x}} n_{i_{Si}}} + \frac{kT}{2q} \ln \frac{N_{V_{ZrN_x}} N_{C_{Si}}}{N_{C_{ZrN_x}} N_{V_{Si}}}$$
(5.11)

Through self-consistent solving of (5.10) and (5.11), the built-in potential, doping concentrations and band-gap were extracted to Table 5.5. The measured and theoretical builtin potential confirmed a pN-junction was formed along interface rather than a metallic  $ZrN_x$ /Si Schottky barrier. The extracted data confirmed a lightly doped silicon wafer  $(2x10^{16}cm^{-3})$ , while the  $ZrN_x$  was effectively higher doped  $(5x10^{17}cm^{-3})$ . Likewise, the intrinsic carrier concentration, band-gap, effective density of states and effective electron and hole masses were extracted from the capacitance-voltage measurements.

Table 5.5: Extracted Electronic Properties of  $ZrN_x$ 

$N_C$	$6.1 \times 10^{19} cm^{-3}$	
$N_V$	$1.9 \mathrm{x} 10^{19} cm^{-3}$	
$n_i$	$5.9 \mathrm{x} 10^4 cm^{-3}$	
$E_G$	$1.7 \ \mathrm{eV}$	
$\chi_s$	$3.95~{\rm eV}$	
$\epsilon_r$	7.2	
$m_e$	1.80	
$m_h$	0.85	

The properties of the  $ZrN_x$  were obtained from the unique pN heterojunction formed with silicon, which confirmed the semiconductor present rather than a metallic refractory metal nitride. Knowing the electrical properties of silicon, the characterization of the thin film was possible through the band diagram and Poisson and Gauss equations. Furthermore, the understanding of the electrical behavior for  $ZrN_x$  grown on other materials will have a stronger impact knowing the doping concentration and band structure. Films were grown on various doped GaN templates to evaluate  $ZrN_x$ /GaN interface for the potential use in GaN based applications. Similar device structures fabricated for AlN/GaN characterization were used to extract electrical properties of  $ZrN_x$ /GaN interface.

## 5.5.2 $ZrN_x$ on GaN

The extraction of many fundamental properties of the  $ZrN_x$  material was possible since a non-polarized well known semiconductor, silicon, was used to evaluate the pN heterojunction. In-turn, we were able to predict the band diagram of  $ZrN_x$  and the expected band bending when deposited on different GaN templates, shown in Figure 5.15). The N<sup>+</sup> GaN template was expected to produce a 2DEG along the GaN interface due to the free electron carriers moving from the  $ZrN_x$  to the GaN. This effect was similar to modulation doping in AlGaAs/GaAs heterojunction devices [14].



Figure 5.15: The expected band diagrams for  $ZrN_x$  on different background doped GaN templates based on the extracted electrical parameters obtained from the silicon measurements.

To evaluate the electrical interaction between  $ZrN_x$  and GaN, ultra-thin 3 nm PEALD  $ZrN_x$  (40 cycles) films were deposited on Si-doped GaN (N<sup>+</sup>) and unintentionally-doped GaN (N<sup>-</sup>) templates. These templates have different Fermi levels, which will change the built-in potential and electrical behavior based on the expected Fermi level within the  $ZrN_x$ . The TLM test structures and MOSHCAPs where fabricated with Al metal contacts and a  $ZrO_2$  (40 cycles) gate dielectric.

#### **Current-Voltage Characterization**

To evaluate the quality of the contact resistance of the  $Al/ZrN_x/GaN$  system, I-V measurements of conventional TLM structures were used on both GaN templates. By varying the gap distance, the sheet resistance of the films would indicate whether a 2DEG was formed along the GaN interface. Similar to the AlN film, the GaN doping and Fermi level, played a significant role in the current density of the structures, shown in Figure 5.16a. As expected the Si-doped GaN reduced the resistances which increased the current by roughly 8x. Furthermore, the N<sup>+</sup> template was less dependent on the gap distance, which confirmed the sheet resistance for both templates (Figure 5.16b) were extracted from the TLM resistance results.

As expected from the band diagram, the N<sup>+</sup> GaN formed a 2DEG along the interface



Figure 5.16: (a) The current-voltage results of the N<sup>-</sup> (Solid) and N<sup>+</sup> (Dashed) GaN templates. (b) The total resistance with respect to TLM distance used to extract the contact resistance and sheet resistance of the  $ZrN_x$ /GaN interface.

as the sheet resistance of the GaN reduced from 600  $\Omega/sqr$  to 171  $\Omega/sqr$ . Similarly, the N<sup>-</sup> GaN template reduced the sheet resistance by roughly half of the original unintentionally doped GaN. If we assumed non-polarization effect occurred within the  $ZrN_x$  the effectively N<sup>+</sup> doped  $ZrN_x$  would have transfered free electrons to the GaN quantum well based on (5.12) [14]. The amount of carriers transfered into the quantum well was dependent on the number of carriers in the  $ZrN_x$ , the thickness of the  $ZrN_x$ , the conduction band offset and Fermi level. The ultra-thin layer of  $ZrN_x$  was assumed to provide all of the carriers within the film to the quantum well. Based on modulation doping effect the expected number of carriers within the quantum well was  $2.15 \times 10^{12} cm^{-2}$  and  $0.67 \times 10^{12} cm^{-2}$  for N<sup>+</sup> and N<sup>-</sup> GaN, respectively.

$$n_s = \frac{\epsilon_{ZrN}}{qt_{ZrN}} \left( \frac{qN_d}{2\epsilon_{ZrN}} t_{ZrN}^2 + \frac{\Delta E_C - E_f}{q} - \phi_{bi} \right)$$
(5.12)

#### Capacitance-Voltage Characterization

To confirm the predicted modulation doped 2DEG along the interface, capacitance-voltage frequency spectroscopy of MOSHCAPs were obtained. The techniques used to evaluate the AlN films were applied to the  $ZrN_x$  films on GaN. Similar to the AlN, the reverse bias depletion region provided the information about the number of carriers within the quantum well.

The reverse voltage bias depleted the carriers along the quantum channel to determine the total electron density at equilibrium, obtained from C-V measurements (Figure 5.17). Again, the increased capacitance compared to the expected depletion capacitance was due to 2DEG carriers. The 2DEG measured along the interface was significantly larger than the expect 2DEG resulting from the modulation doped free carriers within the film. The additional carriers provided were caused from the polarization of  $ZrN_x$ . This confirmed that the  $ZrN_x$  films had a piezoelectric polarization effect similar to AlN and AlGaN.

To confirm the dielectric properties of  $ZrN_x$ , the capacitance of forward bias MOSH-CAPs were evaluated to confirm the channel location and band diagram. Interestingly, the forward bias C-V results of the MOSHCAPs were different depending on the background doping of the GaN templates. The doped GaN, which formed a larger 2DEG carrier concentration along the interface, produced the expected peak capacitance density (Figure 5.18a) for a  $ZrO_2/ZrN_x$  MOSHCAP. The capacitance was lower than the peak capacitance of 40 cycles of  $ZrO_2$  on GaN, which was similar to the AlN MOSHCAPs which had multiple layers of dielectric films. This confirmed carriers accumulated along the GaN interface during forward bias, depicted in Figure 5.18c. The lower doped GaN templates produced capacitance densities corresponding to a dielectric film of only a 40 cycle  $ZrO_2$  film, shown in Figure 5.18b. For this to happen the accumulation of electrons occured within the  $ZrN_x$  film, rather than the GaN interface. At zero bias a small 2DEG was formed along the GaN interface, however at forward bias, accumulation occurred within the  $ZrN_x$ , shown in Figure 5.18d.

To confirm the dielectric constant and accumulation thickness of the  $ZrN_x$ , the peak capacitance was modeled for both the N<sup>+</sup> (5.13) and N<sup>-</sup> (5.14) GaN templates. The  $ZrO_2$ film capacitance was assumed equivalent to the results extracted from the previous chapter on gate dielectrics. Similarly, the channel thickness in GaN and  $ZrN_x$  were 0.9 nm and 1.0 nm, respectively. The tighter 2DEG channel within the GaN was determined from the dielectric ratio difference between AlN and  $ZrN_x$ . Solving both capacitance equations confirmed a  $ZrN_x$  dielectric constant of 7.4, which was within 5% of the extracted value obtain from the silicon test structures.

$$\frac{1}{C_{N^+}} = \frac{\epsilon_{ZrO_2}\epsilon_0}{t_{ZrO_2}} + \frac{\epsilon_{ZrN_x}\epsilon_0}{t_{ZrN_x}} + \frac{\epsilon_{GaN}\epsilon_0}{t_{acc}}$$
(5.13)

$$\frac{1}{C_{N^-}} = \frac{\epsilon_{ZrO_2}\epsilon_0}{t_{ZrO_2}} + \frac{\epsilon_{ZrN_x}\epsilon_0}{t_{acc}}$$
(5.14)



Figure 5.17: Reverse bias C-V measurements of the  $ZrO_2/ZrN_x/GaN$  MOSHCAPs extracted the 2DEG along the GaN interface for (a) N<sup>+</sup> and (b) N<sup>-</sup> GaN templates. The plots included the theoretical 2DEG expected from  $ZrN_x$  modulation doping  $(N_{ZrN})$ . The additional charge in the reverse bias confirmed that  $ZrN_x$  had polarization properties.



Figure 5.18: Low frequency C-V to obtain the dielectric characteristics of  $ZrN_x$  on (a) N<sup>+</sup> GaN and (b) N<sup>-</sup> GaN. (c) The band diagram at zero bias and strong accumulation regime on N<sup>+</sup> GaN were both along the  $ZrN_x/\text{GaN}$  interface. (d) The accumulation of the electrons for the N<sup>-</sup> GaN template occurred along the  $Zr0_2/ZrN_x$  interface.

The band structure and electrical properties extracted from  $ZrN_x/\text{GaN}$  were comparable to the results extracted from the results on silicon. The increased 2DEG with respect to expected modulation doping effect, confirmed that  $ZrN_x$  had polarization effects similar to AlN. However, the electron carrier density formed along the  $ZrN_x/\text{GaN}$  interface was lower than the AlN, which suggested the polarization of the  $ZrN_x$  was weaker than AlN films. The measurements of the sheet resistance and 2DEG provided the channel mobility, Table 5.6. Based on the interface mobility, the films produced a high quality interface with the GaN.

Table 5.6:  $ZrN_x$  2DEG Properties

GaN	$ZrN_x$	$R_{ZrN_x}$	$N_{2DEG}$	$\mu_e$
Doping	(nm)	$(\Omega/sqr)$	$(10^{13} cm^{-2})$	$(cm^2/Vs)$
N <sup>+</sup>	2.9	$200\pm25$	2.37	$1318 \pm 100$
N <sup>-</sup>	2.9	$3586 \pm 200$	0.12	$1469 \pm 100$

#### **Thickness Dependence**

To evaluate the thickness dependence of the 2DEG and contact resistance of the Metal/ $ZrN_x$ /GaN, 1 nm films were grown on unintentionally doped GaN. The contacts were deposited with Pt and Al to compare the effect of workfunction with contact resistance. Through I-V and C-V measurements, the contact resistance and carrier density were extracted, shown in Table 5.7. The 2DEG of the 1 nm  $ZrN_x$  produced a lower 2DEG than the 3 nm film for both metal contacts. However, the contact resistance of the Pt was an order of magnitude larger than the Al. The large workfunction difference between Pt and Al significantly effected the contact resistance on the  $ZrN_x$ /GaN, as expected.

Table 5.7:  $ZrN_x$  Contact Resistance Properties

GaN	$ZrN_x$	Metal	$N_{2DEG}$	$R_c$
Doping	(nm)	Contact	$(10^{13} cm^{-2})$	$(\Omega cm^2)$
N <sup>-</sup>	1.0	Pt	0.04	$4.3 \mathrm{x} 10^{-1}$
N <sup>-</sup>	1.0	Al	0.04	$6.5 \mathrm{x} 10^{-2}$
N <sup>-</sup>	3.0	Al	0.12	$5.9 \mathrm{x} 10^{-2}$
N <sup>+</sup>	3.0	Al	2.37	$1.4 \mathrm{x} 10^{-2}$

## 5.5.3 $ZrN_x$ Summary

The  $ZrN_x$  films were characterized on p-type silicon and various doped GaN templates to extract multiple electrical properties of the semiconductor. The pN heterojunction formed with silicon confirmed the properties obtained from the nN heterojunction on GaN. Through C-V characterization of the  $ZrN_x$  films the dielectric constant, band-gap, intrinsic carrier concentration, conduction and valence band effective density of states, electron affinity, effective masses of holes and electrons were obtained electrically. The band-gap was determine as 1.7 ev with a dielectric semiconductor relative dielectric constant of 7.2. Based on the band-gap, the intrinsic carrier concentration was estimated in the range of  $10^4 cm^{-3}$ . Through electrical extraction the effective density of state were found to infer the effective electron mass ( $m_e = 1.8$ ) and hole mass ( $m_e = 0.85$ ). As the  $ZrN_x$  was grown on GaN templates the heterojunction produced a 2DEG along the GaN interface which confirmed the films had polarization properties similar to AlN. The  $ZrN_x$  films were able to produce ohmic contacts without the requirement of high temperature annealing.

## **5.6** AlN and $ZrN_x$ Comparison

Through various film thicknesses, the 2DEG carrier density (Figure 5.19a) and contact resistance (Figure 5.19b) were compared for the purpose of the potential incorporation into a GaN MOSFET. As expected, the 2DEG was more dependent on film thickness for



Figure 5.19: (a) 2DEG carriers density and (b) Contract resistance based on the film thickness of AlN and  $ZrN_x$ .

All compared to  $ZrN_x$ , which remained constant beyond 3 nm. Likewise, the 2DEG was independent of the contact metal for both the AlN and  $ZrN_x$  on unintentionally doped GaN. The contact resistance was independent based on film thickness for both nitride films. The lowest contact tunneling resistance was not the thinnest layer of AlN, which agreed with the  $ZrN_x$ . Conversely the  $ZrN_x$  was not depend on film thickness, however contact metal increased the resistance by an order of magnitude.

The contact resistance (Figure 5.20) of the AlN films were strongly dependent on the 2DEG within the quantum well, which confirmed direct tunneling was the mode of the conduction. Conversely, the low conduction band offset between  $ZrN_x$  and GaN produced a contact that was independent on the 2DEG. Thus, the contact resistance was strongly dependent on the Schottky barrier height that was formed along the metal/ $ZrN_x$  interface. Post-fabrication annealing of these films was shown to significantly reduce the contact resistance of the films. The annealing temperature and time was strongly dependent on the metal stack, which requires further investigation to improve the parasitic contact resistance in the device.



Figure 5.20: Contact resistance of the nitride films compared to the 2DEG formed along the GaN interface.

## 5.7 Nitride Reliability

The reliability of high voltage switching devices and consistent material stability within high electric fields and high temperatures is an ongoing issue [135–137]. The GaN based device has the potential to breakdown much earlier under high electric fields [138]. Conventional low electric field stress measurements have overestimated the true lifetime of the device before electrical degradation occurs. The materials must ensure consistent device operation under high electric fields, where the nominal device will be operating. Optimal reliability tests should examine the gate-source region, to inspect the effects of electrons traveling through a controlled and measurable high electric field. In this study, we examined the reliability of AlN,  $ZrN_x$  and  $ZrO_2$  films incorporated in GaN MOSHCAPs, Figure 5.3, under high electric fields.

The devices were exposed to gate-injected current which was ramped with respect to time, shown in Figure 5.21 As the current was injected through the gate, the voltage was monitored to inspect potential breakdown of the device. This method has been commonly used to evaluate the channel interface for MOS devices. Capacitance-voltage measurements, shown in Figure 5.22, were obtained after each gate current injection step to track the shift of the flat-band voltage. The sensitivity of the flat-band voltage due to defects accurately



Figure 5.21: Measured gate voltage based on injected gate current for different MOS GaN devices. The voltage increased as the materials in the device failed due to defects produced within the films.



Figure 5.22: Capacitance-voltage measurements (100 kHz) for (a)  $AlN/ZrO_2$  (b)  $ZrN_x/ZrO_2$  and (c)  $ZrO_2$  were shifted as breakdown occurred due to electron induced defects within the films.

allowed the ability to track the failure after stressing the device. Multiple frequency measurements (5 kHz - 1 MHz) were used to confirm the flat-band voltage of each device [14]. The flat-band voltage shift was due to the induced defects formed along the interface of the films as electrons were injected through the gate and source dielectric stack [16].

The time dependent current injection measurements were used to track the number of electrons that were sent through the gate films (Figure 5.23). The shift in the capacitance-voltage measurements allowed for the comparison of flat-band voltage with the number of



Figure 5.23: Flat-band voltage shift and the associated traps within the film caused from electrons injected through the gate. Lifetime of the devices based on worst case scenario of 10  $\mu A/mm$  gate leakage. The lifetime ratio remains constant regardless of the leakage current; i.e. The AlN film degraded 8x faster than the  $ZrN_x$  film.

electrons that traveled through the films. Initial fluctuations of the flat-band voltage shift was expected and commonly known as the burn-in effect of the device. Typically hydrogen and mobile ions within the film cause fluctuations until removed through soft biasing (low gate current injection). The induced trap densities within the films were obtained through the shift of the  $\Delta V_{FB}$  [139].

Breakdown was considered to occur once the flat-band voltage of the device migrated 10% from the initial measurement. The lifetime of the devices, shown in Figure 5.23, were based on a constant leakage current traveling through the material over the lifetime of the device. A worst case gate leakage of 10  $\mu A/mm$  was assumed and determined that the AlN would fail after two years. However, at the same current density the  $ZrN_x$  and  $ZrO_2$  films would maintain consistent operation for 8 and 25 times longer than AlN. We believe the large Zr atoms were able to withstand the electrical migration effects compared to the smaller and more mobile Al atoms.

The migration and defect effects were also portrayed through breakdown of the devices. Based on the breakdown voltage, shown in Figure 5.24a, we were able to extract the critical electric field of the films. The AlN films were consistent with literature and withstood electric fields up to 2 MV/cm [140, 141]. Conversely, the  $ZrN_x$  films maintained electric



Figure 5.24: (a) Reverse breakdown characteristics of the AlN and  $ZrN_x$  films to extract the materials critical electric field at 25°C. (b) The materials reverse breakdown based on temperature. Based on the temperature dependence, the  $ZrN_x$  was more stable compared to the AlN.

fields beyond 20 MV/cm. The large critical electric field again demonstrated that  $ZrN_x$  has the capability of operating within high power devices to improve the overall reliability. We believe this is the first reported critical electric field of an ultra-thin  $ZrN_x$  film incorporated within a GaN based device. As the temperature was increased, the voltage required to break the film decreased accordingly. This relationship between critical breakdown and temperature was used to extract the breakdown activation energy of the films [138]. The activation energy represented the films breakdown behavior during higher temperature operation (Figure 5.24b). The activation energy associated with temperature of the AlN films on GaN was consistent with other reported values [136]. In addition, we found that the activation energy of the  $ZrN_x$  was nearly 2 times improved over the AlN based device which provides reliability at higher operating temperatures.

Based on the gate injection stress technique the degradation of the films occurred at different rates. The AlN film was found to be more susceptible to failure than both  $ZrN_x$ and  $ZrO_2$  films. Through reliability projection techniques, the  $ZrN_x$  and  $ZrO_2$  based films provide 8x and 25x lifetime improvements over the AlN based device. Furthermore, the critical voltage of the AlN was determined to be more sensitive to temperature than  $ZrN_x$ . The integration of novel films, such as  $ZrN_x$  and  $ZrO_2$ , has the potential to improve the overall reliability of GaN based devices for high power applications.

## 5.8 Summary

The goal of the source/drain regions within a GaN MOSFET is to provide high carriers to the channel and reduce the contact resistance. Low temperature PEALD AlN and  $ZrN_x$ were examined for the potential use along the source and drain. Electrical measurements confirmed that both of the nitride based films had polarizing behavior, which produced a 2DEG along the GaN interface. Examining the reliability of the nitride films, provided a strong insight into the use of both films within the MOSFET. The less stable, high 2DEG AlN films have the potential for use along the lower electric field portions of the device. These regions are typically beneath the contacts, assuming near ideal ohmic contacts are achieved. Furthermore, the region between the source and gate has shown to have lower electric field strengths compared to the drain region. The  $ZrN_x$  films have the potential to provide 2DEG carriers along the GaN interface, while able to provide high material stability during large electric fields.

The ultra-thin nitride films characterized have shown important properties required within the GaN MOSFET to reduce the contact resistance, provide carriers and device stability. Through the incorporation of both films within the device has to potential to advance the GaN MOSFET for high power electronics with improved device reliability.

## Chapter 6

# GaN MOSFET Fabrication and Characterization

## 6.1 Introduction

The previous sections examined the potential of low temperature PEALD films to replace high temperature processing for GaN MOSFETs. The electrical characteristics of low temperature grown  $ZrO_2$  and  $HfO_2$  gate dielectrics demonstrated significant improvement over current technology and recent literature. Likewise, PEALD AlN and  $ZrN_x$  films were capable of supplying large carrier densities along the GaN interface without the complexity of selective area growth and ion implantation. This chapter presents a demonstration of a novel GaN MOSFET fabricated with ultra-thin PEALD films. The  $ZrO_2$  film was used based on the extracted dielectric characteristics determined in Chapter 4. The  $ZrO_2$  on GaN improved the interface traps, dielectric constant and channel mobility compared to the  $HfO_2$ . The selectively grown AlN films were used for the source and drain region to provide electrons to the channel. The following sections described the techniques and processes used to fabricate the GaN MOSFET with low temperature PEALD films. The GaN MOSFET was characterized through electrical current and capacitance measurements based on gate and drain voltages. The GaN MOSFET has a legitimate attempt to dominate the power electronic market with the incorporation of novel films and fabrication techniques.

## 6.2 Device Fabrication

In this section, the process flow and methods used during each fabrication step were described with the corresponding schematic of the films following each fabrication step. Devices were fabricated on two types of GaN-on-sapphire templates; Si-doped and unintentionally doped GaN. The majority of the device regions were sub-10nm films deposited with a novel low temperature atomic layer deposition technique. The major advantage that PEALD grown films have over other techniques are the atomically consistent film thickness and large scale consistency across the entire wafer. This provided a negligible variation between similar designed devices on the wafer.

#### Alignment Marks

The GaN template and the PEALD films for the gate, source and drain regions were transparent which made layer to layer alignment impossible. To overcome this issue metal alignment marks (Figure 6.1) were deposited across the wafer to ensure each layer was aligned on top of the previous layer correctly. A negative photoresist lift-off technique was used to prevent metal from being in contract where the active region of the devices were.

First photoresist was spun onto the template to produce a 1.3  $\mu m$  film which was followed by a soft bake at 90°C for 120 seconds. The photoresist film was exposed to 356 nm ultraviolet light through the mask for 3 seconds. The wafer was again baked for 90 seconds at 115°C to activate the negative resist properties of the film. Following the second bake, which converted the resist from positive to negative, the template was exposed to UV light for 60 seconds and developed in AZ400K developer for 35 secs. A thin layer (20 nm) of titanium tungsten (TiW) was sputtered on the wafer. The metal of choice for this step was irreverent other than the stickiness to the GaN and a large melting point to withstand various annealing steps. Following the metal deposition, the template was soaked in acetone to remove the resist along with the unwanted metal film. The alignment marks on the mask were designed to ensure the largest misalignment of a device was sub-micro (< 1  $\mu m$ ).



Figure 6.1: Alignment marks were placed on the GaN template to align future layers.

#### Source and Drain Regions

The first film of the device grown on the template was the AlN films along the source and drain region, shown in Figure 6.2. Again, the negative photoresist lift-off technique was used to pattern the regions where AlN would be deposited. The AlN film was deposited with the same technique discussed in Chapter 5. The photoresist and unwanted AlN was removed with acetone which was significantly easier than attempting to control the etching of ultra-thin AlN.

During the growth of the AlN, the photoresist was near the maximum suggested temperature use before potential re-flow began. This resulted in the photoresist adhering to the template compared to room temperature depositions. Following the growth of the film, the templates were submerged into acetone for 10 min with ultrasonic vibrations to ensure all of the resist was removed.



Figure 6.2: The negative photoresist lift-off technique defined the source and drain regions where the ultra-thin ALN films was deposited. The gate region was protected with photoresist to ensure volatile Al atoms did not disturb the GaN interface.

#### Gate Dielectric

The gate dielectric was defined in the same fashion as the source and drain regions. The negative photoresist lift-off techniques provided a quick process to produce the selective deposition of ultra-thin  $ZrO_2$  films. As mentioned earlier, the 40 cycle PEALD  $ZrO_2$  film was used as the gate dielectric for the GaN MOSFET. This films was used to reduce the leakage current and improve the control of the channel. The advantage of using the  $ZrO_2$  film was on the order of  $10^{10} \ eV^{-1}cm^{-2}$ . In addition the film was able to handle a maximum gate voltage of 4 V.



Figure 6.3: The regions for the  $ZrO_2$  gate dielectric were defined with same negative photoresist process that defined the source and drain regions. The  $ZrO_2$  overlapped the ALN along the source and drain to ensure the gate metal would not come into contact with AlN.

The growth of the gate dielectric was grown at  $100^{\circ}C$  reducing the adhesion of the photoresist to the GaN template. This reduced the time length of the templates required in the acetone to lift-off the film. Furthermore, the ultrasonic vibrations were not required to remove the unwanted photoresist.

## Metalization

With all of the PEALD films selectively deposited in the specific regions, the gate, source and drain contact metals (Figure 6.4) were defined with negative photoresist for metal liftoff. Again the lift-off technique provided a quick and efficient method to define the contacts without the use of chemical etching. The major advantage of using a single mask for the source, gate and drain was the self-aligned gate process. Regardless of the misalignment between the layers the contacts were consistently the designed distance apart. This provided more consistent device characteristics, however the optimal source/drain contact was not necessarily the ideal gate metal and vise versa.

#### $SiN_x$ Encapsulation

During the encapsulation fabrication step of the device, the  $SiN_x$  was deposited by PECVD at 300°C for 15 minutes. The purpose of this step was to protect the device from contaminates on the surface, as well as allowing for isolated contact pads for testing. Growing the film at a higher temperature was intended to simultaneously active the gate dielectric. The gate dielectric annealing typically removes unwanted mobile ions within the film along with



Figure 6.4: The source, drain and gate contacts were all deposited at the same time to reduce the fabrication steps of the device.

reducing interface defects which lead to threshold voltage shifts. Furthermore, the improvement of the relative dielectric constant of the gate dielectric was demonstrated through post fabrication annealing [142,143]. This in turn will improve the potential current density and channel mobility of the MOSFET.

## 6.3 Device Characterization

The GaN MOSFETs were fabricated with the process described above to compare the material properties behavior within a fully fabricated device. Most notably, the behavior based on the doping concentration of the GaN templates were examined in attempt to reduce the substrate leakage and ensure enhancement-mode behavior. The metalization layer (Cr and Al) was examined to determine the quality of the depletion effect under the channel based on the doping. The  $100\mu m \ge 100\mu m$  GaN MOSFET devices, shown in Figure 6.5, were characterized to ensure the contact resistance was minimized by the large source and drain contacts pads. To characterize the GaN MOSFET FatFETs, conventional current-voltage and capacitance-voltage measurements were obtained with the Keithley 4200-SCS. The GaN MOSFET device characteristics were organized into two sections; one which examined the on-state regime of the device while the other examined the transistors off-state and breakdown voltage. To evaluate the on-state portion of the transistor the threshold voltage, sub-threshold slope, density of interface traps, peak current density and electron field effect mobility were examined. The threshold voltage of the MOSFET provided the information regarding the transistor operation; i.e. normally-on or normally-off. Furthermore, the sub-threshold slope,  $D_{it}$  and channel mobility were used to determine the quality of the metal/dielectric and GaN interface.



Figure 6.5: Confocal image of the  $100\mu m$  x  $100\mu m$  FatFETs used to evaluate the quality of the GaN MOSFET fabricated with the ultra-thin PEALD films. The gate was extended  $10 \ \mu m$  beyond the width of the source and drain to reduce the substrate leakage current.

## 6.3.1 GaN MOSFET Operation

The on-state operation of the GaN MOSFETs were examined with respect to different materials incorporated within the devices. The background doping, gate metal and device dimensions were adjusted to determine the effects for the potential use in power electronics. Furthermore, the intrinsic transconductance of the GaN MOSFET was extracted to determine the true potential of the device with commercially fabricated source and drain interconnect contacts. The devices were also exposed to post fabrication annealing to examine the effect of oxide degradation and contact resistance reduction.

#### Substrate Doping

The quality of the off-state has a strong dependence on the threshold voltage and substrate leakage current of the device. To produce an enhancement-mode MOSFET, the accumulation of carriers must be limited to gate voltages greater than zero. In addition, the substrate leakage of the device must be minimal to prevent conduction at a gate voltage of  $V_{GS} = 0$ V. The leakage current was based on the depletion width of the channel which was strongly dependent on the doping concentration of the GaN, shown in Figure 6.6. FatFET GaN MOSFETs were examined to determine the threshold voltage and off-state leakage current of the devices based on the background doping of the GaN templates.

Capacitance-voltage measurements, shown in Figure 6.7, confirmed the accumulation of electron were consistent with the doping of the templates,  $V_{acc} = 0.4$  V and  $V_{acc} = 1.3$  V for



Figure 6.6: A representation of the leakage current under the channel based dependence on the depletion width (dashed region). The lower doped GaN produced a larger depleted channel which was expected to create a lower leakage current.

N<sup>+</sup> GaN and N<sup>-</sup> GaN, respectively. Based solely on the extracted accumulation voltage of the MOS channels would suggest that both substrates would produce enhancement-mode MOSFETs. However, based on the transfer characteristics of the GaN MOSFET, shown in Figure 6.7b, the N<sup>+</sup> GaN was unable turn-off the device and control the channel. The N<sup>+</sup> GaN template was unable to control the substrate leakage current, due to the low resistivity (< 0.05  $\Omega$  cm) of the GaN template. While the device was off ( $V_{GS} < 0$  V), the current would flow around the depleted gate region. The depletion width discussed in Chapter 2



Figure 6.7: (a) The inverse squared capacitance with respect to gate voltage for the  $N^+$  GaN (red) and  $N^-$  GaN (green). The x-axis intercept was used to extract the accumulation voltage of the channel MOS. (b) The drain current (red) at  $V_{DS} = 0.1$  V for  $N^+$  GaN with respect to gate voltage.
required less than 50 nm to ensure a fully depleted channel in the N<sup>+</sup> GaN. This confirmed the 4.3  $\mu m$  N<sup>+</sup> GaN was not suitable to produce an enhancement-mode GaN MOSFET. Conversely, the resistive N<sup>-</sup> template was able to significantly reduce the substrate leakage current during the off state of the device.

The reduced off-state current, allowed for the enhancement-mode behavior of the MOS-FET (Figure 6.8) to be achieved with a positive gate bias. The extracted doping concentration  $(1.4 \times 10^{14} \ cm^{-3})$  obtained from the slope of the  $1/C^2 - V$  suggested the chromium gate would produce a depleted channel depth of 2.9  $\mu m$ . The partially depleted channel depth was close to the range of the N<sup>-</sup> GaN template thickness 5 +/- 1.5  $\mu m$ . However, the leakage current was a result of the gate unable to fully deplete the GaN under the channel.

The drain saturation voltage,  $V_{D_{sat}}$ , of long channel devices (6.1) can be approximated for a MOSFET architecture. The family of curves (Figure 6.8a) and the MOSFET transfer curves (Figure 6.8b) were used to determine the threshold voltage of the MOSFET. The threshold voltage obtained from the drain saturation voltage was estimated as 1.4 +/- 0.1 V and consistent to the x-axis intercept obtained from the transfer curve (1.35 V). The GaN MOSFET further confirmed enhancement-mode operation as the drain current increased with positive gate bias. The on-resistance (6.2) was extracted from the linear regime of family of curves during on-state operation at  $V_{GS} = 3$  V. The specific on-resistance was determined as  $32.9 \ \Omega \cdot cm^2$ . This was roughly  $10^3$ x larger than current technology, which was



Figure 6.8: (a) The family of curves for the  $N^-$  GaN MOSFET. The drain saturation voltage of the MOSFET was consistent with conventional MOSFET models where  $V_{Dsat} = V_{GS} - V_t$ . (b) The drain current (green solid) and transconductance (blue dashed) at  $V_{DS} = 1$  V for  $N^-$ GaN with respect to gate voltage. The device demonstrated enhancement-mode operation, with a turn-on voltage of 1.35 V.

a result of the large parasitic resistances at the source and drain interconnects. Through optimized metalization the on-resistance of the device would be managed to a value more comparable to literature.

$$V_{D_{sat}} = V_{GS} - V_{th} \tag{6.1}$$

$$R_{on} = R_{tot} \cdot W \cdot (L_G + L_{GS} + L_{GD}) \tag{6.2}$$

#### Gate Metal Stack

It is well known that the threshold voltage strictly dictates the required gate voltage for the MOSFET transistor to be considered on and off. The ability to adjust the threshold voltage is important for design purposes of the transistors behavior. Depending on the device application, the precision of the threshold voltage can be crucial. For power electronics, the ability to ensure a positive and consistent threshold voltage with respect to devices across a wafer is important. The threshold voltage (6.3) can be designed based on the metal-semiconductor workfunction difference ( $\Phi_{ms}$ ), the interface traps ( $Q_{it}$ ), depletion charge ( $Q_{Dep}$ ), gate dielectric ( $C_{ox}$ ) and the surface potential ( $2\phi_F$ ). The interface traps are conventionally minimized to reduce the interface traps becoming an issue. Furthermore, the depletion and surface potential and capacitance density are typically predetermined as the semiconductors background doping and gate oxide are known. The most simplistic method to control the threshold voltage of the transistor was through an adjustment the gate metal workfunction.

$$V_{th} = \Phi_{ms} - \frac{Q_{it}}{C_{ox}} - \frac{Q_{Dep}}{C_{ox}} - 2\phi_F$$
(6.3)

The gate metal workfunction strongly dictated the depletion depth beneath the channel. For example, a Al gate metal was expected to deplete only 1.7  $\mu m$ , whereas a Cr gate metal depleted nearly twice as much 2.9  $\mu m$ . Based on the depletion depth, the Cr metal was expected to produce a larger turn-on voltage compared to the Al gate. The MOSFETs turnon voltage (threshold voltage) was determined from the transfer curves; the gate voltage  $(V_{GS})$  was swept from -3 V to 3 V while the drain voltage  $(V_{DS})$  was held constant at 1 V. The drain current and transconductance was measured with respect to the gate voltage, shown in Figure 6.9. The difference in workfunction of the metals from Al to Cr shifted the



Figure 6.9: The transfer and transconductance characteristics of the GaN MOSFET had a strong dependence with the work function of the gate metal. When the gate metal was exchanged from Al ( $\Psi_{Al} = 4.2 \ eV$ ) to Cr ( $\Psi_{Cr} = 4.7 \ eV$ ) the MOSFET characteristics shifted from normally-on ( $V_t = -0.25 \ V$ ) to normally-off ( $V_t = 1.3 \ V$ ). The enhancementmode MOSFET improved current density (2x) and transconductance (5x) over the Al gate device.

GaN MOSFET behavior from a depletion-mode transistor to an enhancement-mode device, respectively.

The enhancement-mode transistor produced a device turn-on voltage of 1.3 V, whereas the Al gate required a negative gate voltage to turn the device off. Furthermore, the Cr gate allowed for the MOSFET to produce a larger peak current density compared to the depletion-mode Al gate device, which was also confirmed from the transconductance. As expected, the larger transconductance MOSFET produced the larger peak current density. Furthermore, the substrate leakage current in the reverse gate bias was equivalent. Larger drain biases on the N<sup>-</sup> GaN templates allowed current to conduct around the gate depletion region. However, the reduced peak current density agreed with a depletion-mode device and a potential indication of an undesirable interface defects caused from gate metal.

To confirm the channel interface quality, the sub-threshold slope of the MOSFETs were extracted at low drain biases ( $V_{DS} = 0.2$  V). The sub-threshold slope (6.4) of the devices were extracted from the transfer curves during the transistors off-state. The inverse slope of the logarithmic drain current with respect to gate voltage is an excellent method of determining the quality of the gate dielectric interface [144, 145]. The sub-threshold slope (S), shown in Figure 6.10, confirmed the Al gate degraded the dielectric interface compared to Cr metal. The Cr gate produced a sub-threshold slope of 123 mV/dec which was 7x lower



Figure 6.10: The sub-threshold slopes of the GaN MOSFETs were extracted from the low transverse electric field transfer characteristics for the Al (Green) and Cr (Blue) gate metals. The enhancement-mode GaN MOSFET produced a sharper sub-threshold slope compared to the depletion-mode transistor.

than the Al device (956 mV/dec). This was an indication that the channel of the Al device was degraded from the  $300^{\circ}C$  SiN encapsulation process. Furthermore, the sub-threshold drain leakage current was 3x lower for the Cr gate compared to the Al gate. Again, the smaller depletion width caused leakage issues for the Al gate devices.

$$\frac{1}{S} = \frac{d}{dV_{GS}} log(I_D) \tag{6.4}$$

The sub-threshold slope (6.5) was approximated by the differential capacitance of the depletion  $(C_{Dep})$ , the interface traps  $(C_{it})$  and the dielectric capacitance  $(C_{ox})$  [14,16]. The sub-threshold slope (6.6) was further expanded to only have the density on interface traps as an unknown variable. Rearranging the expression provided an insight into the density of interface traps (6.7) along the channel. Based on the sub-threshold slope of the GaN MOS-FETs, the extracted  $D_{it}$  for Cr and Al were  $2.5 \times 10^{11} cm^{-2} eV^{-1}$  and  $4.8 \times 10^{12} cm^{-2} eV^{-1}$ , respectively. The extracted  $D_{it}$  from the sub-threshold slope has been shown as a method of comparison between devices, however the conductance frequency measurements are a more accurate method for extracting the interface trap density.

$$S = \frac{kT}{q} ln(10) \left[ 1 + \frac{C_{Dep} + C_{it}}{C_{ox}} \right]$$
(6.5)

$$S = \frac{kT}{q\log(e)} \left[ 1 + \frac{qD_{it} + \left(\frac{qN_A\epsilon_s}{3\phi_F - 2\frac{kT}{q}}\right)^{1/2}}{C_{ox}} \right]$$
(6.6)

$$qD_{it} = C_{ox} \left(\frac{qS\log(e)}{kT} - 1\right) - \left(\frac{qN_A\epsilon_s}{3\phi_F - 2\frac{kT}{q}}\right)^{1/2}$$
(6.7)

Capacitance-voltage measurements, shown in Figure 6.11a, were used to confirm the transistor characteristics of the gate metals. The capacitance with respect to gate voltage was used to determine the flatband voltage beneath the channel. These values were comparable to the extract turn-on voltage of the MOSFETs based on the current-voltage measurements. Furthermore, the lower depletion capacitance Cr gate was able to reduce the number of carriers within the channel during the off-state of the device compared to the



Figure 6.11: (a)The gate-source capacitance of the oxide was obtain at low frequency (50 kHz) to determine the peak capacitance density and extract the flat band voltage of the MOS stack, shown in the inset. During reverse gate bias, the Cr metal was capable of depleting more charge from under the gate compared to the Al gate stack. (b) The density of interface traps were obtained from the conductance frequency spectrum. The  $Al/ZrO_2$  gate stack degraded the interface, which produced a 5x more traps compared to the Cr metal.

Al gate metal. This confirmed that the larger workfunction metal was able to depleted the channel most effectively. The conductance frequency spectrum was used to extract the density of interface traps along the channel through the conductance frequency spectrum [16]. The conductance method, shown in Figure 6.11b, and sub-threshold slope confirmed the Al degraded the channel an device characteristics compared to the Cr gate metal. The extracted density of interface traps confirmed the Al gate metal produced more defects (5x) along the GaN channel following the encapsulation of the devices.

The density of interface traps along the channel, shown in Table 6.1, were extracted from three different techniques; the sub-threshold slope, conductance dispersion and the C-V hysteresis. As expected, the techniques produced slightly different results based on the measurement used to extract the  $D_{it}$ . However the methods which included the AC signal was more consistent to the results discussed in Chapter 4. The sub-threshold  $D_{it}$  extraction was the least accurate for these transistors because of the excess substrate leakage around the depletion region. This slightly increased the overall  $D_{it}$  compared to the frequency dependent techniques. However, all of the extraction techniques confirmed that the Al gate metal diffused more readily through the gate dielectric which increased the density of interface traps compared to the Cr gate metal.

Table 6.1: The Extracted Density of Interface Traps

Extraction	$Al/ZrO_2/GaN$	$Cr/ZrO_2/GaN$
Technique	$(cm^{-2}eV^{-1})$	$(cm^{-2}eV^{-1})$
Sub-threshold Swing	$4.8 \mathrm{x} 10^{12}$	$2.5 \mathrm{x} 10^{11}$
Conductance Dispersion	$1.0 \mathrm{x} 10^{11}$	$1.7 \mathrm{x} 10^{10}$
C-V Hysteresis	$1.1 \mathrm{x} 10^{11}$	$4.8 \mathrm{x} 10^{10}$

The design of the GaN MOSFET architecture utilized the simplicity of the gate metal workfunction to control the threshold voltage of the device. This was demonstrate through the modification of the gate metal workfunction from 4.2 eV to 4.7 eV by using Al and Cr, respectively. The Al gate metal produced a depletion-mode GaN MOSFET due to the MOS flatband voltage occurring at equilibrium ( $V_{GS} \approx 0$  V). Conversely, the Cr metal depleted the MOS channel at equilibrium, which ensured the transistor was a normally-off enhancement-mode MOSFET. The sub-threshold slope and frequency dispersion confirmed the Cr gate metal stack reduced diffusion and interface traps along the channel during fabrication. The reverse voltage depletion capacitance confirmed that the Al was unable to removed the charge under the channel like the enhancement-mode transistor.

#### **Channel Mobility Extraction**

The extracted field effect channel mobility, has a strong dependence on the quality of the ohmic contacts and parasitic resistance along the source and drain. The parasitic resistances, depicted in Figure 6.12, have the potential to significantly reduce the true extracted electron mobility of the channel. The source parasitic resistance degrades the internal gate voltage between the actual source and gate of the transistor. The internal gate voltage (6.8) can be expressed by the voltage drops across the entire device and the unintentional source resistance. Furthermore, the internal drain voltage (6.9) was determined from the applied drain voltage and the resistance of the source and drain.

The internal transconductance (6.10) of the GaN MOSFET was extracted based on the variation of the drain current with respect to the internal gate voltage. This transconductance was more indicative to the true behavior of the MOSFET with optimized source and drain interconnects. Furthermore, the field effect mobility of the transistor based on the internal transconductance represents the MOSFET characteristics compared to the external parasitic results.



Figure 6.12: The internal voltage drops are dependent on the external voltage biases and parasitic resistances along the source and drain interconnects. The MOSFET only experienced the internal bias for the drain  $(V_D)$  and gate  $(V_G)$  based on the external gate and drain voltages.

$$V_G = V_{GS} - I_D R_S \tag{6.8}$$

$$V_D = V_{DS} - I_D \left( R_S + R_D \right)$$
(6.9)

$$g_{m0} = \frac{\partial I_D}{\partial V_G} \tag{6.10}$$

The interconnect parasitic resistance was obtained from TLM measurements discussed in Chapter 5. The additional resistance was roughly 6 orders of magnitude larger than conventional contacts incorporated in conventional GaN devices. This led to large voltage drops across the source and drain contact regions. Furthermore, the variation between the transconductance and internal transconductance, shown in Figure 6.13a, were very different. The extracted external transconductance  $(g_m)$  for the FatFET peaked at 33  $\mu S/mm$  due to the large resistance at the source contact. However, the intrinsic transconductance  $(g_{m0})$ suggested that the GaN MOSFET could produced a transconductance of 0.73 mS/mm if the parasitics were minimized. The peak gate capacitance density was also limited because of the parasitic source resistance, shown in Figure 6.13b.



Figure 6.13: (a) The external and internal transconductance of the 100  $\mu m \ge 100 \ \mu m$  GaN MOSFET at  $V_D = 3$  V. The peak intrinsic transconductance was 0.73 mS/mm. (b) The capacitance-voltage based on the external and internal gate voltage.

The external transconductance of the 100  $\mu m \ge 100 \mu m$  GaN MOSFET was extremely promising as the peak transconductance was around 40  $\mu S/mm$ . This was comparable to smaller channel GaN MOSFET devices with optimized contacts transconductance 12.5  $\mu S/mm$  ( $L_G = 16\mu m$ ) and 7.5  $\mu S/mm$  ( $L_G = 40\mu m$ ) at equivalent drain voltages [96]. The ratio between the transconductance and gate lengths of the devices (6.11) strongly implied that the  $ZrO_2$  PEALD GaN MOSFET improved the combination of the capacitance density and channel mobility ( $\mu_{eff}C_{ox}$ ). The  $ZrO_2$  GaN MOSFET was compared to  $L_G$ = 16 $\mu m$  and  $L_G = 40\mu m$  GaN MOSFETs where  $\mu_{eff}C_{ox}$  was improved by 25x and 16.7x, respectively. The measured capacitance  $ZrO_2$  gate dielectric improved the  $C_{ox}$  over the  $SiO_2$  gate dielectric by a factor of 5.7. This implied that the peak channel electron mobility was improved and surpassed 200  $cm^2/Vs$ .

$$\frac{g_m L}{W} = \mu_{eff} C_{ox} \tag{6.11}$$

In addition, the GaN MOSFET intrinsic transconductance (0.73 mS/mm) was larger than a 200  $\mu m$  GaN HFET (0.42 mS/mm) with reduced parasitic resistances [146]. The improved  $ZrO_2$  gate dielectric interface GaN MOSFET intrinsic transconductance produced an equivalent value to an optimized 20  $\mu m$  GaN MOSFET with  $SiO_2$ . The effective electron mobility (6.12) from the channel was extracted from the low drain voltage, drain current  $(I_{DS})$ . The gate length (L), width (W), charge density  $(Q_{tot})$  were all known and obtained through confirmation of optical microscope or capacitance-voltage measurement.

$$\mu_{eff} = \frac{L}{WQ_{tot}} \frac{I_{DS}}{V_D} \tag{6.12}$$

The intrinsic extracted mobility, shown in Figure 6.14, produced a peak mobility of 480  $cm^2/Vs$  whereas the parasitic resistance reduced the extracted mobility to 240  $cm^2/Vs$ . This extracted channel mobility was equivalent with some of the leading GaN MOSFET channel mobilities [37, 147]. However, with a reduced source parasitic resistance the GaN MOSFET mobility has the ability to surpass literature. As expected, the improved transconductance was in-part due to the improved channel mobility of the  $ZrO_2$ /GaN gate dielectric. With the incorporation of the high quality  $ZrO_2$  dielectric with improved source/drain metalization, the GaN MOSFET has a viable opportunity in the transistor industry.



Figure 6.14: The effective electron mobility of the FatFET obtained at low drain bias. The mobility based on the external gate biases was determined to have a peak mobility of 230  $cm^2/Vs$  Conversely, the peak electron mobility of the GaN MOSFET without parasitic resistance has the potential to reach 480  $cm^2/Vs$ .

#### **Post Fabrication Annealing**

To reduce the parasitic resistances at the source and drain, the tunneling contacts were required to be significantly lower. The most simplistic and cost effective technique was achieved through a post fabrication anneal of the devices. The devices were annealed for 15 minutes at  $400^{\circ}C$ ,  $450^{\circ}C$ ,  $500^{\circ}C$  and  $550^{\circ}C$  in a nitrogen rich environment. Between each annealing temperature, the electrical characteristics of the devices were obtained to extract the threshold voltage, on-resistance, peak current density and peak transconductance from the family of curves and transfer curves.

As the annealing temperature was increased the gate leakage with respect to gate voltage, shown in Figure 6.15a, became significantly degraded. Beyond  $500^{\circ}C$  the depletion gate leakage ( $V_{GS} < 0$  V) was the first indication that severe dielectric degradation occurred. This was most likely due to the Cr atoms from the gate metal diffusing into the  $ZrO_2$  gate dielectric. These metal atoms behave as conduction site within the film, which allow current to flow through the dielectric more readily. The extracted density of interface traps with respect to annealing temperature, shown in Figure 6.15b, used the conductance extraction,



Figure 6.15: (a) The gate leakage current of the FatFET as the gate voltage was swept from reverse to forward bias. Higher annealing temperatures increased the conduction leakage through the dielectric. (b) The extracted density of interface traps based on annealing temperature. The gate oxide was electrically activated between  $400^{\circ}C$  and  $500^{\circ}C$  before the gate dielectric was critically degraded.

low frequency hysteresis and high frequency Terman shift to evaluate the degradation of the gate dielectric. The activation temperature of the  $Cr/ZrO_2/GaN$  MOS stack was found to be within the range of  $400^{\circ}C$  and  $500^{\circ}C$ . Within this range the  $D_{it}$  and the reverse bias gate leakage was minimized before gate dielectric degradation.

The diffusion through the gate dielectric was a good indication that the Cr was able to diffuse through the AlN to improve the contact resistance. This was confirmed through the family of curves, shown in Figure 6.16a, which improved the peak drain saturation current density. The low drain voltage produced high gate leakage, however smaller devices and contact annealing would have reduced this effect. At the peak gate voltage ( $V_{GS}$ = 3 V) the drain saturation current density was increased from 0.04 mA/mm to 0.17 mA/mm as the annealing temp went from 300°C to 500°C. The 4x agreed with the expected reduction of the contact resistance, discussed in Chapter 5. Furthermore, the normalized onresistance obtained from the linear region of the family of curves produced an non-annealed on-resistance to 10.6  $\Omega$  cm<sup>2</sup> ( $\downarrow$  3x). The new on-resistance was still very large; however the annealing demonstrated that the reduction of the parasitic resistance will further improve the device characteristics over other Gan MOSFET devices.

The GaN MOSFET transfer characteristics (Figure 6.16b) based on annealing temperature improved the current density and transconductance compared to the non annealed device. Through temperature activation of the gate dielectric, the threshold voltage was



Figure 6.16: (a) The FatFET family of curves after the device was annealed at  $300^{\circ}C$  (red) and  $500^{\circ}C$  (blue). The drain current was improved by a factor a 4x over the low temperature annealed device. (b) The transfer characteristics after the device was annealed at  $300^{\circ}C$  (red) and  $500^{\circ}C$  (blue). The transconductance quadrupled over the original fabricated device due to the reduce series resistances.

shifted closer to the nominal expected value. This shift of the threshold voltage was not the major factor in the increased transistor properties. Rather the improved contact resistances reduced the degradation of the device characteristics which allowed the transconductance of the GaN MOSFET to increase to 0.08 mS/mm. Comparing the peak mobility of the GaN MOSFET discussed earlier, the 100  $\mu m$  channel with a 80  $\mu S/mm$  transconductance would be expected to obtain a mobility in the range of 433  $cm^2/Vs$ . This was confirmed that the trend was towards the intrinsic transconductance mobility with optimal ohmic contacts, shown in Figure 6.14.

Through post fabrication annealing, the electrical characteristics of the enhancementmode GaN MOSFET were improved at  $500^{\circ}C$ . Beyond this temperature the gate dielectric was unable to prevent conduction through the barrier. With an optimized annealing and process flow the contact resistance will be minimal without the degradation of the gate dielectric and channel.

#### Gate Length Reduction

The GaN MOSFETs test structures discussed earlier were extremely long channeled devices compared to commercially fabrication products. To obtain large current densities, the gate length must be shortened. It has been shown that the benefit of reducing the gate length also arises from the reduced gate leakage current. One of the major advantages of scaling the gate length by L, was the reduced gate leakage in ultra-thin gate dielectrics [116]. This further improves the undesired losses which increase the efficiency of the transistor. In addition, the switching frequency and device area are improved through the shrinking of the gate length. To examine the potential improvement, the gate length of the enhancementmode GaN MOSFET was reduced to 10  $\mu m$  and compared to the 100  $\mu m$  FATFET. The goal of the decreased gate length was to improve the gate leakage, current density and transconductance.

$$Y_{AC} = 2 \frac{\tanh(\gamma \frac{L}{2})}{Z_0} \tag{6.13}$$

The AC gate admittance (6.13) was formulated based on a transmission line circuit model for ultra-thin leaky gate dielectrics [116]. The AC gate admittance  $(Y_{AC})$  was defined by the characteristic intrinsic impedance (6.14) and the propagation delay (6.15). The intrinsic impedance and the propagation delay were defined by the normalized channel series resistance  $(r_s)$ , the normalized tunneling resistance  $(r_t)$ , the normalized capacitance density (C) and frequency  $(\omega)$ . Taking the frequency to zero, the DC leakage admittance was approximated through Taylor series expansion of the dependence of gate length was simplified in (6.16) and predicted that the reduced gate length of L would result in a reduction of the gate leakage by  $L^2$ .

$$Z_0 = \sqrt{\frac{r_s r_t}{1 + j\omega C r_t}} \tag{6.14}$$

$$\gamma = \sqrt{\frac{r_s}{r_t} + j\omega C r_s} \tag{6.15}$$

$$\frac{Y_{DC}}{L} \approx \frac{1}{r_t} - \frac{1}{12} \frac{r_s}{r_t^2} L^2$$
 (6.16)



Figure 6.17: (a) The leakage current through the gate dielectric with respect to gate voltage. The leakage current was reduced by a factor of 100 as the gate length was reduced from 100  $\mu m$  to 10  $\mu m$ . (b) The GaN MOSFETs extracted intrinsic transconductance were compared based on gate length. The 10  $\mu m$  channel increased the intrinsic transconductance to 15 mS/mm over the 100  $\mu m$  GaN MOSFET channel.

The gate leakage current of the 10  $\mu m$  gate length, shown in Figure 6.17a, confirmed the gate leakage was reduced  $L^2$  compared to the 100  $\mu m$  FATFET. The leakage current obtained for both cases were relatively small compared to the drain current; however, the power efficiency has the potential to be improved as the transistor leakage was reduced by 100x.

The peak drain current of the devices was masked by the parasitic resistances at the source and drain. Thus, the intrinsic transconductance of the MOSFET, shown in Figure 6.17b, were compared to examine the effect of gate length. By reducing the gate length by a factor of 10x, the intrinsic transconductance improved by 20x. This extracted transconductance was better than conventional low-dielectric gate oxides with large density of interface traps and low capacitance densities. Again, with the advancement of traditional ohmic contacts, the novel GaN MOSFET will surpass current GaN transistors.

#### 6.3.2 Breakdown Voltage Characterization

The off-state maximum drain voltage has become a very important entity of the power transistor. Without the ability to withstand large drain voltages the device cannot support high power electronic applications. Numerous techniques have been examined to suppress the electric fields along the drain to improve the voltage rating of the device. The most common technique for GaN HFETs are optimized gate biased field plates exposed along the drain region [25,60,148]. Other non-conventional techniques have demonstrated methods that removed the semiconductor where the peak electric field occurred [27], grounded multiple field plates [62], or grounded source and gate staircase field plates along the drain [149]. These techniques have been able to improve the voltage rating of HFET architectures; however, the GaN MOSFET has the potential to efficiently remove the carriers from the channel and withstand large drain voltages. The fabricated GaN MOSFET architecture was examined to determine the effect of device breakdown with respect to transistor architecture, device dimensions, template background doping and field plate structures. The variation of these parameters allowed for an in-depth understanding of the quality of the transistors off-state.

The off-state drain current of the GaN MOSFET was examined through high voltage drain biasing while the device was considered off. The source and gate of the transistor, shown in Figure 6.18, were grounded whilst the drain was swept from 0 V to 400 V. The drain and source currents were measured to determine the potential voltage rating of the devices. The conventional standard for GaN power devices considered device breakdown to occur when the off-state drain current reached 80  $\mu A/mm$  [25]. Beyond this drain current



Figure 6.18: A confocal image of the GaN MOSFET used to measure the breakdown voltage. (a) The dashed lines are where the transparent films were patterned and deposited. (b) The source and gate contacts were grounded while the drain voltage was increased. The confocal image allowed for the transparent  $ZrO_2$  and ZrN films to be imaged. The AlN optical properties were identical to GaN making imaging nearly impossible.

the device was no longer consider off. This universal standard has allowed easy comparison of breakdown voltage between similar GaN based devices.

#### GaN Template Doping

The drain current with respect to drain voltage was obtained for three different GaN templates, shown in Figure 6.19. Similar to the threshold voltage, the breakdown voltage has a strong dependence on the background doping of the GaN. The silicon doped GaN template was unable to prevent current flow as soon as the drain was bias; which was indicative from the operation examined earlier. The  $N^-$  GaN partially depleted the channel to push the off-state breakdown until 12 V. The on-state provided a demonstration of enactment-mode behavior, however the  $N^-$  GaN was unable to prevent off-state leakage current for high power electronics. Conversely, the semi-insulating GaN templates were able to form a fully depleted channel under the gate. This reduced the off-state drain leakage current and significantly increased the maximum drain voltage. For a GaN MOSFET to have the potential for power electronics, the background doping caused from film defects during growth must be minimal.

The examination of background doping confirmed simulations, discussed in Chapter 2, that during the off-state a low electron concentration of carriers along the channel are critical GaN power MOSFETs. The following sections examined the effects of substrate leakage with respect to the device architecture and device dimensions.



Figure 6.19: The substrate leakage current with respect to GaN background doing as a function of drain voltage. The semi-insulating GaN was capable of preventing the substrate leakage current beyond 50 V.

#### **Gate-Drain Separation**

The simulations discussed in Chapter 2 predicted that the breakdown voltage of the transistor was strongly dependent on the distance between the gate and drain contacts. Two transistor designs, shown in Figure 6.20, were used to compare the ability to withstand the electric fields along the drift region. The conventional MOS-HFET and MOSFET designs utilized the  $ZrO_2$  gate dielectric (6.9 nm) to reduce the leakage current to the gate. Both transistors utilized the AlN films along the source and drain regions to provide carriers to the device. The significant variation between the two transistor architecture, was the AlN (5.7 nm) along the channel in the MOS-HFET. This increased the overall minimum carrier concentration at zero gate bias, which prevented the device from withstanding large drain voltages. Conversely, the MOSFET has been shown to reduce the minimum carrier concentration during the depleted channel which improved the breakdown voltage [46,150].



Figure 6.20: The GaN MOS-HFET and MOSFET transistor architectures were compared for the potential maximum drain voltage before the leakage current becomes 80  $\mu A/mm$ . The significant variation between transistors was the ultra-thin AlN film under the gate metal.

The current density of the transistors were obtained as the drain voltage was swept to 200 V, shown in Figure 6.21a. The HFET architecture was determine to reach the breakdown current below 20 V for all gate-drain separations. This suggested that the GaN HFET was unable to fully the deplete the channel (i.e. depletion mode transistor) compared to the GaN MOSFET (i.e enhancement-mode transistor). The additional resistance along the drain from the GaN MOSFET in-conjunction with the lower carrier concentration beneath the channel improved the breakdown voltage. The drain leakage current was considered as a soft-breakdown due to the lack of a p-type GaN region beneath the gate. However, the semi-insulating GaN channel was able to reduce the leakage current sufficient to prevent significant conduction before 200 V. The limitation of the Keithley 4200-SCS prevented extremely high voltage measurements beyond 400 V. The leakage current and drain voltage

were compared in log scales to extrapolate the potential breakdown voltage of the devices. The electric field strength was significantly lower than the maximum critical field strength of GaN which allowed for the extrapolation of the leakage current to be assumed as the mechanism for device breakdown. This device breakdown voltage was not catastrophic breakdown caused from high electric fields. Instead the device was considered to no longer



Figure 6.21: (a) The measured drain leakage current during the off-state of the GaN MOS-FET. As the gate-drain length was increase, the device was capable of improving the maximum drain voltage. (b) The extrapolated breakdown voltage trend for the GaN MOSFET (red square) with respect to the drift region. The breakdown voltage of the leading GaN HFET device (green star) and the simulated GaN MOSFET (blue) were included.

be off due to the depletion leakage under the channel caused from sub-critical electric field strengths.

The comparison of the extrapolated breakdown with respect to drift region (gate-drain) length, shown in Figure 6.21b, provided the average electric field strength along the drain. The GaN MOSFET without field plate architectures were able to maintain a 93  $V/\mu m$  electric field strength along the drain. This was larger than the expected simulations due to the recessed AlN drain, shown in Figure 6.20. The extract depleted region along the drain improved the breakdown voltage over the highly doped drain obtained from simulation. The average field strength along the drain of the recessed drain and the leading GaN HFET design with optimized field plates and gate stacks [25] were found to produce an equivalent drain voltage rating. Moreover, the HFET required a source and gate field plate to minimize breakdown effects compared to the inherently normally-off MOSFET without field plates.

The inherent MOS design has a preferential off-state compared to the HFET architectures, which required optimized structures to obtain equivalent drain voltage ratings. The MOS incorporated within a GaN power transistor and the appropriately designed field plates will significantly reduce the leakage current and suppress the peak electric fields to ensure high voltage applications are possible.

#### Field Plate Architecture

The leading GaN MOS-HFET breakdown voltage shown in Figure 6.21b, was capable of suppressing the electric fields because of the inclusion of a field plate along the drain. It has been shown that a field plate along the drain has the potential to reduce the electric field along the gate-drain edge by half [21, 46, 59, 151]. Based on the optimal height, length and dielectric, the field place has to ability to prolong the off-state of the transistor before breakdown by suppressing the peak electric field by more than half [46, 150]. To improve the drain voltage rating of the PEALD GaN MOSFET, floating field plates were fabricated on the  $ZrO_2$  film midway in the drift region, shown in Figure 6.18. The metal field plates were self aligned during the metalization step of the gate, source and drain contacts. The floating field plate was to behave as a voltage divider across the drain region. This would potentially reduce the electric field near the drain edge of the gate. The floating field plate was not the optimal design examined in Chapter 2, however the effects would demonstrate the advantage of the incorporation of a field plate for prolonged breakdown voltages in GaN MOSFET.

The drain leakage current of a GaN MOSFET ( $L_G = 10 \ \mu m$  and  $L_{GD} = 2 \ \mu m$ ) with and without the floating field plates was compared with respect to drain voltage, shown in Figure 6.22. The gate and source contacts were grounded, while the drain voltage was swept from 0 V to 400 V. The breakdown voltage was obtained when the drain leakage current



Figure 6.22: The measured drain off-state leakage current for a 10  $\mu m$  gate GaN MOSFET with (solid green) and without (dashed blue) a floating field plate. The floating field plate increased the average critical electric field along the drain from 0.74 MV/cm to 1.4 MV/cm.

reached 80  $\mu A/mm$ . The non-field plate GaN MOSFET reached the breakdown limit at 148 V compared to the floating field plated MOSFET which suppressed breakdown to 273 V. The leakage current with respect to drain voltage produced a soft breakdown; whereas, the field plate suppressed the electric field to reduce the leakage current before breakdown occurred. The field plated MOSFET pushed the breakdown voltage by 2x and created a hard breakdown, rather than a soft breakdown. Through the addition of a floating field plate the breakdown voltage was able to suppress the average drain electric field strength to 140  $V/\mu m$ . With optimized field plates demonstrated in simulations, the GaN MOSFET will further improve the average drain field strength.

#### Gate Length Dependence

The channel length of a MOSFET has the ability to increase current density, improve switching frequency and reduce gate leakage current. However, during the off-state the breakdown voltage has been shown to have negligible effect of the drain voltage rating of the device. The majority of the drain voltage drop occurs along the drain regions rather than below the gate. The gate region is essentially only used to reduce the current flowing from the drain current. Simulations shown in Chapter 2 of the breakdown voltage of the MOSFET confirmed the majority of the high electric field strength occurred along the drain.



Figure 6.23: The gate length extrapolated maximum drain voltage of GaN MOSFET architectures with different drift regions;  $L_{GD} = 1 \ \mu m$  (squares)  $L_{GD} = 5 \ \mu m$  (circles).

The electric field under the MOS channel was well below the point of avalanche effects and catastrophic breakdown. Furthermore, initial simulations predicted that the breakdown voltage of the GaN MOSFET was not significantly effected by larger gate lengths.

The drain leakage current for different gate length MOSFETs ( $L_G = 3, 5$  and 10  $\mu m$ ) were examined for channel length variation. The off-state drain leakage current for different gate lengths and drain-gate lengths, shown in Figure 6.23, confirmed the drains drift region dictated the breakdown voltage rather than the gate length. Again, this was in agreement with the simulations obtained in Chapter 2 which examined the dependence on the gate length. The negligible gate length effect on the drain breakdown voltage has the potential to increase current density (i.e. power rating), while maintaining a large voltage rating. Overall, the maximum power rating of the GaN MOSFET has the ability to surpass current HFET technology while maintaining a higher quality off-state.

#### Gate-Source Overlap

It has been shown for low voltage silicon technology, that the source region must be present beneath the gate metal. This overlap ensured a reduced drain-induce barrier lowering effect (DIBL) [152–154] and gate-source edge electric field [155, 156]. As the source overlap was pulled out from the gate the DIBL increased, shown in Figure 6.24, which implied the



Figure 6.24: The breakdown of the device can be considered when the shift of the threshold voltage reached the low drain threshold voltage  $(V_{th}^{low})$ . This is where the large drain voltage was considered as  $V_{th} = 0$  V.

threshold voltage degraded at large drain voltages. Furthermore, it has been demonstrated that the off-state leakage current is reduced as the source is overlapped under the gate [156]. Threshold voltage variation creates a problem regarding the on-state operation; however, a large DIBL effect has the potential to significantly degrade the quality of the off-state. This would reduce the maximum drain voltage applied to the drain before the device is no longer considered off, shown in Figure 6.24. Through simulation is was determined that the breakdown occurs from higher electron energy and electric field strength at the gate-source edge. Ultimately, the device trends towards a depletion mode transistor, rather than an enhancement mode normally-off device.

The optimal location of the gate-source overlap is critical for the peak transistor characteristics. However, due to the alignment fabrication difficulties present in the facility the extreme cases of overlap and under-lap were examined. The gate-source overlap of the GaN MOSFET was examined at two locations; 0.1  $\mu m$  under the gate ( $L_{ov} = 0.1 \ \mu m$ ) and 0.5  $\mu m$  away from the gate ( $L_{ov} = -0.5 \ \mu m$ ). Pulling the AlN away from the gate ultimately degraded the maximum voltage on the drain before the device was considered on. The source was not fully depleted at the gate edge which resulted in a larger leakage current and early onset of breakdown, shown in Figure 6.25. The extrapolated breakdown voltage with respect to drain drift region of GaN MOSFETs were examined for the recessed source and overlapped source. The overlapped gate-source breakdown line was 10x larger than



Figure 6.25: The extrapolated breakdown voltage based on the substrate leakage current was examined as the AlN on the source-gate edge was removed from under the gate metal. A source-gate overlap of 100 nm ensured the breakdown voltage was close to 100 V/um. Removing the source carriers from under the gate prevented the transistor from withstanding larger drain voltages.

the recessed source. This was somewhat counter intuitive, pulling the carriers farther away on the source edge increased the leakage current and reduced the overall maximum drain voltage.

The gate-source overlap was demonstrated as one of the most imperative regions of the GaN MOSFET to ensure an off-state was achieved. The GaN MOSFET was able to produce an improved breakdown voltage with a non-optimized gate-source overlap of 100 nm. The optimal length of this region will have a significant improvement on the transistor properties. Thus, advanced simulations of the gate-source region, film stack and overlap dimensions are imperative to obtain the minimal DIBL effect to improve the overall maximum drain voltage.

### 6.4 Summary

The demonstration of an enhancement-mode GaN MOSFET was achieved with low temperature PEALD films along the gate, source and drain regions. Utilizing the low temperature deposition allowed for the GaN MOSFET to be fabricated without the use of chemical to etch films or high temperature growth techniques. This reduced the overall number of fabrication steps, as well as photo-lithography masks.

The turn-on voltage of the transistor was controlled through the workfunction of the gate metal. By using Cr as the gate metal the device turn-on voltage was determined as 1.3 V, whereas the lower workfunction metal (Al) produced a depletion-mode device. The control of the flat-band voltage of the MOS channel based on the metallic workfunction provided a simplistic design feature for the turn-on voltage of the MOSFET.

The breakdown voltage of fabricated GaN MOSFET and GaN MOS-HFET were examined with different gate-drain lengths. The non-field plated GaN MOSFET architecture was able to withstand drain voltages equivalent to the leading optimized field plated MOS-HFET in literature. With the improvement of optimized field plates, the MOSFET has the ability to reduce the gate-drain separation to achieve an equivalent drain voltage rating.

The extraction of the channel mobility through the intrinsic transconductance, suggested that the novel GaN MOSFET demonstrated has the potential to reach current HFET technology once the parasitic resistances are maintained. Furthermore, a peak channel mobility above 200  $cm^2/Vs$  was one of the leading MOS mobility for GaN transistors. This was obtained from the improved interface along the channel due to the developed PEALD deposition technique for rare earth metal oxides. The MOSFET architecture has been shown to produce promising current densities and potential maximum drain voltages. Through the incorporation of low temperature PEALD films along with conventional GaN fabrication techniques, the GaN MOSFET has the potential for high power electronics.

## Chapter 7

# **Summary and Conclusions**

To advance high power electronic devices, novel GaN transistors are required to surpass current GaN HFET technology. The wide range of applications that require a switching device with high power, high frequency and high temperature capabilities has mainly driven GaN research in HFETs, MOSHFETs and MOSFETs. Conventional gate dielectric fabrication techniques have been unable to maintain the density of interface traps, which in-turn reduce the peak capacitance density and channel mobility. With hindered device characteristics, the GaN MOSFET has been unable to compete with modern HFET technology. This thesis examined the use of low temperature PEALD films for the incorporation into the gate, source and drain regions of the device to improve the characteristics of the GaN MOSFET.

Preliminary high- $\kappa$  rare earth metal oxides on GaN were examined for potential use in a GaN MOSFET. Ultra-thin low temperature PEALD  $ZrO_2$  and  $HfO_2$  films were examined to determine the optimal material type and thickness for the proposed GaN MOSFET architecture. Through electrical extraction and characterization, Table 7.1, the  $ZrO_2$  film produced larger capacitance densities (> 3  $\mu F/cm^2$ ) and lower density of interface traps (< 5x10<sup>10</sup>  $eV^{-1}cm^{-2}$ ) compared to the  $HfO_2$  and current GaN MOS structures found in

Table 7.1: Low	Temperature	PEALD	Gate	Dielectrics	on	$\operatorname{GaN}$

Film	Growth	Film	Dielectric	$C_{ox}$	$D_{it}$	$\mu_n$
Type	Cycles	Thickness	$\epsilon_r$	$(\mu F/cm^2)$	$(eV^{-1}cm^{-2})$	$(cm^2/Vs)$
	30	3.2 nm	22.9	3.0	$40 x 10^{10}$	263
$HfO_2$	40	$5.4 \mathrm{nm}$	16.9	2.1	$50 x 10^{10}$	243
	58	$8.5 \ \mathrm{nm}$	8.1	0.8	$20 x 10^{10}$	233
	30	3.2 nm	34.8	3.8	$5 x 10^{10}$	360
$ZrO_2$	40	$5.4 \mathrm{nm}$	22.8	2.6	$3x10^{10}$	387
	58	$8.5 \ \mathrm{nm}$	12.1	1.1	$6 x 10^{10}$	370

Film	Growth	Film	$R_{AlN}$	$N_{2DEG}$	$\mu_e$
Type	Cycles	Thickness	$(\Omega/sqr)$	$(10^{13} cm^{-2})$	$(cm^2/Vs)$
	40	3.2  nm	210	2.39	1245
$N^+$ GaN	70	$5.7 \mathrm{~nm}$	175	3.04	1175
	100	$7.9 \mathrm{nm}$	550	0.91	1249
	40	3.2  nm	2178	0.28	1024
$N^{-}$ GaN	70	$5.7 \mathrm{~nm}$	592	1.14	926
	100	$7.9 \mathrm{nm}$	1307	0.47	1017

Table 7.2: Ultra-thin Low Temperature PEALD AlN Films on GaN Templates

literature. The  $ZrO_2$  films produced larger relative dielectric constants and higher peak electron mobilities compared to the  $HfO_2$  films. Based on the extracted dielectric interface characteristics, the rare-earth metal oxide films have the ability to improve GaN MOS applications for switching applications.

Low temperature plasma-enhanced atomic layer deposition AlN grown films were demonstrated on GaN for the purpose to produce high carrier densities along the interface. The AlN film thickness and GaN template background doping were strongly dependent on the resulting sheet resistance and 2DEG, shown in Table 7.2. The low temperature deposition technique was able to produce equivalent carrier densities compared to conventional higher temperature (> 800°C) techniques. A selective PEALD photoresist lift-off technique was capable to pattern the films without the use of complex etching and mask fabrication processes.

Enhancement-mode GaN MOSFETs were fabricated using low temperature PEALD AlN films for the active source and drain regions. This produced carriers along the source and drain without the difficulties associated with ion implantation. The characterized high quality 40 cycles  $ZrO_2$  film was used as the gate dielectric for the fabricated MOSFET device. The GaN MOSFETs exhibited normally-off behavior ( $V_t \approx 1.3$  V) with the use of a Cr gate metal workfunction. Switching the gate metal to Al significantly degraded the device characteristics, shown in Table 7.3, due to the depletion-mode operation. The use

Gate	$V_t$	Sub-threshold	$I_{DS}$	$g_m$	$D_{it}$
Metal	(V)	(mV/dec)	(mA/mm)	$(\mu S/mm)$	$(cm^{-2}eV^{-1})$
Al	-0.15	956	0.01	6.5	$1.1 \mathrm{x} 10^{11}$
Cr	1.3	123	0.17	45	$1.7 \mathrm{x} 10^{10}$

Table 7.3: GaN MOSFET Device Characteristics

of unintentionally doped GaN templates did not prevent substrate leakage current during the off state of the device for both Cr and Al gate metal. To correct a lower doped GaN template, which produced a true off-state, was examined for drain breakdown voltage. The parasitic contact resistance was not optimized for the test devices which limited overall peak current density. However, the intrinsic transconductance of the fabricated MOSFET was able to surpass many GaN MOSFET devices with smaller channel lengths. Based on the intrinsic transconductance, the GaN MOSFET has the potential to compete with current HFET technology as the channel was reduced.

Further investigation of the breakdown voltage on semi-insulating GaN-on-sapphire templates confirmed simulations which predicted the improved off-state performance the MOS-FET architecture has over HFET designs. The most important region of the MOSFET which dictated the breakdown voltage and off-state characteristics of the device was the gate-source edge and gate carrier density overlap. Without the overlap of the gate metal and source carriers, the device was a severally degraded from an increased DIBL effect. With an appropriated gate-source overlap, the average drain electric field strength was extracted from the extrapolated breakdown voltage with respect to gate-drain separation. The non-field plated MOSFET design was capable of withstanding equivalent field strengths ( $\sim$ 90 V/ $\mu m$ ) to the optimized gate and source field plated MOS-HFET designed. With the incorporation of field plates, the simulated GaN MOSFET has shown the ability to improve the maximum drain voltage along the drift region compared to the HFET architecture. Floating field plates were fabricated along the drain region to suppress peak electric fields in a MOSFET structure, shown in Table 7.4. The extracted average field strength along the drain without a field plate and a floating field plate were 90 V/ $\mu m$  and 147 V/ $\mu m$ , respectively. Through the inclusion of floating field plates, the GaN MOSFET has demonstrated significant off-state characteristics over the GaN HFET for potential power applications. With the addition of optimized field plates the GaN MOSFET design has a significant advantage over HFET architecture for high voltage applications.

Simulation							
$L_{GD}$	$1 \ \mu m$	$2 \ \mu m$	$5~\mu m$	$10 \ \mu m$	Avg Field		
No Field Plate	41 V	80 V	220 V	420  V	$42 \text{ V}/\mu m$		
Source FP	-	$350 \mathrm{V}$	1200 V	$2150~\mathrm{V}$	$210 \text{ V}/\mu m$		
Gate/Source FP	-	-	$1800 \mathrm{V}$	$4990~\mathrm{V}$	$429~\mathrm{V}/\mu m$		
Experimental							
$L_{GD}$	$1 \ \mu m$	$2~\mu m$	$5 \ \mu m$	$10 \ \mu m$	Avg Field		
No Field Plate	$65 \mathrm{V}$	148 V	446 V	787 V	$90 \text{ V}/\mu m$		
Floating FP	-	$273 \mathrm{~V}$	$681 \mathrm{V}$	$1495~\mathrm{V}$	147 V/ $\mu m$		

Table 7.4: GaN MOSFET Breakdown Voltage Characteristics

Low temperature PEALD films demonstrated the simplicity of ALD lift-off fabrication and the ability to attain or surpass high temperature processing techniques. The GaN MOSFET demonstrated a normally-off ( $V_{th} > 0$  V), enhancement-mode transistor with the ability of producing a channel mobility of 230  $cm^2/Vs$ . The potential of low temperature processing in-conjunction with high temperature processing has the potential to further improve the GaN MOSFET device for use in high power applications.

### 7.1 Directions for Future Work

The enhancement-mode GaN MOSFET with the incorporation of a rare earth metal oxide was used to demonstrate the potential the MOSFET has to surpass current GaN technology. To produce a fully functioning GaN MOSFET with the desired breakdown voltage and current density, the investigation of leakage current and parasitic resistances are imperative. The use of semi-insulating GaN templates will reduce the leakage current, which in-turn will improve the maximum operating drain voltage. Future work will address the parasitic resistance to reduce the contact resistance through conventional HFET fabrication techniques. Diffusion barriers along interfaces should also be considered in order to prevent degradation from diffusion during various annealing steps during fabrication.

The incorporation of conventional techniques used in GaN HFETs and low temperature PEALD films discussed in this dissertation should be examined for the potential to improve the GaN MOSFET characteristics. The use of this novel technology has demonstrated the potential for GaN MOS applications and will further advance the GaN MOSFET design for power electronics.

## References

- [1] R. Kirschman, *High Temperature Electronics*.
- [2] D. A. R. P. Agency, Nitride Electronic Next-Generation Technology (NEXT).
- [3] Nitronex, "GaN Applications," 2014.
- [4] Y. Development, "Compound Semiconductor: Power GaN 2012," *I-Micronews*, 2012.
- [5] S. Strite, "GaN, AlN, and InN: A review," Journal of Vacuum Science & Technology B: Microelectronics and Nanometer Structures, vol. 10, p. 1237, July 1992.
- [6] V. Siklitsky, "New semiconductor material: Characteristics and properties," *Physico-Technical Institue*.
- [7] A. M. Ozbek, S. Member, and B. J. Baliga, "Finite-Zone Argon Implant Edge Termination for High-Voltage GaN Schottky Rectifiers," vol. 32, no. 10, pp. 1361–1363, 2011.
- [8] S. Adachi, Handbook on Physical Properties of Semiconductors.
- [9] G. Harris, Properties of Silicon Carbide.
- [10] R. Hull, Properties of Crystalline Silicon.
- [11] B. J. Baliga, "Power Semiconductor Device Figure of Merit for High-Frequency Applications," vol. 10, no. 10, pp. 455–457, 1989.
- [12] E. Johnson, "Physical limitations on frequency and power parameters of transistors," RCA Review, pp. 163–177, 1965.
- [13] B. J. Baliga, "Semiconductors for high-voltage, vertical channel field effect transistors," vol. 12301, no. September 1981, pp. 1759–1764, 1982.
- [14] S. Sze and M. Lee, Semiconductor Devices: Physics and Technology.
- [15] C. Hu, Modern Semiconductor Devices for Integrated Circuits.
- [16] D. Schroder, Semiconductor Material and Device Characterization.
- [17] S. Pearton, F. Ren, a.P. Zhang, and K. Lee, "Fabrication and performance of GaN electronic devices," *Materials Science and Engineering: R: Reports*, vol. 30, pp. 55– 212, Dec. 2000.
- [18] D. T, P. G, M. H, K. T, C. WJ, S. A, and N. N, "Enhancement-mode AlGaN/GaN HEMTs on Silicon Substrate," *Physica Status Solidi* (a), vol. 162, p. 199, 1997.
- [19] F. L. Via, Silicon Carbide Epitaxy.
- [20] M. Asif Khan, a. Bhattarai, J. N. Kuznia, and D. T. Olson, "High electron mobility transistor based on a GaNAlxGa1xN heterojunction," *Applied Physics Letters*, vol. 63, no. 9, pp. 1214–1215, 1993.

- [21] Y. Takada, W. Saito, M. Kuraguchi, I. Omura, and K. Tsuda, "High voltage GaNbased power HEMTs with field plate technique: Breakdown voltage and switching characteristics," *Physica Status Solidi (C)*, vol. 2350, pp. 2347–2350, Dec. 2003.
- [22] N. Ikeda, K. Kato, J. Li, K. Hataya, and S. Yoshida, "Normally-off operation GaN HFET using a thin AlGaN layer for low loss switching devices," *MRS Proceedings*, vol. 831, pp. 1–6, 2004.
- [23] Y. Uemoto, D. Shibata, M. Yanagihara, H. Ishida, H. Matsuo, N. Batta, M. Li, and T. Tanaka, "8300V Blocking," pp. 861–864, 2007.
- [24] Y. Uemoto, M. Hikita, H. Ueno, H. Matsuo, H. Ishida, M. Yanagihara, T. Ueda, T. T., and D. Ueda, "Gate Injection Transistor (GIT) A Normally-Off Conductivity Modulation," vol. 54, no. 12, pp. 3393–3399, 2007.
- [25] K. Boutros, S. Burnham, D. Wong, K. Shinohara, B. Hughes, D. Zehnder, and C. McGuire, "Normally-off 5A/1100V GaN-on-silicon device for high voltage applications," *Electron Devices Meeting (IEDM)*, 2009 IEEE International, vol. 29, no. 8, pp. 161–163, 2009.
- [26] S. D. Burnham, K. Boutros, P. Hashimoto, C. Butler, D. W. Wong, M. Hu, and M. Micovic, "Gate-recessed normally-off GaN-on- Si HEMT using a new O2-BCl3 digital etching technique," *Physica Status Solidi (C)*, vol. 7, pp. 2010–2012, June 2010.
- [27] M. Kim, Y.-h. Choi, J. Lim, Y.-s. Kim, O. Seok, and M.-k. Han, "High Breakdown Voltage AlGaN / GaN HEMTs Employing Recessed Gate Edge Structure," 2010.
- [28] F. Medjdoub, M. Zegaoui, D. Ducatteau, N. Rolland, and P. a. Rolland, "High-Performance Low-Leakage-Current AlN/GaN HEMTs Grown on Silicon Substrate," *IEEE Electron Device Letters*, vol. 32, pp. 874–876, July 2011.
- [29] H. Kambayashi, Y. Satoh, S. Ootomo, T. Kokawa, T. Nomura, S. Kato, and T.s. P. Chow, "Over 100A operation normally-off AlGaN/GaN hybrid MOS-HFET on Si substrate with high-breakdown voltage," *Solid-State Electronics*, vol. 54, pp. 660– 664, June 2010.
- [30] L. Yuan, H. Chen, and K. Chen, "Normally Off AlGaN/GaN Metal 2DEG Tunnel-Junction Field-Effect Transistors," *IEEE Electron Device Letters*, vol. 32, no. 3, pp. 303–305, 2011.
- [31] K. Cico, K. Husekova, M. Tapajna, D. Gregusova, R. Stoklas, J. Kuzmik, J.-F. Carlin, N. Grandjean, D. Pogany, and K. Frohlich, "Electrical properties of InAlN/GaN high electron mobility transistor with Al[sub 2]O[sub 3], ZrO[sub 2], and GdScO[sub 3] gate dielectrics," Journal of Vacuum Science & Technology B: Microelectronics and Nanometer Structures, vol. 29, no. 1, p. 01A808, 2011.
- [32] D. J. Meyer, D. S. Katzer, D. a. Deen, D. F. Storm, S. C. Binari, and T. Gougousi, "HfO2-insulated gate N-polar GaN HEMTs with high breakdown voltage," *Physica Status Solidi* (a), vol. 208, pp. 1630–1633, July 2011.
- [33] U. Singisetti, M. Wong, J. Speck, and U. Mishra, "Enhancement-Mode N-Polar GaN MOS-HFET with 5-nm GaN Channel, 510-mS/mm, and 0.66 Ohm mm," *IEEE Elec*tron Device Letters, vol. 33, no. 1, pp. 26–28, 2012.
- [34] C. Liu, S. Liu, S. Huang, K. J. Chen, and S. Member, "Plasma-Enhanced Atomic Layer Deposition of AlN Epitaxial Thin Film for AlN / GaN Heterostructure TFTs," vol. 34, no. 9, pp. 1106–1108, 2013.
- [35] U. Singisetti, M. H. Wong, S. Dasgupta, J. S. Speck, and U. K. Mishra, "Enhancement-Mode N-Polar GaN MetalInsulatorSemiconductor Field Effect Transistors with Current Gain Cutoff Frequency of 120 GHz," *Applied Physics Express*, vol. 4, p. 024103, Jan. 2011.

- [36] C. Tsai, T. Wu, and A. Chin, "High-Performance GaN MOSFET with High-k LaAlO3/SiO2 Gate Dielectric," *IEEE Electron Device Letters*, vol. 33, no. 1, pp. 35– 37, 2012.
- [37] W. Huang, T. Khan, and T. Chow, "Enhancement-Mode n-Channel GaN MOSFETs on p and n-GaN/Sapphire Substrates," *IEEE Electron Device Letters*, vol. 27, no. 10, pp. 796–798, 2006.
- [38] W. Huang, T. Chow, Y. Niiyama, T. Nomura, and S. Yoshida, "730V, 34mOhmcm2 Lateral Epilayer RESURF GaN MOSFET," *IEEE International Electron Device Meeting*, 2009.
- [39] Y. C. Chang, W. H. Chang, H. C. Chiu, L. T. Tung, C. H. Lee, K. H. Shiu, M. Hong, J. Kwo, J. M. Hong, and C. C. Tsai, "Inversion-channel GaN metal-oxidesemiconductor field-effect transistor with atomic-layer-deposited Al[sub 2]O[sub 3] as gate dielectric," *Applied Physics Letters*, vol. 93, no. 5, p. 053504, 2008.
- [40] B. Lu, E. Matioli, and T. Palacios, "Tri-Gate normally-Off GaN Power MISFET," IEEE Electron Device Letters, vol. 33, no. 3, pp. 360–362, 2012.
- [41] K. Im, J. Ha, J. Lee, S. Kim, K. Kim, D. Kim, H. Kang, E. Kwak, S.Lee, and J. Lee, "Normally-off Operation of Recess-Gated GaN MOSFET on Silcon Substrate Using AlGaN/GaN Source/Drain with High 2DEG Density," *International Semiconductor Device Research Symposium*, 2009.
- [42] F. Medjdoub, M. Zegaoui, N. Rolland, and P. a. Rolland, "Demonstration of low leakage current and high polarization in ultrathin AlN/GaN high electron mobility transistors grown on silicon substrate," *Applied Physics Letters*, vol. 98, no. 22, p. 223502, 2011.
- [43] G. Simin, a. Koudymov, H. Fatima, M. Asif Khan, X. Hu, a. Tarakji, R. Gaska, and M. Shur, "SiO/sub 2//AlGaN/InGaN/GaN MOSDHFETs," *IEEE Electron Device Letters*, vol. 23, pp. 458–460, Aug. 2002.
- [44] G. Baksht, "Performance Stability of AlGaN/GaN HFETs: Effect of Plasma Processing," *Physica Status Solidi*, vol. 5, no. 6, pp. 2033–2036, 2008.
- [45] W. Jai, "Enhancement-mode AlGaN/GaN HEMTs on Silicon Substrate," Physica Status Solidi, vol. 3, no. 6, pp. 2368–2372, 2006.
- [46] K. M. Bothe and D. W. Barlage, "Underlying design advantages for GaN MOSFETs compared with GaN HFETs for power applications," *Journal of Computational Elec*tronics, Sept. 2013.
- [47] Synopsys, "Sentaurus device datasheet," 2007.
- [48] C. Sarkar, Technology Computer Aided Design: Simulation for VLSI MOSFET.
- [49] M. Levinshtein, J. Kostamovaara, and S. Vainshtein, Breakdown Phenomena in Semiconductors and Semiconductor Devices.
- [50] M. M. A. Dmitriev and I. Yassievich, "Impact Ionization in Semiconductors in High Electric Fields," *Physica Status Solidi* (b), vol. 140, no. 1, pp. 9–37, 1987.
- [51] A. Chynoweth, "Ionization rates for electrons and hole in silicon," *Physical Review*, vol. 109, no. 5, pp. 1537–1540, 1958.
- [52] Y. Okuto and C. Crowell, "Threshold energy effect on avalanche breakdown voltage in semiconductor junctions," *Solid-State Electronics*, vol. 18, no. 2, pp. 161–168, 1975.
- [53] T. Lackner, "Avalanche multiplication in semiconductors: A modification of chynoweth's law," *Solid-State Electronics*, vol. 34, no. 1, pp. 33–42, 1991.

- [54] R. Overstraeten and H. de Man, "Measurement of ionization rates in diffused silicon p-n junctions," *Solid-State Electronics*, vol. 13, no. 1, pp. 583–608, 1970.
- [55] R. Chau, B. Doyle, J. Kavalieros, D. Barlage, A. Murthy, and M. Doczy, "Advanced Depleted-Substrate Transistors : Single-gate , Double-gate and Tri-gate (Invited Paper)," pp. 1–2, 2001.
- [56] A. Grove, *Physics and Technology of Semiconductor Devices*.
- [57] E. Brandes and G. Brook, *Smithells Metal Reference Book*.
- [58] R. J. Trew, U. K. Mishra, and S. Member, "Gate Breakdown in MESFET s and HEMT s," vol. 12, no. 10, pp. 524–526, 1991.
- [59] S. Karmalkar and U. Mishra, "Enhancement of breakdown voltage in AlGaN/GaN high electron mobility transistors using a field plate," *IEEE Transactions on Electron Devices*, vol. 48, no. 8, pp. 1515–1521, 2001.
- [60] Y. Dora, a. Chakraborty, L. Mccarthy, S. Keller, S. Denbaars, and U. Mishra, "High Breakdown Voltage Achieved on AlGaN/GaN HEMTs With Integrated Slant Field Plates," *IEEE Electron Device Letters*, vol. 27, pp. 713–715, Sept. 2006.
- [61] G. Fernholz and H. Beneking, "Ion Implanted Si MESFET's with High Cutoff Frequency," *IEEE Transactions on Electron Devices*, vol. 30, no. 7, pp. 837–840, 1983.
- [62] H. Xing, Y. Dora, A. Chini, S. Heikman, S. Keller, U. K. Mishra, A. H.-v. Al, and G. N. Gan, "High Breakdown Voltage AlGaN GaN HEMTs Achieved by Multiple Field Plates," vol. 25, no. 4, pp. 161–163, 2004.
- [63] K. M. Bothe, P. A. V. Hauff, A. Afshar, A. Foroughi-abari, K. C. Cadien, and D. W. Barlage, "Capacitance Modeling and Characterization of Planar MOSCAP Devices for Wideband-Gap Semiconductors With High-  $\kappa$  Dielectrics," vol. 59, no. 10, pp. 2662–2666, 2012.
- [64] K. M. Bothe, P. A. V. Hauff, A. Afshar, A. Foroughi-abari, K. C. Cadien, and D. W. Barlage, "Electrical Comparison of HfO 2 and ZrO 2 Gate Dielectrics on GaN," vol. 60, no. 12, pp. 4119–4124, 2013.
- [65] A. A. A. F.-A. D. B. P. von hauff, K. Bothe and K. Cadien, "High Mobility (210cm2/Vs). High Capacitance (7.2uF/cm2) ZrO2 on GaN Metal Oxide Semiconductor Capacitor via ALD," CS Mantech Conference, 2012.
- [66] H. C. Casey, G. G. Fountain, R. G. Alley, B. P. Keller, and S. P. DenBaars, "Low interface trap density for remote plasma deposited SiO2 on n-type GaN," *Applied Physics Letters*, vol. 68, no. 13, p. 1850, 1996.
- [67] S. Campbell, D. Gilmer, and W. Gladfelter, "MOSFET transistors fabricated with high permitivity TiO/sub 2/ dielectrics," *IEEE Transactions on Electron Devices*, vol. 44, no. 1, pp. 104–109, 1997.
- [68] L. W. Tu, W. C. Kuo, K. H. Lee, P. H. Tsao, C. M. Lai, a. K. Chu, and J. K. Sheu, "High-dielectric-constant Ta[sub 2]O[sub 5]/n-GaN metal-oxide-semiconductor structure," *Applied Physics Letters*, vol. 77, no. 23, p. 3788, 2000.
- [69] J. W. Johnson, B. Luo, F. Ren, B. P. Gila, W. Krishnamoorthy, C. R. Abernathy, S. J. Pearton, J. I. Chyi, T. E. Nee, C. M. Lee, and C. C. Chuo, "Gd[sub 2]O[sub 3]/GaN metal-oxide-semiconductor field-effect transistor," *Applied Physics Letters*, vol. 77, no. 20, p. 3230, 2000.
- [70] B. Gila, J. Johnson, R. Mehandru, B. Luo, a.H. Onstine, V. Krishnamoorthy, S. Bates, C. Abernathy, F. Ren, and S. Pearton, "Gadolinium Oxide and Scandium Oxide: Gate Dielectrics for GaN MOSFETs," *Physica Status Solidi (a)*, vol. 188, pp. 239–242, Nov. 2001.

- [71] B. Gaffey, L. J. Guido, S. Member, and X. W. Wang, "High-Quality Oxide / Nitride / Oxide Gate Insulator for GaN MIS Structures," vol. 48, no. 3, pp. 458–464, 2001.
- [72] R. Mehandru, B. P. Gila, J. Kim, J. W. Johnson, K. P. Lee, B. Luo, a. H. Onstine, C. R. Abernathy, S. J. Pearton, and F. Ren, "Electrical Characterization of GaN Metal Oxide Semiconductor Diode Using Sc[sub 2]O[sub 3] as the Gate Oxide," *Electrochemical and Solid-State Letters*, vol. 5, no. 7, p. G51, 2002.
- [73] J. Kim, R. Mehandru, B. Luo, F. Ren, B. P. Gila, a. H. Onstine, C. R. Abernathy, S. J. Pearton, and Y. Irokawa, "Characteristics of MgO/GaN gate-controlled metaloxide semiconductor diodes," *Applied Physics Letters*, vol. 80, no. 24, p. 4555, 2002.
- [74] Y. Nakano, T. Kachi, and T. Jimbo, "Characteristics of SiO[sub 2]/n-GaN interfaces with  $\beta$ -Ga[sub 2]O[sub 3] interlayers," *Applied Physics Letters*, vol. 83, no. 21, p. 4336, 2003.
- [75] H. Cho, K. Lee, B. Gila, C. Abernathy, S. Pearton, and F. Ren, "Influence of gate oxide thickness on Sc2O3/GaN MOSFETs," *Solid-State Electronics*, vol. 47, pp. 1757–1761, Oct. 2003.
- [76] P. D. Ye, B. Yang, K. K. Ng, J. Bude, G. D. Wilk, S. Halder, and J. C. M. Hwang, "GaN metal-oxide-semiconductor high-electron-mobility-transistor with atomic layer deposited Al[sub 2]O[sub 3] as gate dielectric," *Applied Physics Letters*, vol. 86, no. 6, p. 063501, 2005.
- [77] Y. Dora, S. Han, D. Klenov, P. J. Hansen, K.-s. No, U. K. Mishra, S. Stemmer, and J. S. Speck, "ZrO[sub 2] gate dielectrics produced by ultraviolet ozone oxidation for GaN and AlGaNGaN transistors," *Journal of Vacuum Science & Technology B: Microelectronics and Nanometer Structures*, vol. 24, no. 2, p. 575, 2006.
- [78] S. Abermann, G. Pozzovivo, J. Kuzmik, G. Strasser, D. Pogany, J.-F. Carlin, N. Grandjean, and E. Bertagnolli, "MOCVD of HfO 2 and ZrO 2 high- k gate dielectrics for InAlN/AlN/GaN MOS-HEMTs," *Semiconductor Science and Technology*, vol. 22, pp. 1272–1275, Dec. 2007.
- [79] Y. C. Chang, H. C. Chiu, Y. J. Lee, M. L. Huang, K. Y. Lee, M. Hong, Y. N. Chiu, J. Kwo, and Y. H. Wang, "Structural and electrical characteristics of atomic layer deposited high κ HfO[sub 2] on GaN," *Applied Physics Letters*, vol. 90, no. 23, p. 232904, 2007.
- [80] D. Deen, D. Storm, D. Meyer, D. S. Katzer, R. Bass, S. Binari, and T. Gougousi, "AlN/GaN HEMTs with high-κ ALD HfO2 or Ta2O5 gate insulation," *Physica Status Solidi (C)*, vol. 8, pp. 2420–2423, July 2011.
- [81] P. V. Hauff, K. Bothe, A. Afshar, D. Barlage, and K. Cadien, "High Mobility (," 2012.
- [82] D. Deen, D. Storm, D. Katzer, D. Meyer, and S. Binari, "Dependence of ohmic contact resistance on barrier thickness of AlN/GaN HEMT structures," *Solid-State Electronics*, vol. 54, pp. 613–615, June 2010.
- [83] S. Taking, A. Z. Khokhar, D. Macfarlane, S. Sharabi, A. M. Dabiran, and E. Wasige, "New Process for Low Sheet and Ohmic Contact Resistance of AlN / GaN MOS-HEMTs," no. September, pp. 306–309, 2010.
- [84] K. Chabak, A. Crespo, D. Tomich, D. Langley, V. Miller, M. Trejo, J. K. Gillespie, G. D. Via, and W.-p. A. F. Base, "Processing Methods for Low Ohmic Contact Resistance in AlN / GaN MOSHEMTs," 2009.
- [85] H. Xing, T. Zimmermann, D. Deen, K. Wang, C. Yu, T. Kosel, P. Fay, and D. Jena, "Ultrathin all-binary aln/gan based high performance rf hemt technology," CS Mantech, 2008.

- [86] I. P. Smorchkova, S. Keller, S. Heikman, C. R. Elsass, B. Heying, P. Fini, J. S. Speck, and U. K. Mishra, "Two-dimensional electron-gas AlN/GaN heterostructures with extremely thin AlN barriers," *Applied Physics Letters*, vol. 77, no. 24, p. 3998, 2000.
- [87] I. P. Smorchkova, L. Chen, T. Mates, L. Shen, S. Heikman, B. Moran, S. Keller, S. P. DenBaars, J. S. Speck, and U. K. Mishra, "AlN/GaN and (Al,Ga)N/AlN/GaN two-dimensional electron gas structures grown by plasma-assisted molecular-beam epitaxy," *Journal of Applied Physics*, vol. 90, no. 10, p. 5196, 2001.
- [88] A. Pérez-Tomás, a. Fontserè, M. Placidi, N. Baron, S. Chenot, Y. Cordier, J. Moreno, P. M. Gammon, and M. R. Jennings, "Ohmic Contact Resistance to GaN Devices Dependence with on Temperature for GaN Devices T," *Materials Science Forum*, vol. 679-680, pp. 816–819, Mar. 2011.
- [89] M. Gonschorek, J. Carlin, E. Feltin, M. Py, N. Grandjean, V. Darakchieva, B. Monemar, M. Lorenz, and G. Ramm, "Two-dimensional electron gas density in alinn/aln/gan heterostructures (0.03jxj0.23)," *Journal of Applied Physics*, vol. 103, no. 9, p. 093714, 2008.
- [90] K. Bothe, A. Ma, K. Voon, A. Afshar, P. Motamedi, K. Cadien, and D. Barlage, "Selective deposition of low temperature aln ohmic contacts for gan devices," CS Mantech Conference, pp. 369–371, 2014.
- [91] K. Voon, K. Bothe, P. Motamedi, K. Cadien, and D. Barlage, "Engineered tunneling contacts with low-temperature atomic layer deposition of aln on gan," CS Mantech Conference, pp. 175–178, 2014.
- [92] M. E. Lin, Z. Ma, F. Y. Huang, Z. F. Fan, L. H. Allen, H. Morkog, and Z. Gan, "Low resistance ohmic contacts on wide band-gap GaN s," vol. 64, no. May 2001, pp. 1003–1005, 1994.
- [93] J. Yun, K. Choi, K. Mathur, V. Kuryatkov, B. Borisov, G. Kipshidze, S. Nikishin, and H. Temkin, "Low-resistance Ohmic contacts to digital alloys of n-AlGaN/AlN," *IEEE Electron Device Letters*, vol. 27, pp. 22–24, Jan. 2006.
- [94] D. Selvanathan, L. Zhou, V. Kumar, and I. Adesida, "Low resistance ti/al/mo/au ohmic contacts for algan/gan heterostructure field effect transistors," *Physics Status Soldi* (a), vol. 194, no. 2, pp. 583–586, 2002.
- [95] A. Fontsere, A. Perez-Tomas, M. Placidi, P. Fernandez-Martinez, N. Baron, S. Chenot, Y. Cordier, J. Moreno, M. Jennings, P. Gammon, and D. Walker, "Gan ohmic contact resistance vs temperature," *Spanish Conference on Electron Devices*, vol. 8, 2011.
- [96] M. Veety, "Four terminal gallium nitride mosfets," PhD. Dissertation, 2011.
- [97] J. Shiu, J. Huang, C. Desmaris, V. Chang, C. Lu, K. Kumakura, T. Makimoto, H. Zirath, N. Rorsman, and E. Chang, "Oxygen ion implatation isolation planar process for algan/gan hemts," *Electron Device Letters*, vol. 28, no. 6, pp. 476–478, 2007.
- [98] Y. Yeo, T. King, and C. Hu, "Mosfet gate leakage modeling and selection guide for alternative gate dielectrics based on leakage considerations," *Electron Devices*, *IEEE Transactions on*, vol. 50, pp. 1027–1035, April 2003.
- [99] W. Lee and C. Hu, "Modeling cmos tunneling currents through ultrathin gate oxide due to conduction- and valence-band electron and hole tunneling," *Electron Devices*, *IEEE Transactions on*, vol. 48, pp. 1366–1373, Jul 2001.
- [100] F. Sacconi, J. M. Jancu, M. Povolotskyi, and A. D. Carlo, "Full-Band Tunneling in High- κ Oxide MOS Structures," vol. 54, no. 12, pp. 3168–3176, 2007.
- [101] D. Griffiths, Introduction to Quantum Mechanics.

- [102] G. D. Wilk, R. M. Wallace, and J. M. Anthony, "Hafnium and zirconium silicates for advanced gate dielectrics," *Journal of Applied Physics*, vol. 87, no. 1, p. 484, 2000.
- [103] S. Monaghan, P. Hurley, K. Cherkaoui, M. Negara, and A. Schenk, "Determination of electron effective mass and electron affinity in hfo2 using {MOS} and {MOSFET} structures," *Solid-State Electronics*, vol. 53, no. 4, pp. 438 – 444, 2009. Special Issue with papers selected from the Ultimate Integration on Silicon Conference, {ULIS} 2008.
- [104] Y. Fan, R. Nieh, J. C. Lee, G. Lucovsky, G. Brown, L. Register, and S. K. Banerjee, "Voltage- and temperature-dependent gate capacitance and current model: application to zro2 n-channel mos capacitor," *Electron Devices, IEEE Transactions on*, vol. 49, pp. 1969–1978, Nov 2002.
- [105] A. Foroughi-Abari, "Atomic layer deposition of metal oxide thin films on metallic substrates," University of Alberta PhD Dissertation, 2012.
- [106] W. R. Harrell, T. H. Cordella, and K. F. Poole, "Implications of non-linear poolefrenkel plots on high-," vol. 1, no. 5, pp. 705–716, 2006.
- [107] K. Y. Cheong, J. H. Moon, H. J. Kim, W. Bahng, and N.-K. Kim, "Current conduction mechanisms in atomic-layer-deposited HfO[sub 2]/nitridedSiO[sub 2] stacked gate on 4H silicon carbide," *Journal of Applied Physics*, vol. 103, no. 8, p. 084113, 2008.
- [108] J. Frenkel, "On pre-breakdown phenomena in insulators and electronic semiconductors," Phys. Rev., vol. 54, pp. 647–648, Oct 1938.
- [109] T. Yamaguchi, H. Satake, and N. Fukushima, "Band diagram and carrier conduction mechanisms in zro2 mis structures," *Electron Devices*, *IEEE Transactions on*, vol. 51, pp. 774–779, May 2004.
- [110] J. Robertson, "Band offsets of wide-band-gap oxides and implications for future electronic devices," Journal of Vacuum Science & Technology B: Microelectronics and Nanometer Structures, vol. 18, no. 3, p. 1785, 2000.
- [111] B. Jonsson and S. Eng, "Solving the schrodinger equation in arbitrary quantum-well potential profiles using the transfer matrix method," *Quantum Electronics, IEEE Journal of*, vol. 26, pp. 2025–2035, Nov 1990.
- [112] S.-H. Lo, D. a. Buchanan, and Y. Taur, "Modeling and characterization of quantization, polysilicon depletion, and direct tunneling effects in MOSFETs with ultrathin oxides," *IBM Journal of Research and Development*, vol. 43, pp. 327–337, May 1999.
- [113] X. Wang, C. Yin, M. Sang, M. Dai, and Z. Cao, "Investigation on tunneling in optoelectronic devices with consideration of subwaves," *Science China: Physics, Mechanics and Astronomy*, vol. 54, 2011.
- [114] G. Bastard and J. A. Brum, "Electronic states in semiconductor heterostructures," *Quantum Electronics, IEEE Journal of*, vol. 22, pp. 1625–1644, Sep 1986.
- [115] L. Register, E. Rosenbaum, and K. Yang, "Analytic model for direct tunneling current in polycrystalline silicon-gate metal-oxide-semiconductor devices," *Applied Physics Letters*, vol. 74, no. 3, pp. 457–459, 1999.
- [116] D. Barlage, J. O'Keeffe, J. Kavalieros, M. Nguyen, and R. Chau, "Inversion mos capacitance extraction for high-leakage dielectrics using a transmission line equivalent circuit," *Electron Device Letters, IEEE*, vol. 21, pp. 454–456, Sept 2000.
- [117] D. Pozar, *Microwave Engineering*.

- [118] T. Hashizume, S. Ootomo, T. Inagaki, and H. Hasegawa, "Surface passivation of gan and gan/algan heterostructures by dielectric films and its application to insulated-gate heterostructure transistors," *Journal of Vacuum Science B*, vol. 21, no. 4, pp. 1828– 1838, 2003.
- [119] E. Miller, E. Yu, C. Poblenz, C. Elsass, and S. S., "Direct measurement of the polarization charge in algan/gan heterostructures using capacitance-voltage carrier profiling," *Applied Physics Letters*, vol. 80, no. 19, pp. 3551–3553, 2002.
- [120] Y. Tang, P. Saunier, R. Wang, A. Ketterson, X. Gao, G. Guo, S. Snider, D. Jena, H. Xing, and P. Fay, "High performance monolithically-integrated e/d mode inaln/aln/gan hemts for mixed-signal applications," *Internation Electron Device Meetings*, pp. 30.4.1–30.4.4, 2010.
- [121] P. Prieto, F. Yubero, E. Elizadle, and J. Sanz, "Dielectric properties of zr, zrn, zr3n4 and zro2 determined by qantitative analysis of electron energy loss spectra," *Journal* of Vacuum Science and Technology, vol. 14, no. 6, pp. 3181–3188, 1996.
- [122] I. Milosev, H. Strehblow, M. Gaberscek, and B. Navinsek, "Electrochemical oxidation of zrn hard (pvd) coatings studied by xps," *Surface and Interface Analysis*, vol. 24, pp. 448–458, 1995.
- [123] P. Prieto, L. Galan, and J. Sanz, "Interaction of oxygen with zrn at room temperature: An xps study," Surface and Interface Analysis, vol. 21, pp. 395–399, 1994.
- [124] P. Motamedi and K. Cadien, "Low-temperature deposition of crystalline aln thin films using nitrogen plasma enhance atomic layer deposition," *ALD 2013 Conference*, 2013.
- [125] B. Johansson, H. Hentzell, J. Harper, and J. Cuomo, "Higher nitrides of hafnium, zirconium, and titanium synthesized by dual ion beam deposition," *Journal of Materials Research*, vol. 1, no. 3, pp. 442–451, 1986.
- [126] W. Ching, Y. Xu, and L. Ouyang, "Electronic and dielectric properties of insulating zr3n4," *Physical Review B*, vol. 66, no. 23, p. 235106, 2002.
- [127] M. Fay, G. Moldovan, P. Brown, I. Harrison, J. Birbeck, M. Hughes, M. Uren, and T. Martin, "Structural and electrical characterization of autialti/algan/gan ohmic contacts," *Journal of Applied Physics*, vol. 92, pp. 94–100, 2002.
- [128] A. Agrawal, J. Lin, B. Zheng, S. Sharma, S. Chopra, K. Wang, A. Gelator, S. Mohney, and S. Datta, "Barrier height reduction to 0.15ev and contact resistivity reductino to 9.1x10-9 ohm-cm2 using ultrathin tio2 interlayer between metal and silicon," Proceedings of the Digest of Technical Papers-Symposium on VLSI Technology, vol. 6, pp. 9–10, 2012.
- [129] K. Voon, K. Bothe, P. Motamedi, K. Cadien, and D. Barlage, "Polarization charfe properties of low-temperature atomic layer deposition of aln on gan," *Journal of Physics D: Applied Physics*, vol. 47, p. 345104, 2014.
- [130] K. Voon, K. Bothe, P. Motamedi, K. Cadien, and D. Barlage, "Engineered tunneling contact with low-temperature atomic layer deposition of aln on gan," CS Mantech, pp. 175–178, 2014.
- [131] J. Kwak, S. Mohney, J. Lin, and R. Kern, "Low resistance al/ti/n-gan ohmic contacts with improved surface morphology and thermal stability," *Semiconductor Science Technology*, vol. 15, pp. 756–760, 2000.
- [132] A. Adikimenakis, K. Aretouli, E. Iliopoulos, A. Kostopoulos, K. Tsagaraski, G. Konstantinidis, and A. Georgakilas, "High electron mobility transistors based on the aln/gan heterojunction," *Microelectronic Engineering*, vol. 86, pp. 1071–1073, 2009.

- [133] O. Ambacher, B. Foutz, J. Smart, J. Shealy, N. Weimann, K. Chu, M. Murphy, A. Sierakowski, W. Schaff, L. Eastman, R. Dimitrov, A. Mitchell, and M. Stutzmann, "Two dimensional electron gases induced by spontaneous and piezoelectric polarization in undoped and doped algan/gan heterostructures," *Journal of Applied Physics*, vol. 87, pp. 334–344, 2000.
- [134] K. Yang, Y. King, and C. Hu, "Quantum effect in oxide thickness determination from capacitance measurement," Symposium on VLSI Technology, pp. 77–78, 1999.
- [135] J. Wuerfl, E. Bahat-Treidel, F. Brunner, E. Cho, O. Hilt, P. Ivo, A. Knauer, P. Kurpas, R. Lossy, S. S. a. M. Schulz, M. and, and R. Zhytnytska, "Reliability issues of gan based high voltage power devices," *Microelectronics Reliability*, vol. 51, pp. 1710–1716, 2011.
- [136] G. Meneghesso, G. Verzellesi, F. Danesin, F. Rampazzo, F. Zanon, A. Tazzoli, M. Meneghini, and E. Zanoni, "Reliability of gan high-electron-mobility transistors: State of the art and perspectives," *IEEE Transactions on Device and Materials Reliability*, vol. 8, 2008.
- [137] S. Park, C. Floresca, U. Chowdhury, J. Jimenez, C. Lee, E. Beam, P. Saunier, T. Balistren, and M. Kim, "Physical degragation of gan hemt devices under high drain bias reliability testing," *Microelectronics Reliability*, vol. 49, 2009.
- [138] D. Cheney, E. Douglas, L. Liu, Y. Lo, C. adn Xi, B. Gila, F. Ren, D. Horton, M. Law, D. Smith, and S. Pearton, "Reliability studies of algan/gan high electron mobility transistors," *Semiconductor Science and Technology*, vol. 28, 2013.
- [139] S. Witczak, J. Suehle, and M. Gaitan, "An experimental comparison of measurement techniques to extract si-sio2 interface trap density," *Solid-State Electronics*, vol. 35, 1992.
- [140] T. Ruemenapp and D. Peier, "Dielectric breakdown in aluminium nitride," Symposium on High Voltage Engineering, 1992.
- [141] Z. An, C. Men, Z. Xu, P. Chu, and C. Lin, "Electrical properties of aln thin films prepared by ion beam enhanced deposition," *Surface and Coatings Technology*, vol. 196, 2005.
- [142] P. Punchaipetch, M. Miyashita, Y. Uraoka, T. Fuyuki, T. Sameshima, and S. Horii, "Improving high-k gate dielectric properties by high-pressure water vapor annealing," *Japanese Journal of Applied Physics*, vol. 45, 2006.
- [143] H. Iwai, S. Ohmi, S. Akama, C. Ohshima, A. Kikuchi, I. Kashiwagi, J. Taguchi, H. Yamamoto, J. Tonotani, Y. Kim, I. Ueda, A. Kuriyama, and Y. Yoshihara, "Advanced gate dielectric materials for sub-100nm cmos," *International Electron Device Meeting*, pp. 625–628, 2002.
- [144] W. Kao, "Impact on interface-states on sub-threshold response of iii-v mosfets, mos hemts and tunnel fets," *MSc. Thesis*, 2010.
- [145] P. McWhorter and P. Winokur, "Simple technique for separating the effects of interface traps and trapped oxide charge in metal oxide semiconductor transistors," *Applied Physics Letters*, vol. 48, no. 2, pp. 133–135, 1986.
- [146] J. Ao, T. Wang, D. Kikuta, Y. Liu, S. Sakai, and Y. Ohno, "Algan/gan high electron mobility transistor with thin buffer layers," *Japanese Journal of Applied Physics*, vol. 42, pp. 1588–1589, 2003.
- [147] K. Im, J. Ha, K. Kim, J. Lee, D. Kim, S. Hahm, and J. Lee, "Normally off gan mosfets based on algan/gan heterostructure with extremely high 2deg density grown on silicon substrate," *IEEE Electron Device Letters*, vol. 31, 2010.

- [148] W. Dongfang, Y. Tingting, W. Ke, C. Xiaojuan, and L. Xinyu, "Gate-structure optimization for high frequency power AlGaN/GaN HEMTs," *Journal of Semiconductors*, vol. 31, p. 054003, May 2010.
- [149] D. F. Brown, K. Shinohara, A. L. Corrion, R. Chu, A. Williams, J. C. Wong, I. Alvarado-rodriguez, R. Grabar, M. Johnson, C. M. Butler, D. Santos, S. D. Burnham, J. F. Robinson, D. Zehnder, S. J. Kim, T. C. Oh, and M. Micovic, "High-Speed , Enhancement-Mode GaN Power Switch With Regrown n + GaN Ohmic Contacts and Staircase Field Plates," vol. 34, no. 9, pp. 1118–1120, 2013.
- [150] K. Bothe, M. Johnson, and D. Barlage, "TCAD Modeling and Simulation of a Field Plated GaN MOSFET for High Voltage Applications," no. 780, pp. 367–370, 2011.
- [151] Y. Dora, "Understanding material and process limits for high breakdown voltage algan/gan hemts," *Phd. Thesis*, 2006.
- [152] P. Ko, T. Chan, A. Wu, and C. Hu, "The effects of weak gate-to drain(source) overlap on mosfet characteristics," *International Electron Device Meeting*, 1986.
- [153] Y. Yamaguchi, T. Iwamatsu, H. Joachim, H. Oda, Y. Inoue, T. Nishimura, and K. Tsukamoto, "Source-to-drain breakdown voltage improvement in ultrathin-film soi mosfet's using a gate-overlapped ldd structure," *IEEE Transactions on Electron Devices*, vol. 41, no. 7, pp. 1222–1226, 1944.
- [154] A. Rana, N. Chand, and V. Kapoor, "A novel source/drain-to-gate non-overlap mosfet to reduced gate leakage current in nano regime," World Academy of Science, Enigneering and Technology, vol. 5, pp. 755–761, 2011.
- [155] J. Jang, T. Lim, and Y. Kim, "Optimum source/drain overlap design for 16nm highk/metal gate mosfets," *Semiconductor Science and Technology*, vol. 24, p. 105009, 2009.
- [156] K. Maitra and N. Bhat, "Impact of gate-to-source/drain overlap length on 80-nm cmos circuit performace," *IEEE Transactions on Electron Devices*, vol. 51, 2004.